# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

## FORM 10-Q

	QUARTERLI REPORT PURSUANT TO SEC	LITON 15 OK 15(u) OF THE SE	CURITIES EXCHANGE ACT OF 1954
	Fo	r the quarterly period ended Jur	e 30, 2020
		OR	
	TRANSITION REPORT PURSUANT TO SEC	CTION 13 OR 15(d) OF THE SE	CURITIES EXCHANGE ACT OF 1934
		nsition period from	to
		Commission file number 001-32	559
		Commission file number 333-17	7186
	MEDICA	L PROPERTIES	TRUST, INC.
		ERATING PARTN	
		Name of Registrant as Specified	· · · · · · · · · · · · · · · · · · ·
	·		_
	MARYLAND DELAWARE		20-0191742 20-0242069
	(State or other jurisdiction of		(I. R. S. Employer
	incorporation or organization)	7 501	Identification No.)
	1000 URBAN CENTER DRIVE, SUITI BIRMINGHAM, AL (Address of principal executive offices)	2 501	35242 (Zip Code)
	REGISTRANT'S TELE	PHONE NUMBER, INCLUDIN	G AREA CODE: (205) 969-3755
12 m	cate by check mark whether the registrant (1) has filed all roonths (or for such shorter period that the registrant was reasys. Yes $oxtimes$ No $oxdot$	1 1	3 or 15(d) of the Securities Exchange Act of 1934 during the preceding been subject to such filing requirements for the past
	cate by check mark whether the registrant has submitted el- ng the preceding 12 months (or for such shorter period that		required to be submitted pursuant to Rule 405 of Regulation S-T uch files). Yes $\boxtimes$ No $\square$
	9		celerated filer, a smaller reporting company or an emerging growth any" and "emerging growth company" in Rule 12b-2 of the Exchange
_	•	operties Trust, Inc. only) ting Partnership, L.P. only)	Accelerated filer  Smaller reporting company  Emerging growth company
	emerging growth company, indicate by check mark if the icial accounting standards provided pursuant to Section 13	9	ended transition period for complying with any new or revised
	Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Cor	nmon stock, par value \$0.001 per share, of Medical Properties Trust, Inc.	MPW	The New York Stock Exchange
Indic	cate by check mark whether the registrant is a shell compar	ny (as defined in Rule 12b-2 of the Exc	hange Act). Yes □ No ⊠
As o	f July 31, 2020, Medical Properties Trust, Inc. had 528,829	9,474 shares of common stock, par val	ue \$0.001, outstanding.

## EXPLANATORY NOTE

This report combines the Quarterly Reports on Form 10-Q for the three and six months ended June 30, 2020 of Medical Properties Trust, Inc., a
Maryland corporation, and MPT Operating Partnership, L.P., a Delaware limited partnership, through which Medical Properties Trust, Inc. conducts
substantially all of its operations. Unless otherwise indicated or unless the context requires otherwise, all references in this report to "we," "us," "our,"
"Medical Properties," "MPT," or the "company" refer to Medical Properties Trust, Inc. together with its consolidated subsidiaries, including MPT
Operating Partnership, L.P. Unless otherwise indicated or unless the context requires otherwise, all references to "operating partnership" refer to MPT
Operating Partnership, L.P. together with its consolidated subsidiaries.

## MEDICAL PROPERTIES TRUST, INC. AND MPT OPERATING PARTNERSHIP, L.P. AND SUBSIDIARIES **QUARTERLY REPORT ON FORM 10-Q**

## FOR THE QUARTERLY PERIOD ENDED June 30, 2020

### **Table of Contents**

	Page
PART I — FINANCIAL INFORMATION	3
<u>Item 1 Financial Statements</u>	3
Medical Properties Trust, Inc. and Subsidiaries	
Condensed Consolidated Balance Sheets at June 30, 2020 and December 31, 2019	3
Condensed Consolidated Statements of Net Income for the Three and Six Months Ended June 30, 2020 and 2019	4
Condensed Consolidated Statements of Comprehensive Income for the Three and Six Months Ended June 30, 2020 and 2019	5
Condensed Consolidated Statements of Equity for the Three and Six Months Ended June 30, 2020 and 2019	6
Condensed Consolidated Statements of Cash Flows for the Six Months Ended June 30, 2020 and 2019	7
MPT Operating Partnership, L.P. and Subsidiaries	
Condensed Consolidated Balance Sheets at June 30, 2020 and December 31, 2019	8
Condensed Consolidated Statements of Net Income for the Three and Six Months Ended June 30, 2020 and 2019	9
Condensed Consolidated Statements of Comprehensive Income for the Three and Six Months Ended June 30, 2020 and 2019	10
Condensed Consolidated Statements of Capital for the Three and Six Months Ended June 30, 2020 and 2019	11
Condensed Consolidated Statements of Cash Flows for the Six Months Ended June 30, 2020 and 2019	12
Medical Properties Trust, Inc. and MPT Operating Partnership, L.P.	
Notes to Condensed Consolidated Financial Statements	13
Item 2 Management's Discussion and Analysis of Financial Condition and Results of Operations	22
Item 3 Quantitative and Qualitative Disclosures about Market Risk	33
<u>Item 4 Controls and Procedures</u>	34
PART II — OTHER INFORMATION	35
Item 1 Legal Proceedings	35
Item 1A Risk Factors	35
Item 2 Unregistered Sales of Equity Securities and Use of Proceeds	35
Item 3 Defaults Upon Senior Securities	35
Item 4 Mine Safety Disclosures	35
Item 5 Other Information	35
Item 6 Exhibits	36
SIGNATURE	37

## PART I — FINANCIAL INFORMATION

### Item 1. Financial Statements.

## MEDICAL PROPERTIES TRUST, INC. AND SUBSIDIARIES

### Condensed Consolidated Balance Sheets

		June 30, 2020		December 31, 2019
(In thousands, except per share amounts)		(Unaudited)		(Note 2)
Assets				
Real estate assets	ф.	0.000.000	Φ.	0.400 == 4
Land, buildings and improvements, intangible lease assets, and other	\$	9,990,860	\$	8,102,754
Investment in financing leases		2,078,209		2,060,302
Mortgage loans		1,339,258		1,275,022
Gross investment in real estate assets		13,408,327		11,438,078
Accumulated depreciation and amortization		(684,444)		(570,042)
Net investment in real estate assets		12,723,883		10,868,036
Cash and cash equivalents		374,962		1,462,286
Interest and rent receivables		41,321		31,357
Straight-line rent receivables		377,999		334,231
Equity investments		841,098		926,990
Other loans		792,011		544,832
Other assets		296,796		299,599
Total Assets	\$	15,448,070	\$	14,467,331
Liabilities and Equity		<u>.                                      </u>		_
Liabilities				
Debt, net	\$	7,795,890	\$	7,023,679
Accounts payable and accrued expenses		443,453		291,489
Deferred revenue		18,638		16,098
Obligations to tenants and other lease liabilities		122,812		107,911
Total Liabilities		8,380,793		7,439,177
Equity				
Preferred stock, \$0.001 par value. Authorized 10,000 shares; no shares outstanding		_		_
Common stock, \$0.001 par value. Authorized 750,000 shares; issued and outstanding — 528,641 shares at June 30, 2020 and 517,522 shares at December 31, 2019		529		518
Additional paid-in capital		7,200,203		7,008,199
Retained (deficit) earnings		(19,771)		83,012
Accumulated other comprehensive loss		(113,013)		(62,905)
Treasury shares, at cost		(777)		(777)
Total Medical Properties Trust, Inc. stockholders' equity		7,067,171		7,028,047
Non-controlling interests		106		107
Total Equity		7,067,277		7,028,154
Total Liabilities and Equity	\$	15,448,070	\$	14,467,331
Total Elaumites and Equity	<b>3</b>	15,440,070	Ф	14,407,331

## Condensed Consolidated Statements of Net Income (Unaudited)

		For the Th Ended .			For the Six Months Ended June 30,			
(In thousands, except per share amounts)		2020		2019		2020		2019
Revenues								
Rent billed	\$	173,557	\$	110,882	\$	345,324	\$	219,480
Straight-line rent		21,151		25,136		52,572		45,787
Income from financing leases		52,489		17,386		104,925		34,666
Interest and other income		44,645		39,145		83,153		73,070
Total revenues		291,842		192,549		585,974		373,003
Expenses								
Interest		80,376		52,326		161,275		102,877
Real estate depreciation and amortization		61,463		33,976		122,384		67,328
Property-related		9,985		8,290		15,557		11,356
General and administrative		32,018		22,272		65,403		45,723
Total expenses		183,842		116,864		364,619		227,284
Other income (expense)								
Loss on sale of real estate		(3,101)		(147)		(1,776)		(147)
Real estate impairment charges		_		_		(19,006)		_
Earnings from equity interests		5,291		4,441		9,370		8,161
Unutilized financing fees		_		(914)		(611)		(914)
Other (including mark-to-market adjustments on equity								
securities)		4,291		581		(9,684)		785
Total other income (expense)		6,481		3,961		(21,707)		7,885
Income before income tax		114,481		79,646		199,648		153,604
Income tax (expense) benefit		(4,829)		274		(8,839)		2,607
						<u> </u>		
Net income		109,652		79,920		190,809		156,211
Net income attributable to non-controlling interests		(184)		(482)		(349)		(951)
Net income attributable to MPT common stockholders	\$	109,468	\$	79,438	\$	190,460	\$	155,260
Earnings per common share — basic and diluted								
Net income attributable to MPT common stockholders	\$	0.21	\$	0.20	\$	0.36	\$	0.40
			_		_		_	
Weighted average shares outstanding — basic		527,781		394,574		524,428		387,563
Weighted average shares outstanding — diluted		528,880		395,692		525,530		388,683
	_	220,000	_	333,332	_	3_2,230	_	555,555
Dividends declared per common share	\$	0.27	\$	0.25	\$	0.54	\$	0.50
					<u> </u>		<u> </u>	

Condensed Consolidated Statements of Comprehensive Income (Unaudited)

	For the Th Ended .		For the Si Ended J			
(In thousands)	2020		2019	2020		2019
Net income	\$ 109,652	\$	79,920	\$ 190,809	\$	156,211
Other comprehensive income:						
Unrealized loss on interest rate swap	(7,908)		(1,486)	(33,011)		(5,258)
Foreign currency translation gain (loss)	6,175		2,848	(17,097)		(3,070)
Total comprehensive income	 107,919		81,282	 140,701		147,883
Comprehensive income attributable to non-controlling						
interests	(184)		(482)	(349)		(951)
Comprehensive income attributable to MPT common	 			 		
stockholders	\$ 107,735	\$	80,800	\$ 140,352	\$	146,932

Condensed Consolidated Statements of Equity (Unaudited)

	Preferred Common										
(In thousands, except per share amounts)	Shares	Par Value	Shares	Par Value	Additional Retained Paid-in Earnings Capital (Deficit)		Accumulated Other Comprehensive Loss	Treasury Stock	Non- Controlling Interests	Total Equity	
Balance at December 31, 2019		\$ —	517,522	\$ 518	\$ 7,008,199	\$ 83,012	\$ (62,905)	\$ (777)	\$ 107	\$ 7,028,154	
Net income				_		80,992			165	81,157	
Cumulative effect of change in accounting principles	_	_	_	_	_	(8,399)	_	_	_	(8,399)	
Unrealized loss on interest rate swap, net of tax	_	_	_	_	_	_	(25,103)	_	_	(25,103)	
Foreign currency translation loss	_	_	_	_	_	_	(23,272)	_	_	(23,272)	
Stock vesting and amortization of											
stock-based compensation			2,312	2	10,034					10,036	
Distributions to non-controlling interests Proceeds from offering (net of	_	_	_	_	_	_	_	_	(165)	(165)	
offering costs)	_		2,601	2	61,680	_		_	_	61,682	
Dividends declared (\$0.27 per			2,001	_	01,000					01,002	
common share)						(141,580)				(141,580)	
Balance at March 31, 2020		<u>\$</u>	522,435	\$ 522	\$ 7,079,913	\$ 14,025	\$ (111,280)	<b>\$</b> (777)	\$ 107	\$ 6,982,510	
Net income	_	_	_	_	_	109,468	_	_	184	109,652	
Unrealized loss on interest rate swap, net of tax	_	_	_	_	_	_	(7,908)	_	_	(7,908)	
Foreign currency translation gain	_	_	_	_	_	_	6,175	_	_	6,175	
Stock vesting and amortization of stock-based compensation			189	1	12,191					12,192	
Distributions to non-controlling interests			103		12,131	_	_		(185)	(185)	
Proceeds from offering (net of									(105)	(105)	
offering costs)	_	_	6,017	6	108,099	_	_	_	_	108,105	
Dividends declared (\$0.27 per common share)						(143,264)				(143,264)	
Balance at June 30, 2020		<u> </u>	528,641	\$ 529	\$ 7,200,203	\$ (19,771)	\$ (113,013)	\$ (777)	\$ 106	\$ 7,067,277	
			Common								
	Prefe	erred	Com	ımon							
	Prefe		Con		Additional	Retained	Accumulated Other	Тиоления	Non-	Total	
(In thousands, except per share amounts)		Par		Par	Paid-in	Earnings	Other Comprehensive	Treasury Stock	Controlling	Total Equity	
(In thousands, except per share amounts) Balance at December 31, 2018	Shares		Shares 370,637				Other	Treasury Stock \$ (777)		Total <u>Equity</u> \$ 4,560,938	
` ' I I ' '		Par Value	Shares	Par Value	Paid-in Capital	Earnings (Deficit)	Other Comprehensive Loss	Stock	Controlling Interests	Equity	
Balance at December 31, 2018		Par Value	Shares	Par Value	Paid-in Capital	Earnings (Deficit) \$ 162,768	Other Comprehensive Loss	Stock	Controlling Interests \$ 13,830	<b>Equity</b> \$ 4,560,938	
Balance at December 31, 2018  Net income  Unrealized loss on interest rate swap, net of		Par Value	Shares	Par Value	Paid-in Capital	Earnings (Deficit) \$ 162,768	Other Comprehensive Loss \$ (58,202)	Stock	Controlling Interests \$ 13,830	Equity \$ 4,560,938 76,291	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of		Par Value	Shares 370,637	Par Value \$ 371 —	Paid-in Capital \$ 4,442,948 — — —	Earnings (Deficit) \$ 162,768	Other Comprehensive Loss \$ (58,202)	Stock	Controlling Interests \$ 13,830 469	Equity \$ 4,560,938 76,291 (3,772) (5,918)	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation		Par Value	Shares 370,637	Par Value \$ 371	Paid-in Capital	Earnings (Deficit) \$ 162,768	Other Comprehensive Loss \$ (58,202)	Stock	Controlling Interests \$ 13,830 469	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715	
Balance at December 31, 2018  Net income Unrealized loss on interest rate swap, net of tax  Foreign currency translation loss  Stock vesting and amortization of stock-based compensation  Distributions to non-controlling interests		Par Value	Shares 370,637	Par Value \$ 371 —	Paid-in Capital \$ 4,442,948 — — —	Earnings (Deficit) \$ 162,768	Other Comprehensive Loss \$ (58,202)	Stock	Controlling Interests \$ 13,830 469	Equity \$ 4,560,938 76,291 (3,772) (5,918)	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of		Par Value	Shares 370,637 — — — — 1,055	Par Value \$ 371 — — — 1	Paid-in Capital \$ 4,442,948 ————————————————————————————————————	Earnings (Deficit) \$ 162,768	Other Comprehensive Loss \$ (58,202)	Stock	Controlling Interests \$ 13,830 469	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645)	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per		Par Value	Shares 370,637	Par Value \$ 371 —	Paid-in Capital \$ 4,442,948 — — —	Earnings (Deficit) § 162,768 75,822	Other Comprehensive Loss \$ (58,202)	Stock	Controlling Interests \$ 13,830 469	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030	
Balance at December 31, 2018  Net income Unrealized loss on interest rate swap, net of tax  Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per common share)		Par Value	Shares 370,637 — — 1,055 — 20,147	Par Value \$ 371 — — — — — — 20	Paid-in Capital \$ 4,442,948 ————————————————————————————————————	Earnings (Deficit) \$ 162,768 75,822	Other Comprehensive Loss \$ (58,202)	Stock (777)	Controlling   Interests   13,830   469	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030 (97,163)	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per		Par Value	Shares 370,637 — — — — 1,055	Par Value \$ 371 — — — — — — 20	Paid-in Capital \$ 4,442,948 ————————————————————————————————————	Earnings (Deficit) § 162,768 75,822	Other Comprehensive Loss \$ (58,202)	Stock	Controlling   Interests   13,830   469	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per common share)  Balance at March 31, 2019		Par Value	Shares 370,637 — — 1,055 — 20,147	Par Value \$ 371 — — — — — — 20	Paid-in Capital \$ 4,442,948 ————————————————————————————————————	Earnings (Deficit) \$ 162,768 75,822 ———————————————————————————————————	Other Comprehensive Loss \$ (58,202)	Stock (777)	Controlling Interests  \$ 13,830 469	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030 (97,163) \$ 4,890,476 79,920	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per common share)  Balance at March 31, 2019 Net income Unrealized loss on interest rate swap, net of		Par Value	Shares 370,637 — — 1,055 — 20,147	Par Value \$ 371 — — — — — — 20	Paid-in Capital \$ 4,442,948 ————————————————————————————————————	Earnings (Deficit) \$ 162,768 75,822 ———————————————————————————————————	Other Comprehensive Loss \$ (58,202)	Stock (777)	Controlling Interests  \$ 13,830 469	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030 (97,163) \$ 4,890,476	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per common share) Balance at March 31, 2019 Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation gain Stock vesting and amortization of		Par Value	Shares 370,637  1,055 20,147 391,839	Par Value \$ 371 — — — — — — 20	Paid-in Capital \$ 4,442,948 ————————————————————————————————————	Earnings (Deficit) \$ 162,768 75,822 ———————————————————————————————————	Other Comprehensive Loss \$ (58,202)	Stock (777)	Controlling   Interests   \$ 13,830   469     (645)     (645)	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030 (97,163) \$ 4,890,476 79,920 (1,486) 2,848	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per common share)  Balance at March 31, 2019 Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation gain Stock vesting and amortization of stock-based compensation		Par Value	Shares 370,637 — — 1,055 — 20,147	Par Value \$ 371	Paid-in Capital \$ 4,442,948 ————————————————————————————————————	Earnings (Deficit) \$ 162,768 75,822 ———————————————————————————————————	Other Comprehensive Loss \$ (58,202)	Stock	Controlling   Interests   13,830   469   469   465   465   482	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030 (97,163) \$ 4,890,476 79,920 (1,486) 2,848 6,317	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per common share) Balance at March 31, 2019 Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation gain Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests		Par Value	Shares 370,637  1,055 20,147 391,839	Par Value \$ 371 — — — — — — 20	Paid-in Capital \$ 4,442,948 ————————————————————————————————————	Earnings (Deficit) \$ 162,768 75,822 ———————————————————————————————————	Other Comprehensive Loss \$ (58,202)	Stock (777)	Controlling   Interests   \$ 13,830   469     (645)     (645)	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030 (97,163) \$ 4,890,476 79,920 (1,486) 2,848	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per common share) Balance at March 31, 2019 Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation gain Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs)		Par Value	Shares 370,637  1,055 20,147 391,839	Par Value \$ 371	Paid-in Capital \$ 4,442,948 ————————————————————————————————————	Earnings (Deficit) \$ 162,768 75,822 ———————————————————————————————————	Other Comprehensive Loss \$ (58,202)	Stock	Controlling   Interests   13,830   469   469   465   465   482	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030 (97,163) \$ 4,890,476 79,920 (1,486) 2,848 6,317	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per common share) Balance at March 31, 2019 Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation gain Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of		Par Value	Shares  370,637   1,055  20,147  391,839  119	Par Value \$ 371	Paid-in Capital \$ 4,442,948	Earnings (Deficit) \$ 162,768 75,822 ———————————————————————————————————	Other Comprehensive Loss \$ (58,202)	Stock	Controlling   Interests   13,830   469   469   465   465   482	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030 (97,163) \$ 4,890,476 79,920 (1,486) 2,848 6,317 (670)	
Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation loss Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per common share) Balance at March 31, 2019 Net income Unrealized loss on interest rate swap, net of tax Foreign currency translation gain Stock vesting and amortization of stock-based compensation Distributions to non-controlling interests Proceeds from offering (net of offering costs) Dividends declared (\$0.25 per		Par Value	Shares  370,637   1,055  20,147  391,839  119	Par Value \$ 371	Paid-in Capital \$ 4,442,948	Earnings (Deficit) \$ 162,768 75,822 ———————————————————————————————————	Other Comprehensive Loss \$ (58,202)	Stock	Controlling   Interests   13,830   469   469   465   465   482	Equity \$ 4,560,938 76,291 (3,772) (5,918) 6,715 (645) 354,030 (97,163) \$ 4,890,476 79,920 (1,486) 2,848 6,317 (670) 45,323	

Condensed Consolidated Statements of Cash Flows (Unaudited)

		For the Six Months Ended June 30,				
		2020	ounc 50,	2019		
		(In tho	usands)			
Operating activities	¢	100 000	¢	150 211		
Net income	\$	190,809	\$	156,211		
Adjustments to reconcile net income to net cash provided by operating activities:		120.215		CO 420		
Depreciation and amortization		128,215		69,430		
Amortization of deferred financing costs and debt discount		6,438		3,816		
Straight-line rent revenue and other		(96,730)		(54,070)		
Share-based compensation		22,228		13,032		
Loss from sale of real estate and other		1,776		147		
Impairment charges		19,006				
Straight-line rent and other write-off, net of tax		26,958		3,002		
Unutilized financing costs		611		914		
Pre-acquisition rent collected - Circle Transaction		(35,020)		_		
Other adjustments		12,373		12,774		
Changes in:						
Interest and rent receivables		186		(265)		
Accounts payable and accrued expenses		12,216		(6,885)		
Net cash provided by operating activities		289,066		198,106		
Investing activities						
Cash paid for acquisitions and other related investments		(2,226,302)		(1,402,315)		
Net proceeds from sale of real estate		78,764		3,449		
Principal received on loans receivable		_		420		
Investment in loans receivable		(62,064)		(2,992)		
Other investments, net		29,322		(205,473)		
Net cash used for investing activities		(2,180,280)		(1,606,911)		
Financing activities						
Proceeds from term debt		915,950		837,240		
Revolving credit facilities, net		_		12,976		
Dividends paid		(279,741)		(192,582)		
Lease deposits and other obligations to tenants		5,001		3,485		
Proceeds from sale of common shares, net of offering costs		169,787		399,353		
Payment of deferred financing costs and other financing activities		(7,005)		(9,432)		
Net cash provided by financing activities		803,992		1,051,040		
Decrease in cash, cash equivalents, and restricted cash for period		(1,087,222)	·	(357,765)		
Effect of exchange rate changes		155		(8,050)		
Cash, cash equivalents, and restricted cash at beginning of period		1,467,991		822,425		
	¢	380,924	¢			
Cash, cash equivalents, and restricted cash at end of period	\$		\$	456,610		
Interest paid	\$	147,502	\$	97,184		
Supplemental schedule of non-cash financing activities:						
Dividends declared, unpaid	\$	143,264	\$	99,093		
Cash, cash equivalents, and restricted cash are comprised of the following:						
Beginning of period:						
Cash and cash equivalents	\$	1,462,286	\$	820,868		
Restricted cash, included in Other assets		5,705		1,557		
	\$	1,467,991	\$	822,425		
End of period:						
Cash and cash equivalents	\$	374,962	\$	451,652		
Restricted cash, included in Other assets		5,962		4,958		
·	\$	380,924	\$	456,610		
	Ψ	300,324	<u> </u>	150,010		

### Condensed Consolidated Balance Sheets

	June 30, 2020	December 31, 2019
(In thousands)	(Unaudited)	(Note 2)
Assets		
Real estate assets		
Land, buildings and improvements, intangible lease assets, and other	\$ 9,990,860	\$ 8,102,754
Investment in financing leases	2,078,209	2,060,302
Mortgage loans	 1,339,258	 1,275,022
Gross investment in real estate assets	13,408,327	11,438,078
Accumulated depreciation and amortization	 (684,444)	 (570,042)
Net investment in real estate assets	12,723,883	10,868,036
Cash and cash equivalents	374,962	1,462,286
Interest and rent receivables	41,321	31,357
Straight-line rent receivables	377,999	334,231
Equity investments	841,098	926,990
Other loans	792,011	544,832
Other assets	296,796	299,599
Total Assets	\$ 15,448,070	\$ 14,467,331
Liabilities and Capital		
Liabilities		
Debt, net	\$ 7,795,890	\$ 7,023,679
Accounts payable and accrued expenses	299,861	152,999
Deferred revenue	18,638	16,098
Obligations to tenants and other lease liabilities	122,812	107,911
Payable due to Medical Properties Trust, Inc.	143,202	138,100
Total Liabilities	8,380,403	7,438,787
Capital		
General Partner — issued and outstanding — 5,287 units at June 30, 2020		
and 5,176 units at December 31, 2019	71,831	70,939
Limited Partners:		
Common units — issued and outstanding — 523,354 units at		
June 30, 2020 and 512,346 units at December 31, 2019	7,108,743	7,020,403
LTIP units — issued and outstanding — 232 units at June 30, 2020		
and December 31, 2019	_	_
Accumulated other comprehensive loss	 (113,013)	 (62,905)
Total MPT Operating Partnership, L.P. capital	7,067,561	7,028,437
Non-controlling interests	106	107
Total Capital	7,067,667	7,028,544
Total Liabilities and Capital	\$ 15,448,070	\$ 14,467,331

## Condensed Consolidated Statements of Net Income (Unaudited)

	For the Three Months Ended June 30,					For the Six Months Ended June 30,			
(In thousands, except per unit amounts)		2020		2019		2020		2019	
Revenues									
Rent billed	\$	173,557	\$	110,882	\$	345,324	\$	219,480	
Straight-line rent		21,151		25,136		52,572		45,787	
Income from financing leases		52,489		17,386		104,925		34,666	
Interest and other income		44,645		39,145		83,153		73,070	
Total revenues		291,842		192,549		585,974		373,003	
Expenses									
Interest		80,376		52,326		161,275		102,877	
Real estate depreciation and amortization		61,463		33,976		122,384		67,328	
Property-related		9,985		8,290		15,557		11,356	
General and administrative		32,018		22,272		65,403		45,723	
Total expenses		183,842		116,864		364,619		227,284	
Other income (expense)									
Loss on sale of real estate		(3,101)		(147)		(1,776)		(147)	
Real estate impairment charges		_		_		(19,006)		_	
Earnings from equity interests		5,291		4,441		9,370		8,161	
Unutilized financing fees		_		(914)		(611)		(914)	
Other (including mark-to-market adjustments on equity									
securities)		4,291		581		(9,684)		785	
Total other income (expense)		6,481		3,961		(21,707)		7,885	
Income before income tax		114,481		79,646		199,648		153,604	
Income tax (expense) benefit		(4,829)		274		(8,839)		2,607	
Net income		109,652		79,920		190,809		156,211	
Net income attributable to non-controlling interests		(184)		(482)		(349)		(951)	
Net income attributable to MPT Operating Partnership									
partners	\$	109,468	\$	79,438	\$	190,460	\$	155,260	
Earnings per unit — basic and diluted									
Net income attributable to MPT Operating Partnership partners	\$	0.21	\$	0.20	\$	0.36	\$	0.40	
	<u> </u>								
Weighted average units outstanding — basic		527,781		394,574		524,428		387,563	
Weighted average units outstanding — diluted		528,880	_	395,692	_	525,530		388,683	
g		323,330		223,032	_	3=3,330		300,000	
Dividends declared per unit	\$	0.27	\$	0.25	\$	0.54	\$	0.50	
= weeker per mine	¥	<b>V.</b> _/	<del>-</del>	<b>0.2</b> 3	<del>-</del>	0.57	_	0.50	

Condensed Consolidated Statements of Comprehensive Income (Unaudited)

	For the Three Months					For the Si Ended J				
(In thousands)		2020		2019		2020		2020		2019
Net income	\$	109,652	\$	79,920	\$	190,809	\$	156,211		
Other comprehensive income:										
Unrealized loss on interest rate swap		(7,908)		(1,486)		(33,011)		(5,258)		
Foreign currency translation gain (loss)		6,175		2,848		(17,097)		(3,070)		
Total comprehensive income		107,919		81,282		140,701		147,883		
Comprehensive income attributable to non-controlling interests		(184)		(482)		(349)		(951)		
Comprehensive income attributable to MPT Operating Partnership	· <u> </u>		'							
partners	\$	107,735	\$	80,800	\$	140,352	\$	146,932		

## Condensed Consolidated Statements of Capital (Unaudited)

	Gen	eral		Limited Pa	artners	Accumulated			
	Par	tner	Con	nmon	LT	IPs	Other	Non-	
(In thousands, except per unit amounts)	Units			Unit Value	Units	Unit Value	Comprehensive Loss	Controlling Interests	Total Capital
Balance at December 31, 2019	5,176	\$ 70,939	512,346	\$ 7,020,403	232	<u>\$</u>	\$ (62,905)	\$ 107	\$ 7,028,544
Net income	_	810	_	80,182	_	_	_	165	81,157
Cumulative effect of change in accounting principles	_	(84)	_	(8,315)	_	_	_	_	(8,399)
Unrealized loss on interest rate swap, net of									
tax	_	_	_	_	_	_	(25,103)		(25,103)
Foreign currency translation loss							(23,272)		(23,272)
Unit vesting and amortization of unit-based compensation	23	100	2,289	9,936	_	_	_	_	10,036
Distributions to non-controlling interests	_	_		_	_	_	_	(165)	(165)
Proceeds from offering (net of offering costs)	26	617	2,575	61,065	_	_	_	_	61,682
Distributions declared (\$0.27 per unit)	_	(1,416)		(140,164)	_	_	_	_	(141,580)
Balance at March 31, 2020	5,225	\$ 70,966	517,210	\$ 7,023,107	232	\$ —	\$ (111,280)	\$ 107	\$ 6,982,900
Net income		1,095		108,373				184	109,652
Unrealized loss on interest rate swap, net of tax	_	_	_	_	_	_	(7,908)	_	(7,908)
Foreign currency translation gain	_	_	_	_	_	_	6,175	_	6,175
Unit vesting and amortization of unit-based compensation	2	122	187	12,070	_	_	_	_	12,192
Distributions to non-controlling interests	_	_		_	_	_	_	(185)	(185)
Proceeds from offering (net of offering costs)	60	1,081	5,957	107,024	_	_	_	_	108,105
Distributions declared (\$0.27 per unit)	_	(1,433)	´ —	(141,831)	_	_	_	_	(143,264)
Balance at June 30, 2020	5,287	\$ 71,831	523,354	\$ 7,108,743	232	\$ <u> </u>	\$ (113,013)	\$ 106	\$ 7,067,667

	Gen	eral	Limited Partners				Accumulated		
	Part	Partner		nmon	LT	IPs	Other	Non-	
	¥1 •.	Unit	TT 1.	Unit	** ·	Unit	Comprehensive	Controlling	Total
(In thousands, except per unit amounts)	Units	Value	Units	Value	Units	Value	Loss	Interests	Capital
Balance at December 31, 2018	3,706	\$ 46,084	366,931	\$ 4,559,616	232	<u>\$</u>	\$ (58,202)	\$ 13,830	\$ 4,561,328
Net income		758		75,064				469	76,291
Unrealized loss on interest rate swap, net of									
tax	_	_	_	_	_	_	(3,772)	_	(3,772)
Foreign currency translation loss							(5,918)	_	(5,918)
Unit vesting and amortization of unit-based									
compensation	11	68	1,044	6,647	_	_	_		6,715
Distributions to non-controlling interests								(645)	(645)
Proceeds from offering (net of offering	204	2.540	10.016	250 400					25 4 222
costs)	201	3,540	19,946	350,490	_	_	_	_	354,030
Distributions declared (\$0.25 per unit)		(972)		(96,191)					(97,163)
Balance at March 31, 2019	3,918	<u>\$ 49,478</u>	387,921	<u>\$ 4,895,626</u>	232	<u>\$</u>	\$ (67,892)	\$ 13,654	\$ 4,890,866
Net income	_	794		78,644	_		_	482	79,920
Unrealized loss on interest rate swap, net of									
tax	_	_	_	_	_	_	(1,486)	_	(1,486)
Foreign currency translation gain							2,848		2,848
Unit vesting and amortization of unit-based									
compensation	1	63	118	6,254	_	_	_	_	6,317
Distributions to non-controlling interests								(670)	(670)
Proceeds from offering (net of offering									
costs)	25	453	2,442	44,870	_	_	_	_	45,323
Distributions declared (\$0.25 per unit)		(991)		(98,102)					(99,093)
Balance at June 30, 2019	3,944	\$ 49,797	390,481	\$ 4,927,292	232	<u>\$</u>	\$ (66,530)	\$ 13,466	\$ 4,924,025

Condensed Consolidated Statements of Cash Flows (Unaudited)

		For the Six Months Ended June 30,			
		2020		2019	
Onewating activities		(In tho	usands)		
Operating activities  Net income	\$	190,809	\$	156 211	
Adjustments to reconcile net income to net cash provided by operating activities:	Þ	190,009	Ф	156,211	
		120 215		60.420	
Depreciation and amortization		128,215		69,430	
Amortization of deferred financing costs and debt discount Straight-line rent revenue and other		6,438		3,816	
ů		(96,730) 22,228		(54,070)	
Unit-based compensation Loss from sale of real estate and other				13,032	
		1,776		147	
Impairment charges		19,006		2.002	
Straight-line rent and other write-off, net of tax		26,958		3,002	
Unutilized financing costs		611		914	
Pre-acquisition rent collected - Circle Transaction		(35,020)		10.774	
Other adjustments		12,373		12,774	
Changes in:		100		(205)	
Interest and rent receivables		186		(265)	
Accounts payable and accrued expenses		12,216		(6,885)	
Net cash provided by operating activities		289,066		198,106	
Investing activities					
Cash paid for acquisitions and other related investments		(2,226,302)		(1,402,315)	
Net proceeds from sale of real estate		78,764		3,449	
Principal received on loans receivable		_		420	
Investment in loans receivable		(62,064)		(2,992)	
Other investments, net		29,322		(205,473)	
Net cash used for investing activities		(2,180,280)		(1,606,911)	
Financing activities					
Proceeds from term debt		915,950		837,240	
Revolving credit facilities, net		_		12,976	
Distributions paid		(279,741)		(192,582)	
Lease deposits and other obligations to tenants		5,001		3,485	
Proceeds from sale of units, net of offering costs		169,787		399,353	
Payment of deferred financing costs and other financing activities		(7,005)		(9,432)	
Net cash provided by financing activities		803,992		1,051,040	
Decrease in cash, cash equivalents, and restricted cash for period		(1,087,222)	-	(357,765)	
Effect of exchange rate changes		155		(8,050)	
Cash, cash equivalents, and restricted cash at beginning of period		1,467,991		822,425	
Cash, cash equivalents, and restricted cash at end of period	\$	380,924	\$	456,610	
	\$	147,502	\$	97,184	
Interest paid Supplemental schedule of non-cash financing activities:	Ф	147,502	Ф	97,104	
	¢.	1.40.004	¢.	00.003	
Distributions declared, unpaid	\$	143,264	\$	99,093	
Cash, cash equivalents, and restricted cash are comprised of the following:					
Beginning of period:	Φ.	4 462 206	Φ.	020.000	
Cash and cash equivalents	\$	1,462,286	\$	820,868	
Restricted cash, included in Other assets		5,705		1,557	
	\$	1,467,991	\$	822,425	
End of period:					
Cash and cash equivalents	\$	374,962	\$	451,652	
Restricted cash, included in Other assets		5,962		4,958	
	\$	380,924	\$	456,610	

## MEDICAL PROPERTIES TRUST, INC. AND MPT OPERATING PARTNERSHIP, L.P. AND SUBSIDIARIES

Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 1. Organization

Medical Properties Trust, Inc., a Maryland corporation, was formed on August 27, 2003, under the Maryland General Corporation Law for the purpose of engaging in the business of investing in, owning, and leasing commercial real estate. Our operating partnership subsidiary, MPT Operating Partnership, L.P., (the "Operating Partnership") through which we conduct all of our operations, was formed in September 2003. Through another whollyowned subsidiary, Medical Properties Trust, LLC, we are the sole general partner of the Operating Partnership. At present, we directly own substantially all of the limited partnership interests in the Operating Partnership and have elected to report our required disclosures and that of the Operating Partnership on a combined basis, except where material differences exist.

We have operated as a real estate investment trust ("REIT") since April 6, 2004, and accordingly, elected REIT status upon the filing in September 2005 of the calendar year 2004 federal income tax return. Accordingly, we will generally not be subject to United States ("U.S.") federal income tax, provided that we continue to qualify as a REIT and our distributions to our stockholders equal or exceed our taxable income. Certain non-real estate activities we undertake are conducted by entities which we elected to be treated as taxable REIT subsidiaries ("TRS"). Our TRS entities are subject to both U.S. federal and state income taxes. For our properties, located outside the U.S., we are subject to the local taxes of the jurisdictions where our properties reside and/or legal entities are domiciled; however, we do not expect to incur additional taxes in the U.S. as the majority of such income flows through our REIT.

Our primary business strategy is to acquire and develop real estate and improvements, primarily for long-term lease to providers of healthcare services, such as operators of general acute care hospitals, inpatient physical rehabilitation hospitals, and long-term acute care hospitals. We also make mortgage and other loans to operators of similar facilities. In addition, we may obtain profits or equity interests in our tenants, from time to time, in order to enhance our overall return.

Our business model facilitates acquisitions and recapitalizations, and allows operators of healthcare facilities to unlock the value of their real estate to fund facility improvements, technology upgrades, and other investments in operations. At June 30, 2020, we have investments in 387 facilities in 34 states in the U.S., in six countries in Europe, and across Australia. We manage our business as a single business segment.

#### 2. Summary of Significant Accounting Policies

Unaudited Interim Condensed Consolidated Financial Statements: The accompanying unaudited interim condensed consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the U.S. for interim financial information, including rules and regulations of the Securities and Exchange Commission ("SEC"). Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles ("GAAP") for complete financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair statement have been included. Operating results for the three and six months ended June 30, 2020, are not necessarily indicative of the results that may be expected for the year ending December 31, 2020. The condensed consolidated balance sheet at December 31, 2019 has been derived from the audited financial statements at that date but does not include all of the information and footnotes required by accounting principles generally accepted in the U.S. for complete financial statements.

The preparation of our consolidated financial statements in conformity with GAAP requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities as of the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. During the first half of 2020, the global outbreak of a novel coronavirus, or COVID-19, has spread all over the world including countries where we own and lease facilities. The World Health Organization has designated COVID-19 as a pandemic, and numerous countries, including the U.S., have declared national emergencies with respect to COVID-19. As the global impact of the outbreak evolved, many countries reacted by instituting quarantines and restrictions on travel, closing financial markets and/or restricting trade- including requiring medically necessary elective surgeries at hospitals to be deferred. Although hospitals are back accepting patients and performing medically necessary elective surgeries, many of these trade restrictions are still in place. We believe the estimates and assumptions underlying our consolidated financial statements are reasonable and supportable based on the information available as of June 30, 2020 (particularly as it relates to our assessments of the recoverability of our real estate and the adequacy of our credit loss reserves on loans and financing receivables). However, the ultimate impact to our tenants' results of operations and liquidity and their ability to pay our rent and interest due to the impact of COVID-19 cannot be predicted with 100% confidence, particularly given we are still learning the full scope, severity, and duration of the pandemic and the actions needed to contain the pandemic or mitigate its impact. This makes any estimates and assumptions as of June 30, 2020 inherently less certain than they would be absent the potential impact of COVID-19. Actual results may ultimately differ from our est

For information about significant accounting policies, refer to the consolidated financial statements and footnotes thereto included in our Annual Report on Form 10-K for the year ended December 31, 2019. There have been no material changes to these significant accounting policies other than the following:

On January 1, 2020, we adopted Accounting Standards Update ("ASU") No. 2016-13, "Measurement of Credit Losses on Financial Instruments" ("ASU 2016-13"). This standard requires a new forward-looking "expected loss" model measured over the contractual life of an asset that considers forecasts of future economic conditions, as well as past and current events, to be used for our financing receivables, including financing leases and loans, which the Financial Accounting Standards Board believes will result in more timely recognition of such losses. Additionally, we have made the accounting policy election to exclude interest receivables from the credit loss reserve analysis and will continue to timely reserve or write-off such short-term receivables.

Pursuant to ASU 2016-13, we grouped our financial instruments into two primary pools of similar credit risk: secured and unsecured. The secured instruments include investment in financing leases and mortgage loans, as all are secured by the underlying real estate among other collateral. The unsecured instruments include acquisition, working capital, and shareholder loans. Within these two major pools, we further grouped our instruments into sub-pools based on several tenant/borrower characteristics, including years of experience in the healthcare industry and in a particular market or region and overall capitalization. We then determined a credit loss percentage per pool based on our history of credit losses incurred on similar instruments. We used an historical period of time in determining such loss rates that closely matches the remaining terms of the financial instruments being analyzed in the respective pools, since our underwriting process has been consistent over this time. Finally, we made specific modifications for current trends, as appropriate.

Upon adoption of this standard, we recorded a credit loss reserve of \$8.4 million with the effect recorded in equity as a cumulative effect of a change in accounting principle.

#### Reclassifications

Certain amounts in the consolidated financial statements for prior periods have been reclassified to conform to the current period presentation.

#### 3. Real Estate and Lending Activities

#### New Investments

We acquired or invested in the following net assets (in thousands):

	 For the Six Months Ended June 30,			
	2020		2019	
Land and land improvements	\$ 265,991	\$	242,682	
Buildings	1,608,771		784,184	
Intangible lease assets — subject to amortization (weighted average useful				
life 30.8 years for 2020 and 18.5 years for 2019)	231,774		91,050	
Equity investments	_		284,399	
Mortgage loans	47,641		_	
Other assets	1,328		_	
Other loans	205,000		_	
Liabilities assumed	(134,203)		_	
Total	\$ 2,226,302	\$	1,402,315	

#### 2020 Activity

#### **Circle Transaction**

On January 8, 2020, we acquired a portfolio of 30 acute care hospitals located throughout the United Kingdom for approximately £1.5 billion from affiliates of BMI Healthcare, Inc. ("BMI"). In a related transaction, affiliates of Circle Health Ltd. ("Circle") entered into definitive agreements to acquire BMI and assume operations of its 52 facilities in the United Kingdom subject to customary regulatory conditions. As part of our acquisition, we inherited 30 existing leases with the operator that had initial fixed terms ending in 2050, with no renewal options but with annual inflation-based escalators. Once final regulatory approval was received in the 2020 second quarter, these 30 leases with Circle were amended (effective June 16, 2020) to include two five-year renewal options and improve the annual inflation-based escalators. These 30 leases are cross-defaulted and guaranteed by Circle.

#### Other Transactions

On June 24, 2020, we originated a CHF 45 million secured loan to Infracore SA ("Infracore").

On May 13, 2020, we formed a joint venture for the purpose of investing in the operations of international hospitals. As part of the formation, we originated a \$205 million acquisition loan. We have a 49% interest in this joint venture and are accounting for our investment using the fair value option election. The joint venture simultaneously purchased from Steward Health Care, Inc.

("Steward") the rights and existing assets related to all present and future international opportunities previously owned by Steward for strategic, regulatory, and risk management purposes.

#### 2019 Activity

On June 10, 2019, we acquired seven community hospitals in Kansas for approximately \$145.4 million. The properties are leased to an affiliate of Saint Luke's Health System ("SLHS") pursuant to seven individual in-place leases that had an average remaining lease term of 14 years upon our acquisition. The leases provide for fixed escalations every five years and include two five-year extension options. All seven hospitals were constructed in either 2018 or 2019, and the leases are guaranteed by SLHS.

On June 6, 2019, we acquired 11 hospitals in Australia for a purchase price of approximately AUD \$1.2 billion plus stamp duties and registration fees of AUD \$66.6 million. The properties are leased to Healthscope, Ltd. ("Healthscope"), pursuant to master lease agreements that had an average initial term of 20 years, upon our acquisition, with annual fixed escalations and multiple extension options. Healthscope was acquired in a simultaneous transaction by Brookfield Business Partners L.P. and certain of its institutional partners.

On May 27, 2019, we invested in a portfolio of 13 acute care campuses and two additional properties in Switzerland for an aggregate purchase price of approximately CHF 236.6 million. The investment was effected through our purchase of a stake in a Swiss healthcare real estate company, Infracore, from the previous majority shareholder, Aevis Victoria SA ("Aevis"). The facilities are leased to Swiss Medical Network, a wholly-owned Aevis subsidiary, pursuant to leases that had an average 23-year remaining term upon our acquisition and are subject to annual escalation provisions. We are accounting for our 40% interest in this joint venture under the equity method. Additionally, we purchased a 4.9% stake in Aevis for approximately CHF 47 million on June 28, 2019 that we are marking to fair value through income each quarter.

Other acquisitions throughout the first half of 2019 included two acute care hospitals and one inpatient rehabilitation hospital for an aggregate investment of approximately \$80 million.

#### **Development Activities**

On May 15, 2020, we agreed to finance the development of and lease an inpatient rehabilitation facility in Bakersfield, California for \$47.9 million. This facility will be leased to Ernest Health, Inc. ("Ernest") and is expected to commence rent in the fourth quarter of 2021.

During the 2020 second quarter, we completed construction on one general acute care facility and one inpatient rehabilitation facility, both located in Birmingham, England. We began recognizing revenue on these two properties on June 29, 2020. These facilities are being leased to Circle pursuant to a long-term lease.

During the 2020 first quarter, we completed construction and began recording rental income on a general acute care facility located in Idaho Falls, Idaho. This facility commenced rent on January 21, 2020 and is being leased to Surgery Partners, Inc. pursuant to an existing long-term lease.

See table below for a status update on our current development projects (in thousands):

<u>Property</u>	Сог	nmitment	Incurred as of ne 30, 2020	Estimated Rent Commencement Date
NeuroPsychiatric Hospitals (Houston, Texas)	\$	27,500	\$ 16,586	4Q 2020
Ernest (Bakersfield, California)		47,929	11,088	4Q 2021
	\$	75,429	\$ 27,674	

#### Disposals

During the first half of 2020, we completed the disposition of five facilities and four ancillary properties for approximately \$79 million. The transactions resulted in a net loss on real estate of \$1.8 million.

#### Leasing Operations (Lessor)

We acquire and develop healthcare facilities and lease the facilities to healthcare operating companies under long-term net leases (typical initial fixed terms ranging from 10 to 20 years) and most include renewal options at the election of our tenants, generally in five year increments. More than 97% of our leases provide annual rent escalations based on increases in Consumer Price Index (or similar index outside the U.S.) and/or fixed minimum annual escalations ranging from 0.5% to 3.0%. Many of our domestic leases contain purchase options with pricing set at various terms but in no case less than our total investment. For five properties with

a carrying value of \$216.3 million, our leases require a residual value guarantee from the tenant. Our leases typically require the tenant to handle and bear most of the costs associated with our properties including repairs/maintenance, property taxes, and insurance. We routinely inspect our properties to ensure the residual value of each of our assets is being maintained. Except for leases classified as financing leases, all of our leases are classified as operating leases.

At June 30, 2020, leases on 14 Ernest facilities and ten Prime Healthcare Services, Inc. ("Prime") facilities are accounted for as direct financing leases and leases on 13 of our Prospect Medical Holdings, Inc. ("Prospect") facilities are accounted for as a financing. The components of our total investment in financing leases consisted of the following (in thousands):

	As of June 30, 2020	As of December 31, 2019
Minimum lease payments receivable	\$ 1,853,967	\$ 1,884,921
Estimated residual values	394,195	394,195
Less: Unearned income and allowance for credit loss	(1,584,334)	(1,618,252)
Net investment in direct financing leases	 663,828	 660,864
Other financing leases (net of allowance for credit loss)	1,414,381	1,399,438
Total investment in financing leases	\$ 2,078,209	\$ 2,060,302

#### Rent Deferrals

Due to the COVID-19 pandemic and its impact on our tenants' business during the first six months of 2020, we agreed to defer approximately 2% of our rent. The amount of this deferral is approximately \$7 million as of June 30, 2020. Pursuant to our agreements with the tenants, we expect such deferred rent to be paid over specified periods with interest. Although we expect similar deferrals for third quarter rent, all of our rent and interest is expected to be paid on time starting October 1, 2020.

#### Adeptus Health

Due to a decline in operating results of 20 freestanding emergency facilities and one acute care facility caused by a reduction in volumes from COVID-19 and other factors, we entered into agreements to sever the remaining leases with Adeptus Health, Inc. ("Adeptus"), despite being current on their rent obligations. As a result, we recorded an approximately \$20 million net charge, primarily all of which was for the write off of straight-line rent, partially offset by approximately \$9 million of proceeds received from a letter of credit in the first half of 2020. Additionally, we recorded a \$9.9 million real estate impairment charge on these severed facilities in the first half of 2020. At June 30, 2020, we no longer lease any properties to Adeptus and our net book value on those properties that were previously leased to Adeptus but are currently vacant approximates less than 1% of our total assets. At June 30, 2020, we believe our investment in these real estate assets are fully recoverable, but no assurances can be given that we will not have any impairment in future periods.

#### Alecto Facilities

At June 30, 2020, we lease one acute care facility to Alecto Healthcare Services LLC ("Alecto") and have a mortgage loan on a second property, representing less than 0.4% of our total assets. During the second quarter of 2020, we re-leased one acute care facility to West Virginia University and sold another facility previously leased to Alecto. In addition, we donated the Wheeling facility to a local municipality, resulting in a \$9.1 million real estate impairment charge in the first quarter of 2020.

#### Loans

The following is a summary of our loans (net of allowance for credit loss in 2020):

(in thousands)	As of June 30, 2020	As of December 31, 2019
Mortgage loans	\$ 1,339,258	\$ 1,275,022
Acquisition loans	330,792	123,893
Other loans	461,219	420,939
Total	\$ 2,131,269	\$ 1,819,854

The increase in acquisition loans relates to the \$205 million loan to the new joint venture described under "New Investments" in this same Note 3.

Other loans consist of loans to our tenants for working capital and other purposes and include our shareholder loan made to the joint venture with Primotop Holdings S.à.r.l. ("Primotop") in the amount of €290 million.

#### Concentrations of Credit Risk

We monitor concentration risk in several ways due to the nature of our real estate assets that are vital to the communities in which they are located and given our history of being able to replace inefficient operators of our facilities, if needed, with more effective operators:

- 1) Facility concentration At June 30, 2020, our largest investment in any single property approximated 3% of our total assets, similar to December 31, 2019.
- 2) Operator concentration For the six months ended June 30, 2020, revenue from Steward, Prospect, Circle, and Prime represented 31%, 13%, 12%, and 11% of our total revenues, respectively. In comparison, Steward and Prime represented 47% and 17%, respectively, of our total revenues for the first six months of 2019, while Prospect and Circle, collectively, represented less than 1%. At June 30, 2020 and December 31, 2019, our lease concentration in any one tenant relationship was less than 20% of our total assets.
- 3) Geographic concentration At June 30, 2020, investments in the U.S., Europe, and Australia represented approximately 70%, 25%, and 5%, respectively, of our total assets, compared to 74%, 20%, and 6% at December 31, 2019.
- 4) Facility type concentration For the six months ended June 30, 2020, approximately 88% of our revenues are from our general acute care facilities, while rehabilitation and long-term acute care facilities make up 9% and 3%, respectively. These percentages are similar to those for the first six months of 2019.

#### 4. Debt

The following is a summary of debt (dollar amounts in thousands):

	As of June 30, 2020	As of December 31, 2019		
Revolving credit facility	\$ 	\$	_	
Term loan	200,000		200,000	
British pound sterling term loan(A)	868,070		_	
Australian term loan facility(A)	828,360		842,520	
4.000% Senior Unsecured Notes due 2022(A)	561,700		560,650	
2.550% Senior Unsecured Notes due 2023(A)	496,040		530,280	
5.500% Senior Unsecured Notes due 2024	300,000		300,000	
6.375% Senior Unsecured Notes due 2024	500,000		500,000	
3.325% Senior Unsecured Notes due 2025(A)	561,700		560,650	
5.250% Senior Unsecured Notes due 2026	500,000		500,000	
5.000% Senior Unsecured Notes due 2027	1,400,000		1,400,000	
3.692% Senior Unsecured Notes due 2028(A)	744,060		795,420	
4.625% Senior Unsecured Notes due 2029	900,000		900,000	
	\$ 7,859,930	\$	7,089,520	
Debt issue costs and discount, net	(64,040)		(65,841)	
	\$ 7,795,890	\$	7,023,679	

(A) Non-U.S. dollar denominated debt that reflects the exchange rate at June 30, 2020 and December 31, 2019, respectively.

As of June 30, 2020, principal payments due on our debt (which exclude the effects of any discounts, premiums, or debt issue costs recorded) are as follows (in thousands):

2020	\$ _
2021	_
2022	761,700
2023	496,040
2024	1,628,360
Thereafter	4,973,830
Total	\$ 7,859,930

#### 2020 Activity

#### **British Pound Sterling Term Loan**

On January 6, 2020, we entered into a £700 million unsecured sterling-denominated term loan facility with Bank of America, N.A., as administrative agent, and a syndicate of financial institutions as the lender. The term loan facility matures on January 15, 2025. We used the proceeds under the facility to help finance our acquisition of the Circle transaction described in Note 3. The applicable margin under the term loan is adjustable based on a pricing grid from 0.85% to 1.65% dependent on our current credit rating. On March 4, 2020, we entered into an interest rate swap transaction (effective March 6, 2020) to fix the interest rate to approximately 0.70% for the duration of the loan. The current applicable margin for the pricing grid (which can vary based on our credit rating) is 1.25% for an all-in fixed rate of 1.95%.

#### 2019 Activity

#### **Australian Term Loan Facility**

On May 23, 2019, we entered into an AUD \$1.2 billion term loan facility agreement with Bank of America, N.A., as administrative agent, and a syndicate of financial institutions as the lender. The term loan facility matures on May 23, 2024. The interest rate under the term loan is adjustable based on a pricing grid from 0.85% to 1.65%, dependent on our current senior unsecured credit rating. On June 27, 2019, we entered into an interest rate swap transaction (effective July 3, 2019) to fix the interest rate to approximately 1.20% for the duration of the loan. The current applicable margin for the pricing grid (which can vary based on our credit rating) is 1.25% for an all-in fixed rate of 2.45%.

#### Covenants

Our debt facilities impose certain restrictions on us, including restrictions on our ability to: incur debts; create or incur liens; provide guarantees in respect of obligations of any other entity; make redemptions and repurchases of our capital stock; prepay, redeem, or repurchase debt; engage in mergers or consolidations; enter into affiliated transactions; dispose of real estate or other assets; and change our business. In addition, the credit agreements governing our revolving credit ("Credit Facility") limit the amount of dividends we can pay as a percentage of normalized adjusted funds from operations ("NAFFO"), as defined in the agreements, on a rolling four quarter basis. At June 30, 2020, the dividend restriction was 95% of NAFFO. The indentures governing our senior unsecured notes also limit the amount of dividends we can pay based on the sum of 95% of NAFFO, proceeds of equity issuances and certain other net cash proceeds. Finally, our senior unsecured notes require us to maintain total unencumbered assets (as defined in the related indenture) of not less than 150% of our unsecured indebtedness.

In addition to these restrictions, the Credit Facility contains customary financial and operating covenants, including covenants relating to our total leverage ratio, fixed charge coverage ratio, secured leverage ratio, consolidated adjusted net worth, unsecured leverage ratio, and unsecured interest coverage ratio. This Credit Facility also contains customary events of default, including among others, nonpayment of principal or interest, material inaccuracy of representations, and failure to comply with our covenants. If an event of default occurs and is continuing under the Credit Facility, the entire outstanding balance may become immediately due and payable. At June 30, 2020, we were in compliance with all such financial and operating covenants.

### 5. Common Stock/Partners' Capital

Medical Properties Trust, Inc.

In the first half of 2020, we sold 8.6 million shares of common stock under our at-the-market equity offering program, resulting in net proceeds of approximately \$170 million; while, in the first half of 2019, we sold 22.6 million shares of common stock under our at-the-market equity offering program, resulting in net proceeds of approximately \$399 million.

#### MPT Operating Partnership, L.P.

At June 30, 2020, the company has a 99.9% ownership interest in the Operating Partnership, with the remainder owned by two other partners, who are employees.

During the six months ended June 30, 2020 and 2019, the Operating Partnership issued approximately 8.6 million and 22.6 million units, respectively, in direct response to the common stock at-the-market offerings by Medical Properties Trust, Inc. during the same periods.

#### 6. Stock Awards

We adopted the 2019 Equity Incentive Plan (the "Equity Incentive Plan") during the second quarter of 2019, which authorizes the issuance of common stock options, restricted stock, restricted stock units, deferred stock units, stock appreciation rights, performance units, and other stock-based awards. The Equity Incentive Plan is administered by the Compensation Committee of the Board of Directors, and we have reserved 12.9 million shares of common stock for awards, out of which 8.4 million shares remain available for future stock awards as of June 30, 2020. Share-based compensation expense totaled \$22.2 million and \$13.0 million for the six months ended June 30, 2020 and 2019, respectively.

#### 7. Fair Value of Financial Instruments

We have various assets and liabilities that are considered financial instruments. We estimate that the carrying value of cash and cash equivalents and accounts payable and accrued expenses approximate their fair values. We estimate the fair value of our interest and rent receivables using Level 2 inputs such as discounting the estimated future cash flows using the current rates at which similar receivables would be made to others with similar credit ratings and for the same remaining maturities. The fair value of our mortgage loans and other loans are estimated by using Level 2 inputs such as discounting the estimated future cash flows using the current rates which similar loans would be made to borrowers with similar credit ratings and for the same remaining maturities. We determine the fair value of our senior unsecured notes using Level 2 inputs such as quotes from securities dealers and market makers. We estimate the fair value of our revolving credit facility and term loans using Level 2 inputs based on the present value of future payments, discounted at a rate which we consider appropriate for such debt.

Fair value estimates are made at a specific point in time, are subjective in nature, and involve uncertainties and matters of significant judgment. Settlement of such fair value amounts may not be a prudent management decision. The following table summarizes fair value estimates for our financial instruments (in thousands):

	As of June 30, 2020			As of December 31, 2019				
Asset (Liability)	Book Value			Fair Book Value Value				Fair Value
Interest and rent receivables	\$	41,321	\$	40,496	\$	31,357	\$	30,472
Loans(1)		1,811,269		1,879,511		1,704,854		1,742,153
Debt, net		(7,795,890)		(7,935,142)		(7,023,679)		(7,331,816)

(1) Excludes \$115 million of mortgage loans related to Ernest and the \$205 million acquisition loan to the new joint venture discussed in Note 3 that are recorded at fair value (which is equal to their book value) based on Level 2 inputs by discounting the estimated cash flows using the market rates which similar loans would be made to borrowers with similar credit ratings and the same remaining maturities.

#### 8. Earnings Per Share

Medical Properties Trust, Inc.

Our earnings per share were calculated based on the following (amounts in thousands):

	For the Three Months Ended June 30,						
		2020		2019			
Numerator:							
Net income	\$	109,652	\$	79,920			
Non-controlling interests' share in net income		(184)		(482)			
Participating securities' share in earnings		(487)		(446)			
Net income, less participating securities' share in earnings	\$	108,981	\$	78,992			
Denominator:							
Basic weighted-average common shares		527,781		394,574			
Dilutive potential common shares		1,099		1,118			
Dilutive weighted-average common shares		528,880		395,692			

	For the Six Months Ended June 30,						
		2020		2019			
Numerator:							
Net income	\$	190,809	\$	156,211			
Non-controlling interests' share in net income		(349)		(951)			
Participating securities' share in earnings		(951)		(922)			
Net income, less participating securities' share in earnings	\$	189,509	\$	154,338			
Denominator:							
Basic weighted-average common shares		524,428		387,563			
Dilutive potential common shares		1,102		1,120			
Dilutive weighted-average common shares		525,530		388,683			

#### MPT Operating Partnership, L.P.

Our earnings per common unit were calculated based on the following (amounts in thousands):

	For the Three Months Ended June 30,				
	2020		2019		
Numerator:					
Net income	\$ 109,652	\$	79,920		
Non-controlling interests' share in net income	(184)		(482)		
Participating securities' share in earnings	(487)		(446)		
Net income, less participating securities' share in earnings	\$ 108,981	\$	78,992		
Denominator:	 				
Basic weighted-average units	527,781		394,574		
Dilutive potential units	1,099		1,118		
Diluted weighted-average units	 528,880		395,692		
	For the Siz Ended J				
	 2020	une 50,	2019		
Numerator:					
Net income	\$ 190,809	\$	156,211		
Non-controlling interests' share in net income	(349)		(951)		
Participating securities' share in earnings	(951)		(922)		
Net income, less participating securities' share in earnings	\$ 189,509	\$	154,338		
Denominator:					
Basic weighted-average units	524,428		387,563		
Dilutive potential units	1,102		1,120		
Diluted weighted-average units	 525,530		388,683		

## 9. Commitments and Contingencies

#### Commitments

We entered into agreements to invest \$300 million for the purpose of acquiring the real estate of St. Francis Medical Center in Lynwood, California. This acute care facility is expected to be leased to Prime.

#### Contingencies

We are a party to various legal proceedings incidental to our business. In the opinion of management, after consultation with legal counsel, the ultimate liability, if any, with respect to those proceedings is not presently expected to materially affect our financial position, results of operations, or cash flows.

#### 10. Subsequent Events

On July 7, 2020, we acquired the fee simple real estate of two general acute care hospitals located in the Salt Lake City, Utah area, Davis Hospital & Medical Center and Jordan Valley Medical Center, in exchange for the reduction of the mortgage loans made to Steward and additional cash consideration of \$200 million based on their relative fair value. The approximate \$950 million investment in the Davis and Jordan Valley facilities is now subject to the Steward master lease.

Our five San Antonio, Texas free standing emergency facilities (with a total investment of approximately \$30 million) have been re-leased to Methodist Healthcare System of San Antonio, a joint venture between HCA Healthcare and Methodist Healthcare Ministries of South Texas on July 24, 2020, pursuant to a long-term master lease. These properties were previously leased to Steward.

Other acquisitions subsequent to June 30, 2020 included one inpatient rehabilitation hospital and one general acute care hospital. The inpatient rehabilitation facility, located in Dahlen, Germany, was acquired on August 5 for €12.5 million and is leased to MEDIAN Kliniken S.à.r.l. pursuant to the existing master lease. The general acute care facility, located in Darlington, United Kingdom, was acquired on August 7 for £29.4 million and is leased to Circle pursuant to the existing master lease.

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

The following discussion and analysis of the consolidated financial condition and consolidated results of operations are presented on a combined basis for Medical Properties Trust and MPT Operating Partnership, L.P. as there are no material differences between these two entities.

The following discussion and analysis of the consolidated financial condition and consolidated results of operations should be read together with the condensed consolidated financial statements and notes thereto contained in this Form 10-Q and the consolidated financial statements and notes thereto contained in our Annual Report on Form 10-K for the year ended December 31, 2019.

During the first six months of 2020, COVID-19's impact on our business was limited as we received approximately 98% of our rent and interest payments in a timely manner. For the 2% not collected, we have agreements in place to collect such deferred amounts plus interest. Our tenants were impacted by the governmental mandates to defer elective surgeries, the takeover of certain facilities by the government in certain countries like the United Kingdom, and the overall downturn in the economy in general. We currently expect to receive a similar amount of our rent and interest in the third quarter as we did in the second quarter but then expect all rent and interest to be collected timely starting in October 2020. However, no assurances can be made that if the pandemic continues for an extended period of time that our rent and interest payments will not be delayed into the future until our tenants can recover

#### Forward-Looking Statements.

This quarterly report on Form 10-Q contains certain "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. Such forward-looking statements involve known and unknown risks, uncertainties, and other factors that may cause our actual results or future performance, achievements or transactions or events to be materially different from those expressed or implied by such forward-looking statements, including, but not limited to, the risks described in our Annual Report on Form 10-K and as updated in our quarterly reports on Form 10-Q for future periods, and current reports on Form 8-K as we file them with the SEC under the Securities Exchange Act of 1934, as amended. Such factors include, among others, the following:

- the political, economic, business, real estate, and other market conditions of the U.S. (both national and local), Europe (in particular Germany, the United Kingdom, Spain, Italy, Portugal, and Switzerland), Australia, and other foreign jurisdictions;
- the risk that a condition to closing under the agreements governing any or all of our outstanding transactions that have not closed as of the date hereof may not be satisfied;
- the possibility that the anticipated benefits from any or all of the transactions we enter into will take longer to realize than expected or will not be realized at all:
- the competitive environment in which we operate;
- the execution of our business plan;
- financing risks;
- acquisition and development risks;
- potential environmental contingencies and other liabilities;
- adverse developments affecting the financial health of one or more of our tenants, including insolvency;
- other factors affecting the real estate industry generally or the healthcare real estate industry in particular;
- our ability to maintain MPT's status as a REIT for federal and state income tax purposes;
- our ability to attract and retain qualified personnel;
- the impact of COVID-19 on our business, our joint ventures, and the business of our tenants/borrowers and the economy in general and other factors that may affect our business, our joint ventures, or that of our tenants/borrowers that are beyond our control, including natural disasters, other health crises, or pandemics along with each individual government's actions to such matters;
- changes in foreign currency exchange rates;
- changes in federal, state, or local tax laws in the U.S., Europe, Australia, or other jurisdictions in which we may own healthcare facilities or transact business;
- · healthcare and other regulatory requirements of the U.S., Europe, Australia, and other foreign countries; and

- the political, economic, business, real estate, and other market conditions of the U.S., Europe, Australia, and other foreign jurisdictions in which we may own healthcare facilities or transact business, which may have a negative effect on the following, among other things:
  - the financial condition of our tenants, our lenders, or institutions that hold our cash balances, which may expose us to increased risks of default by these parties;
  - our ability to obtain equity or debt financing on attractive terms or at all, which may adversely impact our ability to pursue acquisition and development opportunities, refinance existing debt, and our future interest expense; and
  - the value of our real estate assets, which may limit our ability to dispose of assets at attractive prices or obtain or maintain debt financing secured by our properties or on an unsecured basis.

#### **Key Factors that May Affect Our Operations**

Our revenue is derived from rents we earn pursuant to the lease agreements with our tenants, from interest income from loans to our tenants and other facility owners and from profits or equity interests in certain of our tenants' operations. Our tenants operate in the healthcare industry, generally providing medical, surgical, and rehabilitative care to patients. The capacity of our tenants to pay our rents and interest is dependent upon their ability to conduct their operations at profitable levels. We believe that the business environment of the industry segments in which our tenants operate is generally positive for efficient operators. However, our tenants' operations are subject to economic, regulatory, market, and other conditions (such as the impact caused by COVID-19 in the form of requirements to defer elective surgeries, government takeover of facilities in certain of our international locations, and overall negative impact to the economy in general) that may affect their profitability, which could impact our results. Accordingly, we monitor certain key factors, changes to which we believe may provide early indications of conditions that may affect the level of risk in our portfolio.

Key factors that we consider in underwriting prospective tenants and borrowers and in monitoring the performance of existing tenants and borrowers include the following:

- admission levels and surgery/procedure/diagnosis volumes by type;
- the current, historical, and prospective operating margins (measured by earnings before interest, taxes, depreciation, amortization, and facility rent) of each tenant or borrower and at each facility;
- the ratio of our tenants' or borrowers' operating earnings both to facility rent and to facility rent plus other fixed costs, including debt costs;
- changes in revenue sources of our tenants' or borrowers' revenue, including the relative mix of public payors (including Medicare,
  Medicaid/MediCal, managed care in the U.S., pension funds in Germany, and National Health Service in the United Kingdom) and private
  payors (including commercial insurance and private pay patients);
- trends in tenants' cash collections, including comparison to recorded net patient service revenues;
- tenants' free cash flows;
- the effect of evolving healthcare legislation and other regulations on our tenants' or borrowers' profitability and liquidity; and
- the competition and demographics of the local and surrounding areas in which the tenants or borrowers operate.

Certain business factors, in addition to those described above that directly affect our tenants and borrowers, will likely materially influence our future results of operations. These factors include:

- trends in the cost and availability of capital, including market interest rates, that our prospective tenants may use for their real estate assets instead of financing their real estate assets through lease structures;
- changes in healthcare regulations that may limit the opportunities for physicians to participate in the ownership of healthcare providers and healthcare real estate;
- reductions in reimbursements from Medicare, state healthcare programs, and commercial insurance providers that may reduce our tenants' or borrowers' profitability and our lease rates;
- · competition from other financing sources; and
- the ability of our tenants and borrowers to access funds in the credit markets.

#### CRITICAL ACCOUNTING POLICIES

Refer to our 2019 Annual Report on Form 10-K for a discussion of our critical accounting policies, which include revenue recognition, investments in real estate, purchase price allocation, loans, losses from rent and interest receivables, stock-based

compensation, our fair value option election, and our accounting policy on consolidation. During the six months ended June 30, 2020, there were no material changes to these policies except for those described in <u>Note 2</u> to our condensed consolidated financial statements.

#### Overview

We are a self-advised REIT focused on investing in and owning net-leased healthcare facilities across the U.S. and selectively in foreign jurisdictions. We have operated as a REIT since April 6, 2004, and accordingly, elected REIT status upon the filing of our calendar year 2004 federal income tax return. Medical Properties Trust, Inc. was incorporated under Maryland law on August 27, 2003, and MPT Operating Partnership, L.P. was formed under Delaware law on September 10, 2003. We conduct substantially all of our business through MPT Operating Partnership, L.P. We acquire and develop healthcare facilities and lease the facilities to healthcare operating companies under long-term net leases, which require the tenant to bear most of the costs associated with the property. We also make mortgage loans to healthcare operators collateralized by their real estate assets. In addition, we selectively make loans to certain of our operators through our taxable REIT subsidiaries, the proceeds of which are typically used for acquisitions and working capital. Finally, from time to time, we acquire a profits or other equity interest in our tenants that gives us a right to share in such tenant's profits and losses.

At June 30, 2020, our portfolio consisted of 387 properties leased or loaned to 44 operators, of which two are under development and 13 are in the form of mortgage loans.

Our investments in healthcare real estate, including mortgage and other loans, as well as any equity investments in our tenants are considered a single reportable segment. At June 30, 2020, all of our investments are located in the U.S., Europe, and Australia. Our total assets are made up of the following (dollars in thousands):

,	A	as of June 30, 2020	% of Total	Α	As of December 31, 2019	% of Total
Real estate owned (gross)	\$	12,041,395	77.9%	\$	9,994,844	69.1%
Mortgage loans		1,339,258	8.7%		1,275,022	8.8%
Other loans		792,011	5.1%		544,832	3.8%
Construction in progress		27,674	0.2%		168,212	1.2%
Cash and cash equivalents		374,962	2.4%		1,462,286	10.1%
Equity investments		841,098	5.4%		926,990	6.4%
Other		31,672	0.3%		95,145	0.6%
Total assets	\$	15,448,070	100.0%	\$	14,467,331	100.0%

#### **Additional Concentration Details**

On a pro forma gross asset basis (as defined in <u>"Reconciliation of Non-GAAP Financial Measures"</u>), our concentration as of June 30, 2020 as compared to December 31, 2019 is as follows (dollars in thousands):

Total Pro Forma Gross Assets by Operator

Percentage of Total Pro Forma Gross Assets  24.3%  12.4%  9.1%	2,152,951	13.0%
12.4%	2,152,951	
		13.0%
0.10/		
9.1%	1,563,642	9.5%
8.4%	1,144,705	6.9%
7.0%	1,202,319	7.3%
34.7%	5,509,952	33.4%
4.1%	903,543	5.4%
100.00	\$ 16,529,274	100.0%
	4.1% 100.0%	

	 As of June	30, 2020	_	As of December 31, 2019			
U.S. States and Other Countries	 tal Pro Forma Gross Assets	Percentage of Total Pro Forma Gross Assets	7	Total Pro Forma Gross Assets	Percentage of Total Pro Forma Gross Assets		
California	\$ 1,650,520	9.6%	\$	1,298,244	7.9%		
Texas	1,622,600	9.4%		1,390,835	8.4%		
Massachusetts	1,497,182	8.7%		1,497,182	9.1%		
Utah	1,286,423	7.4%		1,087,743	6.6%		
Pennsylvania	858,853	5.0%		905,887	5.5%		
All other states	4,007,474	23.2%		4,022,909	24.2%		
Other domestic assets	623,716	3.6%		798,990	4.8%		
Total U.S.	\$ 11,546,768	66.9%	\$	11,001,790	66.5%		
United Kingdom	\$ 2,570,106	14.9%	\$	2,617,158	15.8%		
Germany	1,225,378	7.1%		1,117,539	6.8%		
Australia	898,328	5.2%		897,915	5.4%		
Switzerland	611,796	3.5%		505,172	3.1%		
Spain	202,042	1.2%		159,451	1.0%		
All other countries	131,052	0.7%		125,696	0.8%		
Other international assets	92,402	0.5%		104,553	0.6%		
Total international	\$ 5,731,104	33.1%	\$	5,527,484	33.5%		
Grand total	\$ 17,277,872	100.0%	\$	16,529,274	100.0%		

On an individual property basis, our largest investment in any single property approximated 3% of our total pro forma gross assets as of June 30, 2020.

On an adjusted revenue basis (as defined in the "Reconciliation of Non-GAAP Financial Measures" section of Item 2 of this Quarterly Report on Form 10-Q), concentration for the six months ended June 30, 2020 as compared to the prior year is as follows (dollars in thousands):

Total Adjusted Revenue by Operator

	For the Six Months Ended June 30,								
		202	0	2019					
<u>Operators</u>	To	otal Adjusted Revenue	Percentage of Total Adjusted Revenue	Total Adjusted Revenue	Percentage of Total Adjusted Revenue				
Steward	\$	179,780	28.2%	\$ 176,290	43.0%				
Prospect		76,500	12.0%	_	—%				
Circle		68,030	10.7%	2,597	0.6%				
Prime		64,381	10.1%	63,996	15.6%				
LifePoint		53,198	8.4%	22,960	5.6%				
Other operators		194,575	30.6%	144,047	35.2%				
Total	\$	636,464	100.0%	\$ 409,890	100.0%				

		For the Six Months Ended June 30,											
			202	10		2019							
U.S. States and Other Countries	•		Adjusted evenue	Percentage of Total Adjusted Revenue		Total Adjusted Revenue	Percentage of Total Adjusted Revenue						
California		\$	70,339	11.1%	\$	32,362	7.9%						
Massachusetts			69,946	11.0%		68,553	16.7%						
Texas			48,674	7.7%		58,504	14.3%						
Utah			45,705	7.2%		43,409	10.6%						
Pennsylvania			38,606	6.1%		7,577	1.9%						
All other states			187,453	29.3%		139,071	33.9%						
Total U.S.		\$	460,723	72.4%	\$	349,476	85.3%						
United Kingdom		\$	80,891	12.7%	\$	2,597	0.6%						
Germany			47,445	7.5%		48,326	11.8%						
All other countries			47,405	7.4%		9,491	2.3%						
Total international		\$	175,741	27.6%	\$	60,414	14.7%						
Grand total		\$	636,464	100.0%	\$	409,890	100.0%						

Total Adjusted Revenue by Facility Type

	_		For the Six Months Ended June 30,						
	_	2020			2019				
Facility Types		Total Adjusted Revenue	Percentage of Total Adjusted Revenue	Т	otal Adjusted Revenue	Percentage of Total Adjusted Revenue			
General acute care hospitals	\$	538,040	84.5%	\$	326,677	79.7%			
Rehabilitation hospitals		81,295	12.8%		68,555	16.7%			
Long-term acute care hospitals		17,129	2.7%		14,658	3.6%			
Total	\$	636,464	100.0%	\$	409,890	100.0%			

#### **Results of Operations**

#### Three Months Ended June 30, 2020 Compared to June 30, 2019

Net income for the three months ended June 30, 2020, was \$109.5 million, compared to \$79.4 million for the three months ended June 30, 2019. This increase is due to incremental revenue from new investments made in 2019 and additional revenue from the Circle transaction in January 2020, partially offset by higher interest (from additional debt to partially finance these new investments) and depreciation expense along with additional general and administrative costs due to the growth of the company. Funds from operations ("FFO"), after adjusting for certain items (as more fully described in the "Reconciliation of Non-GAAP Financial Measures"), was \$199.6 million for the 2020 second quarter as compared to \$120.9 million for the 2019 second quarter. Similar to net income, this increase in FFO is primarily due to incremental revenue from new investments in 2019 and early 2020.

A comparison of revenues for the three month periods ended June 30, 2020 and 2019 is as follows (dollar amounts in thousands):

	2020	% of Total	2019	% of Total	Year over Year Change
Rent billed	\$ 173,557	59.5%	\$ 110,882	57.6%	56.5%
Straight-line rent	21,151	7.2%	25,136	13.1%	-15.9%
Income from financing leases	52,489	18.0%	17,386	9.0%	201.9%
Interest and other income	44,645	15.3%	39,145	20.3%	14.1%
Total revenues	\$ 291,842	100.0%	\$ 192,549	100.0%	51.6%

Our total revenue for the 2020 second quarter is up \$99.3 million, or 52%, over the prior year. This increase is made up of the following:

• Operating lease revenue (includes rent billed and straight-line rent) – up \$58.7 million over the prior year of which approximately \$75.3 million is incremental revenue from acquisitions made post June 30, 2019 (\$33.8 million of which

relates to the Circle acquisition in January 2020 as described in Note 3 to the condensed consolidated financial statements and \$15.1 million of which relates to the LifePoint acquisition completed in the fourth quarter of 2019) and \$2.7 million is from the commencement of rent on the Idaho Falls development property. This increase is partially offset by an \$11.9 million write-off of straight-line rent on Adeptus facilities (as described in Note 3), an approximate \$5 million write-off of straight-line rent on one Steward facility that was sold during the second quarter, and approximately \$0.4 million from unfavorable foreign currency fluctuations.

- As discussed in <u>Note 3</u> to Item 1 of the Form 10-Q, we received final regulatory approval of the Circle/BMI lease variations and amended such leases accordingly on June 16, 2020. We expect our operating lease revenue from these leases to improve by approximately \$8 million in the third quarter versus the 2020 second quarter with a full quarter of such lease variations in place.
- Income from financing leases up \$35.1 million due to \$34.8 million of revenue from the Prospect acquisition in the 2019 third quarter.
- Interest and other income (including \$5.2 million in the 2020 second quarter to gross up the tenant reimbursement of certain expenses) up \$5.5 million from the prior year due to approximately \$6 million of incremental revenue earned on new loan investments, including \$3.1 million related to Prospect loans made in August 2019 and \$2.1 million earned on the loan made to the new international joint venture in May 2020 (see Note 3 for additional details), and \$0.9 million from annual escalations in interest rates in accordance with loan provisions. This increase is partially offset by \$0.1 million from unfavorable foreign currency fluctuations.

Interest expense for the quarters ended June 30, 2020 and 2019 totaled \$80.4 million and \$52.3 million, respectively. This increase is primarily related to new debt issuances post June 30, 2019 to fund new investments, including the £700 million term loan issued in January 2020, the £1 billion senior unsecured notes issued in December 2019, and the \$900 million of senior unsecured notes issued in July 2019. In the 2019 second quarter, we incurred \$0.9 million of accelerated commitment fee amortization expense associated with the new Australian term loan facility. No such expense was incurred in the 2020 second quarter. Our weighted-average interest rate was 3.9% for the three months ended June 30, 2020, as compared to 4.7% in the same period in 2019.

Real estate depreciation and amortization during the second quarter of 2020 increased to \$61.5 million from \$34.0 million in 2019 due to new investments made after June 30, 2019.

Property-related expenses totaled \$10.0 million and \$8.3 million for the quarters ended June 30, 2020 and 2019, respectively. This increase is primarily due to property taxes incurred related to the severed Adeptus properties as described in <a href="Note 3">Note 3</a> to the condensed consolidated financial statements.

As a percentage of revenue, general and administrative expenses represent 11.0% for the 2020 second quarter, which is lower than the 11.6% in the prior year. On a dollar basis, general and administrative expenses totaled \$32.0 million for the 2020 second quarter, which is a \$9.7 million increase from the prior year second quarter. Of this increase, \$6.7 million relates to compensation primarily related to higher stock compensation expense from our performance-based awards. Given our strong performance in 2019 including a 39% total shareholder return and significant growth from \$4.5 billion of new investments, along with our largest ever one-time acquisition in January 2020, we believe it is more likely that certain performance awards will be earned and have adjusted our stock compensation expense accordingly. The balance of the increase is primarily related to other corporate expenses, which are higher due to the growth of the company.

During the three months ended June 30, 2020, we disposed of three facilities and two ancillary properties resulting in a net loss of \$3.1 million.

Earnings from equity interests was \$5.3 million for the quarter ended June 30, 2020, up \$0.9 million from the same period in 2019 due to our investments in Infracore and HM Hospitales made in the second and fourth quarters of 2019, respectively.

Other income of \$4.3 million for the second quarter of 2020 was generated primarily from non-cash fair value adjustments related to the marking of our investment in Aevis Victoria SA stock to market. We acquired this stock as part of our overall Switzerland investment in May 2019.

Income tax expense typically includes U.S. federal and state income taxes on our domestic TRS entities, as well as non-U.S. income based or withholding taxes on certain investments located in jurisdictions outside the U.S. The \$4.8 million income tax expense for the three months ended June 30, 2020 is primarily from our investments in the United Kingdom, particularly the recent Circle/BMI transaction. In comparison, we recorded a \$0.3 million income tax benefit in the second quarter of 2019 from one-time tax benefits recognized by our domestic TRS.

We utilize the asset and liability method of accounting for income taxes. Deferred tax assets are recorded to the extent we believe these assets will more likely than not be realized. In making such determination, all available positive and negative evidence is considered, including scheduled reversals of deferred tax liabilities, projected future taxable income, tax planning strategies, and recent financial performance. Based upon our review of all positive and negative evidence, including our three-year cumulative pre-tax book loss position in certain entities, we concluded that a valuation allowance of \$29.9 million should be reflected against certain of our international and domestic net deferred tax assets at June 30, 2020. In the future, if we determine that it is more likely than not that we will realize our net deferred tax assets, we will reverse the applicable portion of the valuation allowance, recognize an income tax benefit in the period in which such determination is made, and incur higher income taxes in future periods as income is earned.

#### Six Months Ended June 30, 2020 Compared to June 30, 2019

Net income for the six months ended June 30, 2020, was \$190.5 million, compared to \$155.3 million for the six months ended June 30, 2019. This increase is due to incremental revenue from new investments made in 2019 and additional revenue from the Circle transaction in January 2020, partially offset by \$58 million in increased interest expense from new debt incurred to partially fund those new investments, \$55 million of higher depreciation expense, and about \$30 million in non-cash impairment charges and fair value adjustments on equity securities along with higher general and administrative and income tax expenses due to the growth of the company. Funds from operations ("FFO"), after adjusting for certain items (as more fully described in the "Reconciliation of Non-GAAP Financial Measures"), was \$390.8 million for the first six months of 2020 as compared to \$238.7 million for the first six months of 2019. Similar to net income, this increase in FFO is primarily due to incremental revenue from new investments post June 30, 2019.

A comparison of revenues for the six month periods ended June 30, 2020 and 2019 is as follows (dollar amounts in thousands):

	 2020	% of Total	2019	% of Total	Year over Year Change
Rent billed	\$ 345,324	58.9%	\$ 219,480	58.8%	57.3%
Straight-line rent	52,572	9.0%	45,787	12.3%	14.8%
Income from financing leases	104,925	17.9%	34,666	9.3%	202.7%
Interest and other income	83,153	14.2%	73,070	19.6%	13.8%
Total revenues	\$ 585,974	100.0%	\$ 373,003	100.0%	57.1%

Our total revenue for the first six months of 2020 is up \$213.0 million, or 57%, from the prior year. This increase is made up of the following:

- Operating lease revenue (includes rent billed and straight-line rent) up \$132.6 million over the prior year of which approximately \$152.7 million is incremental revenue from acquisitions made post June 30, 2019 (\$64.2 million of which relates to the Circle acquisition as described in Note 3 to the condensed consolidated financial statements and \$30.2 million of which relates to the LifePoint acquisition completed in the fourth quarter of 2019) and \$4.7 million is from the commencement of rent on the Idaho Falls development property. This increase is partially offset by more straight-line rent write-offs than in 2019 (as described in Note 3 to the condensed consolidated financial statements) and approximately \$0.5 million from unfavorable foreign currency fluctuations.
  - As discussed in Note 3 to the condensed consolidated financial statements, we received final regulatory approval of the Circle/BMI lease variations and amended such leases accordingly on June 16, 2020. If such amended leases were in place at the closing date of our acquisition on January 8, 2020, lease revenue would have been approximately \$17 million higher in the first half of 2020.
- Income from financing leases up \$70.3 million due to \$69.6 million of revenue from the Prospect acquisition in the 2019 third quarter.
- Interest and other income up \$10.1 million from the prior year due to the following:
  - Interest from loans up \$10.9 million over the prior year due to approximately \$9.3 million of incremental revenue earned on new loan investments, including \$6.1 million related to Prospect loans made in August 2019 and \$2.1 million earned on the loan made to the new international joint venture in May 2020 (see Note 3 for additional details), and \$1.9 million from our annual escalations in interest rates in accordance with loan provisions. This increase is partially offset by \$0.2 million from unfavorable foreign currency fluctuations.
  - Other income down \$0.8 million from the prior year due to an approximate \$1 million write-off of straight-line rent related to ground leases on certain Adeptus facilities in the first quarter of 2020 (as described in <a href="Note 3">Note 3</a> to condensed consolidated financial statements).

Interest expense for the six months ended June 30, 2020 and 2019, totaled \$161.3 million and \$102.9 million, respectively. This increase is primarily related to new debt issuances post June 30, 2019 including the £700 million term loan issued in January 2020, the £1 billion senior unsecured notes issued in December 2019, and the \$900 million of senior unsecured notes issued in July 2019. In addition, we incurred \$0.6 million of accelerated commitment fee amortization expense associated with our GBP term loan facility in the first six months of 2020 and \$0.9 million of similar expense associated with our Australian term loan facility in the first six months of 2019. Our weighted-average interest rate was 4.0% for the six months ended June 30, 2020, as compared to 4.7% in the same period in 2019.

Real estate depreciation and amortization during the first six months of 2020 increased to \$122.4 million from \$67.3 million in the same period of 2019, due to new investments made after June 30, 2019.

Property-related expenses totaled \$15.6 million and \$11.4 million for the six months ended June 30, 2020 and 2019, respectively. This increase is primarily due to property taxes and other expenses incurred on the severed Adeptus properties in the first half of 2020 and the two Alecto properties that were disposed of in the second quarter of 2020 (as described in <u>Note 3</u> to the condensed consolidated financial statements).

As a percentage of revenue, general and administrative expenses represent 11.2% for the first six months of 2020, a decline from the 12.3% in the prior year same period. On a dollar basis, general and administrative expenses totaled \$65.4 million for the first six months of 2020, which is a \$19.7 million increase from the same period of 2019. Of this increase, \$12.6 million relates to compensation primarily related to higher stock compensation expense from our performance-based awards. Given our strong performance in 2019 including a 39% total shareholder return and significant growth from \$4.5 billion of new investments, along with our largest ever one-time acquisition in January 2020, we believe it is more likely that certain performance awards will be earned and have adjusted our stock compensation expense accordingly. The balance of the increase is primarily related to other corporate expenses, which are higher due to the growth of the company.

During the six months ended June 30, 2020, we disposed of five facilities and four ancillary properties resulting in a net loss of \$1.8 million. In addition, we made a \$19.0 million adjustment to lower the carrying value of the real estate on certain Adeptus properties and one Alecto facility in the first quarter of 2020 (see Note 3 to condensed consolidated financial statements for further details).

Earnings from equity interests was \$9.4 million for the first six months of 2020, up \$1.2 million from the same period of 2019, primarily due to our investments in Infracore and HM Hospitales made in the second and fourth quarters of 2019, respectively.

Other expense of \$9.7 million for the first six months of 2020 represents non-cash fair value adjustments primarily related to the marking of our investment in Aevis Victoria SA stock to market. This stock, like most stocks, declined during the first half of 2020 due to the COVID-19 pandemic. We acquired this stock as part of our overall Switzerland investment in May 2019.

Income tax expense typically includes U.S. federal and state income taxes on our TRS entities, as well as non-U.S. income based or withholding taxes on certain investments located in jurisdictions outside the U.S. The \$8.8 million income tax expense for the six months ended June 30, 2020, is primarily from our investments in the United Kingdom, particularly the recent Circle/BMI transaction. In comparison, we recorded a \$2.6 million income tax benefit in the first six months of 2019 from non-recurring tax benefits recognized by our domestic TRS.

We utilize the asset and liability method of accounting for income taxes. Deferred tax assets are recorded to the extent we believe these assets will more likely than not be realized. In making such determination, all available positive and negative evidence is considered, including scheduled reversals of deferred tax liabilities, projected future taxable income, tax planning strategies, and recent financial performance. Based upon our review of all positive and negative evidence, including our three-year cumulative pre-tax book loss position in certain entities, we concluded that a valuation allowance of \$29.9 million should be reflected against certain of our international and domestic net deferred tax assets at June 30, 2020. In the future, if we determine that it is more likely than not that we will realize our net deferred tax assets, we will reverse the applicable portion of the valuation allowance, recognize an income tax benefit in the period in which such determination is made, and incur higher income taxes in future periods as income is earned.

#### **Reconciliation of Non-GAAP Financial Measures**

Investors and analysts following the real estate industry utilize funds from operations, or FFO, as a supplemental performance measure. FFO, reflecting the assumption that real estate asset values rise or fall with market conditions, principally adjusts for the effects of GAAP depreciation and amortization of real estate assets, which assumes that the value of real estate diminishes predictably over time. We compute FFO in accordance with the definition provided by the National Association of Real Estate Investment Trusts, or Nareit, which represents net income (loss) (computed in accordance with GAAP), excluding gains (losses) on sales of real estate and impairment charges on real estate assets, plus real estate depreciation and after adjustments for unconsolidated partnerships and joint ventures.

In addition to presenting FFO in accordance with the Nareit definition, we also disclose normalized FFO, which adjusts FFO for items that relate to unanticipated or non-core events or activities or accounting changes that, if not noted, would make comparison to prior period results and market expectations less meaningful to investors and analysts.

We believe that the use of FFO, combined with the required GAAP presentations, improves the understanding of our operating results among investors and the use of normalized FFO makes comparisons of our operating results with prior periods and other companies more meaningful. While FFO and normalized FFO are relevant and widely used supplemental measures of operating and financial performance of REITs, they should not be viewed as a substitute measure of our operating performance since the measures do not reflect either depreciation and amortization costs or the level of capital expenditures and leasing costs necessary to maintain the operating performance of our properties, which can be significant economic costs that could materially impact our results of operations. FFO and normalized FFO should not be considered an alternative to net income (loss) (computed in accordance with GAAP) as indicators of our financial performance or to cash flow from operating activities (computed in accordance with GAAP) as an indicator of our liquidity.

The following table presents a reconciliation of net income attributable to MPT common stockholders to FFO for the three and six months ended June 30, 2020 and 2019 (in thousands, except per share data):

	For the Three Months Ended					For the Six Months Ended			
	Jur	ne 30, 2020	June 30, 2019		June 30, 2020		Ju	ine 30, 2019	
FFO information:									
Net income attributable to MPT common stockholders	\$	109,468	\$	79,438	\$	190,460	\$	155,260	
Participating securities' share in earnings		(487)		(446)		(951)		(922)	
Net income, less participating securities' share in earnings	\$	108,981	\$	78,992	\$	189,509	\$	154,338	
Depreciation and amortization		71,823		40,407		142,325		80,261	
Loss on sale of real estate		3,101		147		1,776		147	
Real estate impairment charges				<u> </u>		19,006		<u> </u>	
Funds from operations	\$	183,905	\$	119,546	\$	352,616	\$	234,746	
Write-off of straight-line rent and other, net of tax		19,241		406		26,958		3,002	
Non-cash fair value adjustments		(3,590)		_		10,605		_	
Unutilized financing fees		_		914		611		914	
Normalized funds from operations	\$	199,556	\$	120,866	\$	390,790	\$	238,662	
Per diluted share data:									
Net income, less participating securities' share in earnings	\$	0.21	\$	0.20	\$	0.36	\$	0.40	
Depreciation and amortization		0.14		0.10		0.27		0.20	
Loss on sale of real estate		_		_				_	
Real estate impairment charges		_		_		0.04		_	
Funds from operations	\$	0.35	\$	0.30	\$	0.67	\$	0.60	
Write-off of straight-line rent and other, net of tax		0.03		_		0.05		0.01	
Non-cash fair value adjustments		_		_		0.02		_	
Unutilized financing fees				0.01		<u> </u>		<u> </u>	
Normalized funds from operations	\$	0.38	\$	0.31	\$	0.74	\$	0.61	

#### Pro Forma Gross Assets

Pro forma gross assets is total assets before accumulated depreciation/amortization (adjusted for our unconsolidated joint ventures) and assumes all real estate binding commitments on new investments and unfunded amounts on development deals and commenced capital improvement projects as of the applicable reporting periods are fully funded, and assumes cash on hand is used in these transactions. We believe total pro forma gross assets is useful to investors as it provides a more current view of our portfolio and allows for a better understanding of our concentration levels as our binding commitments close and our other commitments are fully funded. The following table presents a reconciliation of total assets to total pro forma gross assets (in thousands):

	As of June 30, 2020	D	As of ecember 31, 2019
Total assets	\$ 15,448,070	\$	14,467,331
Add:			
Binding real estate commitments on new investments(1)	514,042		1,988,550
Unfunded amounts on development deals and commenced			
capital improvement projects(2)	154,760		163,370
Accumulated depreciation and amortization	684,444		570,042
Incremental gross assets of our joint ventures(3)	851,518		563,911
Proceeds from new debt subsequent to period-end	_		927,990
Less:			
Cash used for funding the transactions above			
(including proceeds from new debt subsequent to period-end)	(374,962)		(2,151,920)
Total pro forma gross assets	\$ 17,277,872	\$	16,529,274

- 1) The 2020 column reflects our commitment at June 30, 2020 to acquire a facility in the U.S. and a facility in Germany, along with an incremental investment to acquire the fee simple interest of two facilities in the U.S. previously subject to a mortgage loan. The 2019 column reflects the acquisition of 30 facilities in the United Kingdom of January 8, 2020.
- (2) Includes \$47.8 million and \$41.7 million of unfunded amounts on ongoing development projects and \$107.0 million and \$121.7 million of unfunded amounts on capital improvement and development projects that have commenced rent, as of June 30, 2020 and December 31, 2019, respectively.
- (3) Adjustment to reflect our share of our joint ventures' gross assets.

#### Adjusted revenue

Adjusted revenues are total revenues adjusted for our pro rata portion of similar revenues in our real estate joint venture arrangements. We believe adjusted revenue is useful to investors as it provides a more complete view of revenue across all of our investments and allows for better understanding of our revenue concentration. The following table presents a reconciliation of total revenues to total adjusted revenues (in thousands):

	 For the Six Months Ended June 30,				
	2020		2019		
Total revenues	\$ 585,974	\$	373,003		
Revenue from real estate properties owned through joint venture arrangements	50,490		36,887		
Total adjusted revenue	\$ 636,464	\$	409,890		

#### LIQUIDITY AND CAPITAL RESOURCES

#### 2020 Cash Flow Activity

During the first half of 2020, we generated \$289 million of cash flows from operating activities (which did not include approximately \$35 million of revenue earned on our new Circle/BMI investment as such rent was prepaid before the acquisition closed), primarily consisting of rent and interest from mortgage and other loans. We used these operating cash flows to fund our dividends of \$280 million. In addition, we invested \$2.3 billion in real estate and other assets, including the £1.5 billion Circle acquisition of 30 properties in January 2020 (as more fully described in Note 3 to Item 1 of this Form 10-Q), using a combination of cash on-hand, and proceeds from a £700 million British pound sterling term loan and the sale of 8.6 million shares of common stock under our at-the-market equity offering program, resulting in net proceeds of approximately \$170 million.

#### 2019 Cash Flow Activity

During the first half of 2019, we generated \$198.1 million of cash flows from operating activities, primarily consisting of rent and interest from mortgage and other loans. We used these operating cash flows to fund our dividends of \$192.6 million.

Certain investing and financing activities in the first half of 2019 included:

- a) Purchased \$1.4 billion in real estate assets representing 36 facilities across five countries;
- b) Funded approximately \$200 million of development, capital addition, and other projects;

- Sold 22.6 million shares of common stock under our at-the-market equity offering program, resulting in net proceeds of approximately \$399 million; and
- d) Closed on an Australian term loan facility for approximately \$837 million to help fund the Healthscope acquisition.

#### Short-term Liquidity Requirements:

As of July 31, 2020, we have no debt principal payments due between now and February 2021 when our revolving credit facility, with a current outstanding amount of \$100 million, comes due (which we can extend by one year). At July 31, 2020, our availability under our revolving credit facility plus cash on-hand approximated \$1.4 billion. We believe this liquidity along with our current monthly cash receipts from rent and loan interest, regular distributions from our joint venture arrangements, and availability under our at-the-market equity program is sufficient to fund our operations, debt and interest obligations, our firm commitments, and dividends in order to comply with REIT requirements for the next twelve months.

As noted previously, we received 98% of rent and interest payments through the first six months of 2020. Although we expect to receive a similar level of rent and interest in the second half of the year, no assurances can be made at this time that such payments will be made in a timely manner if the impact from COVID-19 on our tenants' cash flows significantly worsens. If such rent and interest payments are delayed, we believe such delay will be temporary and we can rely on our current liquidity of \$1.4 billion to cover our cash needs until our tenants are able to pay in full.

#### Long-term Liquidity Requirements:

As of July 31, 2020, we have no debt principal payments due between now and February 2021 when our revolving credit facility, with a current outstanding amount of \$100 million, comes due (which we can extend by one year). With our liquidity at July 31, 2020 of approximately \$1.4 billion, along with our current monthly cash receipts from rent and loan interest, regular distributions from our joint venture arrangements, and availability under our atthe-market equity program, we believe such liquidity is sufficient to fund our operations, debt and interest obligations, our firm commitments, and dividends in order to comply with REIT requirements for the foreseeable future.

However, in order to fund additional investments, to fund debt maturities coming due in later years, or to strategically refinance any existing debt (including our Credit Facility coming due in 2022) in order to reduce interest rates, we may need to access one or a combination of the following sources of capital:

- sale of equity securities;
- issuance of new USD, EUR, or GBP denominated debt securities, including senior unsecured notes;
- amending or entering into a new revolving credit facility and/or bank term loans,
- placing new secured loans on real estate located outside the U.S.; and/or
- proceeds from strategic property sales or joint ventures.

However, there is no assurance that conditions will be favorable for such possible transactions (particularly in light of the ongoing COVID-19 pandemic) or that our plans will be successful.

Principal payments due on our debt (which excludes the effects of any discounts, premiums, or debt issue costs recorded) as of July 31, 2020 are as follows (in thousands):

2020	\$ —
2021	100,000
2022	788,900
2023	523,400
2024	1,657,160
Thereafter	5,089,950
Total	\$ 8,159,410

#### **Disclosure of Contractual Obligations**

We presented our contractual obligations in our Annual Report on Form 10-K for the fiscal year ended December 31, 2019 and updated the schedule in the first quarter of 2020. Except for changes to our purchase obligations, as more fully described in Note 9 to Item 1 of this Form 10-Q, there have been no other significant changes as of June 30, 2020. However, see Note 10 for activities subsequent to June 30, 2020.

The following table updates our contractual obligations schedule for these updates (in thousands):

	Less Than			After						
Contractual Obligations	1 Year(1)		1-3 Years		3-5 Years		5 Years		Total	
Purchase obligations	\$	684,486	\$	104,450	\$	68,784	\$	189,155	\$	1,046,875

(1) This column represents the remaining six months of 2020.

#### **Distribution Policy**

The table below is a summary of our distributions declared during the two year period ended June 30, 2020:

Declaration Date	Record Date	Date of Distribution	_	Distribution per Share
May 21, 2020	June 18, 2020	July 16, 2020	\$	0.27
February 14, 2020	March 12, 2020	April 9, 2020	\$	0.27
November 21, 2019	December 12, 2019	January 9, 2020	\$	0.26
August 15, 2019	September 12, 2019	October 10, 2019	\$	0.26
May 23, 2019	June 13, 2019	July 11, 2019	\$	0.25
February 14, 2019	March 14, 2019	April 11, 2019	\$	0.25
November 15, 2018	December 13, 2018	January 10, 2019	\$	0.25
August 16, 2018	September 13, 2018	October 11, 2018	\$	0.25

We intend to pay to our stockholders, within the time periods prescribed by the Internal Revenue Code of 1986, as amended ("Code"), all or substantially all of our annual taxable income, including taxable gains from the sale of real estate and recognized gains on the sale of securities. It is our policy to make sufficient cash distributions to stockholders in order for us to maintain our status as a REIT under the Code and to avoid corporate income and excise taxes on undistributed income. However, our Credit Facility limits the amount of dividends we can pay - see Note 4 in Item 1 to this Form 10-Q for further information.

#### Item 3. Quantitative and Qualitative Disclosures About Market Risk.

Market risk includes risks that arise from changes in interest rates, foreign currency exchange rates, commodity prices, equity prices, and other market changes that affect market sensitive instruments. We seek to mitigate the effects of fluctuations in interest rates by matching the terms of new investments with new long-term fixed rate borrowings to the extent possible. We may or may not elect to use financial derivative instruments to hedge interest rate or foreign currency exposure. For interest rate hedging, these decisions are principally based on our policy to match investments with comparable borrowings, but are also based on the general trend in interest rates at the applicable dates and our perception of the future volatility of interest rates. For foreign currency hedging, these decisions are principally based on how our investments are financed, the long-term nature of our investments, the need to repatriate earnings back to the U.S., and the general trend in foreign currency exchange rates.

In addition, the value of our facilities will be subject to fluctuations based on changes in local and regional economic conditions and changes in the ability of our tenants to generate profits, all of which may affect our ability to refinance our debt, if necessary. The changes in the value of our facilities would be impacted also by changes in "cap" rates, which is measured by the current base rent divided by the current market value of a facility.

Our primary exposure to market risks relates to fluctuations in interest rates and foreign currency. The following analyses present the sensitivity of the market value, earnings, and cash flows of our significant financial instruments to hypothetical changes in interest rates and exchange rates as if these changes had occurred. The hypothetical changes chosen for these analyses reflect our view of changes that are reasonably possible over a one-year period. These forward looking disclosures are selective in nature and only address the potential impact from these hypothetical changes. They do not include other potential effects which could impact our business as a result of changes in market conditions (such as the impact caused by COVID-19 in the form of greater volatility in exchange rates). In addition, they do not include measures we may take to minimize our exposure such as entering into future interest rate swaps to hedge against interest rate increases on our variable rate debt.

#### **Interest Rate Sensitivity**

For fixed rate debt, interest rate changes affect the fair market value but do not impact net income to common stockholders or cash flows. Conversely, for floating rate debt, interest rate changes generally do not affect the fair market value but do impact net income to common stockholders and cash flows, assuming other factors are held constant. At June 30, 2020, our outstanding debt totaled \$7.8 billion, which consisted of fixed-rate debt of approximately \$7.6 billion and variable rate debt of \$0.2 billion. If market interest rates increase by 1%, the fair value of our debt at June 30, 2020 would decrease by \$9.2 million. Changes in the fair value of our fixed rate debt will not have any impact on us unless we decided to repurchase the debt in the open market.

If market rates of interest on our variable rate debt increase by 1%, the increase in annual interest expense on our variable rate debt would decrease future earnings and cash flows by less than \$0.1 million per year. If market rates of interest on our variable rate debt decrease by 1%, the decrease in interest expense on our variable rate debt would increase future earnings and cash flows by less than \$0.1 million per year. This assumes that the average amount outstanding under our variable rate debt for a year is \$0.2 billion, the balance of such variable rate debt at June 30, 2020.

#### Foreign Currency Sensitivity

With our investments in Germany, Spain, Italy, Portugal, the United Kingdom, Switzerland, and Australia, we are subject to fluctuations in the euro, British pound, Swiss franc, and Australian dollar to U.S. dollar currency exchange rates. Although we generally deem investments in these countries to be of a long-term nature, are able to match any non-U.S. dollar borrowings with investments in such currencies, and historically have not needed to repatriate a material amount of earnings back to the U.S., increases or decreases in the value of the respective non-U.S. dollar currencies to U.S. dollar exchange rates may impact our financial condition and/or our results of operations. Based solely on our 2020 operating results, a 5% change to the following exchange rates would have impacted our net income and FFO by the amounts below (in thousands):

	Net Income In	Net Income Impact			FFO Impact		
Euro (€)	\$	96	\$		1,745		
British pound (£)		958			3,885		
Swiss franc (CHF)		179			774		
Australian dollar (AUD \$)		596			1.536		

#### Item 4. Controls and Procedures.

#### Medical Properties Trust, Inc. and MPT Operating Partnership, L.P.

We have adopted and maintain disclosure controls and procedures that are designed to ensure that information required to be disclosed in our reports under the Securities Exchange Act of 1934, as amended, is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms and that such information is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate, to allow for timely decisions regarding required disclosure. In designing and evaluating the disclosure controls and procedures, management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives, and management is required to apply its judgment in evaluating the cost-benefit relationship of possible controls and procedures.

As required by Rule 13a-15(b), under the Securities Exchange Act of 1934, as amended, we have carried out an evaluation, under the supervision and with the participation of management, including our Chief Executive Officer and Chief Financial Officer, of the effectiveness of the design and operation of our disclosure controls and procedures as of the end of the quarter covered by this report. Based on the foregoing, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures are effective in providing reasonable assurance that information required to be disclosed by us in the reports that we file under the Securities Exchange Act of 1934 is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms.

There has been no change in our internal control over financial reporting during our most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

#### PART II — OTHER INFORMATION

#### Item 1. Legal Proceedings.

The information contained in Note 9 "Commitments and Contingencies" to condensed consolidated financial statements is incorporated by reference into this Item 1.

#### Item 1A. Risk Factors.

Please review the risk factors disclosed under the section entitled "Risk Factors" beginning on page 17 of our Annual Report on Form 10-K for the year ended December 31, 2019 and filed with the SEC on February 27, 2020, in our Current Report on Form 8-K filed with the SEC on April 8, 2020, as well as the additional risk factor disclosed on page 32 in our Quarterly Report on Form 10-Q filed with the SEC on May 11, 2020. There have been no other material changes to the Risk Factors as presented in our Annual Report on Form 10-K for the year ended December 31, 2019.

#### Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

- (a) None.
- (b) Not applicable.
- (c) None.

#### Item 3. Defaults Upon Senior Securities.

None.

#### Item 4. Mine Safety Disclosures.

None.

#### Item 5. Other Information.

- (a) None.
- (b) None.

# Item 6. Exhibits

Exhibit Number	Description
3.1(1)	Amendment to Second Amended and Restated Bylaws of Medical Properties Trust, Inc.
10.1*	Form of Lease Agreement between certain subsidiaries of MPT Operating Partnership, L.P., as Lessor, and Circle Health Ltd. and certain of its subsidiaries, as Lessee
31.1*	Certification of Chief Executive Officer pursuant to Rule 13a-14(a) under the Securities Exchange Act of 1934. (Medical Properties Trust, Inc.)
31.2*	Certification of Chief Financial Officer pursuant to Rule 13a-14(a) under the Securities Exchange Act of 1934. (Medical Properties Trust, Inc.)
31.3*	Certification of Chief Executive Officer pursuant to Rule 13a-14(a) under the Securities Exchange Act of 1934. (MPT Operating Partnership, L.P.)
31.4*	Certification of Chief Financial Officer pursuant to Rule 13a-14(a) under the Securities Exchange Act of 1934. (MPT Operating Partnership, L.P.)
32.1**	Certification of Chief Executive Officer and Chief Financial Officer pursuant to Rule 13a-14(b) under the Securities Exchange Act of 1934 and 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002. (Medical Properties Trust, Inc.)
32.2**	Certification of Chief Executive Officer and Chief Financial Officer pursuant to Rule 13a-14(b) under the Securities Exchange Act of 1934 and 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002. (MPT Operating Partnership, L.P.)
Exhibit 101.INS*	XBRL Instance Document - the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL document.
Exhibit 101.SCH*	Inline XBRL Taxonomy Extension Schema Document
Exhibit 101.CAL*	Inline XBRL Taxonomy Extension Calculation Linkbase Document
Exhibit 101.DEF*	Inline XBRL Taxonomy Extension Definition Linkbase Document
Exhibit 101.LAB*	Inline XBRL Taxonomy Extension Label Linkbase Document
Exhibit 101.PRE*	Inline XBRL Taxonomy Extension Presentation Linkbase Document
Exhibit 104*	Cover Page Interactive Data File (formatted as Inline XBRL with applicable taxonomy extension information contained in Exhibits 101.*)

Filed herewith.
Furnished herewith.
Incorporated by reference to Registrant's current report on Form 8-K, filed with the Commission on May 22, 2020. (1)

### **SIGNATURE**

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrants have duly caused this report to be signed on their behalf by the undersigned thereunto duly authorized.

### MEDICAL PROPERTIES TRUST, INC.

By: /s/ J. Kevin Hanna

J. Kevin Hanna Vice President, Controller, Assistant Treasurer, and Chief Accounting Officer (Principal Accounting Officer)

### MPT OPERATING PARTNERSHIP, L.P.

By: /s/ J. Kevin Hanna

J. Kevin Hanna Vice President, Controller, Assistant Treasurer, and Chief Accounting Officer of the sole member of the general partner of MPT Operating Partnership, L.P. (Principal Accounting Officer)

Date: August 7, 2020

# Lease

[Name of MPT entity]
and
BMI Healthcare Limited
and
Circle Health Holdings Limited
of land and buildings at [Property details]
NOTE: This document is a conformed copy produced for information purposes only. The original lease dated 31 July 2006 and the deeds of variation to the lease dated 14 December 2018 or 3 June 2019 (as applicable) and 16 June 2020 should be referred to for all legal purposes.

# TABLE OF CONTENTS

	Page
1.DEFINITIONS	5
2.INTERPRETATION	10
3.DEMISE	12
4.THE TENANT'S COVENANTS	12
5.THE LANDLORD'S COVENANTS	29
6.INSURANCE	29
7.PROVISOS	29
8.CONFIDENTIALITY	34
9.THE CONTRACTS (RIGHTS OF THIRD PARTIES) ACT 1999	35
10.LANDLORD'S OPTION TO PURCHASE THE TENANT'S ASSETS	35
11.FINANCIAL COVENANTS	35
Schedule 1 Particulars of Matters to which the Premises are Subject and have the benefit of	41
Schedule 2 Rent Review Provisions	42
Schedule 3 Authorised Guarantee Agreement	46
Schedule 4 Reporting Information	53
Schedule 5 Approved Works Covenants	58
Schedule 6 Landlord's call option	60
Schedule 7 Call Option Notice	66
Schedule 8 Security Power of Attorney	67
Schedule 9 Schedule of rent payments between 31 October 2018 and 30 October 2019	68

LR1. Date of lease	31 July 2006
LR2. Title number(s)	LR2.1 Landlord's title number(s)
	The tile number(s) out of which this lease is granted. Leave blank if not registered.
	GM457473 (FH)
	GM31452 (FH)
	GM808975 (FH)
	GM876339 (FH)
	LR2.2 Other title numbers
	Existing title number(s) against which entries of matter referred to in LR9, LR10. LR11 and LR13 are to be made.
LR3. Parties to this lease	Landlord
Give full names, addresses and company's registered number, if any of each of the parties. For Scottish companies use a SC prefix and follimited liability partnerships use an OC prefix. For foreign companies give territory in which incorporated.	7
	CIRCLE HEALTH HOLDINGS LIMITED (Company number 10543098) whose registered office is at 32 Welbeck Street, London W1G 8EU
LR4. Property	In the case of a conflict between this clause and the remainder of this lease then, for the purposes of registration, this clause shall
Insert a full description of the land being leased or Refer to the clause, schedule or paragraph of a schedule in this lease in which the land being leased is more fully described.	prevail.
Where there is a letting of part of a registered title, a plan must be attached to this lease and any floor levels must be specified.	

LR5. Prescribed statements etc.  If this lease includes a statement falling within LR5.1, insert under that sub-clause the relevant statement or refer to the clause, schedule or paragraph of a schedule in this lease which contains the statement.  In LR5.2, omit or delete those Acts which do not apply to this lease.	LR5.1 Statements prescribed under rules 179 (dispositions in favour of a charity), 180 (dispositions by a charity) or 196 (leases under the Leasehold Reform, Housing and Urban Development Act 1993) of the Land Registration Rules 2003.  None  LR5.2 This lease is made under, or by reference to, provisions of:
	From and including 31 July 2006  To and including 30 October 2031  NOTE: There is also a reversionary lease dated 14 December 2018 on materially the same terms as this lease and which has a term running from and including 31 October 2031 to and including 30 October 2050 (the "Reversionary Lease").  There are two Tenant options to renew the lease for an additional term of five years each. The first new lease will commence on the date immediately following the expiry of the Reversionary Lease and will be on the terms and in the form of the Reversionary Lease save in respect of rent. The second new lease will commence on the date immediately following the expiry of the first new lease and will be on the terms and in the form of the Reversionary Lease save in respect of rent.
LR7. Premium  Specify the total premium, inclusive of any VAT where payable.	None
LR8. Prohibitions or restrictions on disposing of this lease  Include whichever of the two statements is appropriate.  Do not set out here the wording of the provision.	This Lease contains a provision that prohibits or restricts dispositions.

	LR9.1 Tenant's contractual rights to renew this lease, to acquire	
Insert the relevant provisions in the sub-clauses or refer to the clause,	the reversion or another lease of the Property, or to acquire an	
schedule or paragraph of a schedule in this lease which contains the		
. • .	None	
	LR9.2 Tenant's covenant to (or offer to) surrender this lease	
	None	
	LR9.3 Landlord's contractual rights to acquire this lease	
	None	
LR10. Restrictive covenants given in this lease by the Landlord in respect of land other than the Property	None	
Insert the relevant provisions or refer to the clause, schedule or paragraph of a schedule in this lease which contains the provisions.		
	LR11.1 Easements granted by this lease for the benefit of the Property	
Refer here only to the clause, schedule or paragraph of a schedule in this lease which sets out the easements.	None	
	LR11.2 Easements granted or reserved by this lease over the Property for the benefit of other property	
	None	
LR12. Estate rentcharge burdening the Property	None	
Refer here only to the clause, schedule or paragraph of a schedule in this lease which sets out the rentcharge		

LR13. Application for standard form of restriction	None
Set out the full text of the standard form of restriction and the title against which it is to be entered. If you wish to apply for more than one standard form of restriction use this clause to apply for each of them, tell us who is applying against which title and set out the full text of the restriction you are applying for.  Standard forms of restriction are set out in schedule 4 to the Land Registration Rules 2003.	
LR14. Declaration of trust where there is more than one person comprising the Tenant	Not applicable
If the Tenant is one person, omit or delete all the alternative statements.	
If the Tenant is more than one person, complete this clause by omitting or deleting all inapplicable alternative statements.	

#### LAND REGISTRY

### **LAND REGISTRATION ACT 2002**

THIS LEASE is made on the date specified in clause LR1 and between the parties specified in clause LR3

#### NOW THIS DEED WITNESSES AS FOLLOWS:

### 1. **DEFINITIONS**

1.1 In this Lease the following words and expressions have the following meanings:

"Act of Insolvency" means in relation to a corporate body that:

- (a) a proposal is made for a voluntary arrangement, or a voluntary arrangement is implemented, under Part I of the Insolvency Act 1986 (referred to as the "**Act**" in the remainder of this definition) or
- (b) any formal step is taken under the Act by it or a holder of a qualifying floating charge (as defined in the Act) to place it into administration or
- (c) an administrator is appointed under Schedule B1 of the Act or
- (d) a receiver and (or) manager or administrative receiver is appointed whether under Part III of the Act or otherwise or
- (e) it goes into liquidation as defined in section 247(2) of the Act (other than a voluntary winding up solely for the purpose of amalgamation or reconstruction while solvent) or
- (f) a provisional liquidator is appointed under section 135 of the Act or
- (g) a scheme of arrangement is made under Part 26 of the Companies Act 2006

and the paragraphs above shall also apply in relation to a partnership or limited partnership (as defined in the Partnership Act 1890 and the Limited Partnerships Act 1907 respectively) subject to the modifications referred to in the Insolvent Partnerships Order 1994 (SI 1994/2421) (as amended) and to a limited liability partnership (as defined in the Limited Liability Partnerships Act 2000) subject to the modifications referred to in the Limited Liability Partnerships Regulations 2001 (SI 2001/1090)

and in relation to an individual that:

- (a) an application is made for an interim order or a proposal for a voluntary arrangement is made under Part VIII of the Act or the taking of any steps in connection with any other compromise or arrangement for the benefit of his creditors
- (b) a bankruptcy petition is presented to the court or his circumstances are such that a bankruptcy petition could be presented under Part IX of the Act or
- (c) he enters into a deed of arrangement

and for the avoidance of doubt "Act of Insolvency" includes any similar or equivalent step or proceeding which may be taken pursuant to the legislation of another jurisdiction in relation to a tenant incorporated or domiciled in such relevant jurisdiction

- 1.2 "Advisers" means, with respect to a person, its professional advisers, including its legal, property and financial advisers, valuers, insurance brokers, underwriters, accountants, bankers, auditors and consultants
- 1.3 "Affiliate" means, with respect to a person: (a) any other person directly or indirectly Controlling or Controlled by, or under direct or indirect common Control with, that person; and (b) any Related Funds of that person
- 1.4 "Authorised Guarantee Agreement" means an authorised guarantee agreement as described in section 16 of the Landlord and Tenant (Covenants) Act 1995
- 1.5 "Business Day" means a day on which clearing banks in the City of London are (or would be but for a strike lockout or other stoppage affecting particular banks or banks generally) open during banking hours and "Business Days" shall be interpreted accordingly
- 1.6 "Care Act" means the Health and Social Care Act 2008
- 1.7 "Circle Group" means Circle Health Holdings Limited (Company number 10543098) and its subsidiaries from time to time
- 1.8 "Conduits" means all conduits sewers drains mains ducts pipes gutters watercourses wires cables fibres channels flues and all other conducting media including any fixings louvres cowls and any other ancillary apparatus
- 1.9 **"Confidential Information"** means the information, received by the Landlord from the Tenant in accordance with clause 4.25 and Schedule 4, and including information given orally and any document, electronic file or any other way of representing or recording information which contains or is derived or copied from such information but excludes information that:
  - (a) is or becomes public information other than as a direct or indirect result of any breach of undertaking or duty of confidentiality by the party receiving such information;
  - (b) is identified in writing at the time of disclosure as non-confidential by the party disclosing such information; or
  - (c) is known before the date the information is disclosed to it or is lawfully obtained after that date and which, as far as the recipient is aware, has not been obtained in breach of, and is not otherwise subject to, any obligation of confidentiality
- 1.10 "Control" means, in relation to a person, the power to direct or cause the direction of the management and policies of that person, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise; and the terms "Controlling" and "Controlled" shall be construed accordingly;
- 1.11 "Deed of Variation" means a deed of variation to this Lease dated 14 December 2018 and made between (1) [Name of MPT entity] and (2) BMI Healthcare Limited
- 1.12 "Development" has the meaning given by section 55 of the Town and Country Planning Act 1990
- 1.13 "Environment" means all or any of the following media: namely air water and land and including such media within any buildings or other natural or man-made structures above on or below ground and any living organisms or systems

- 1.14 "Environmental Law" means European Community legislation (including any legally binding regulation or directive) the common law legislation (including without limitation subordinate legislation legally binding statutory guidance and any order or notice made pursuant to such legislation) any by-law or judgement or order of any court or administrative tribunal in each case relating to the protection of the Environment matters affecting human health and safety the health of other living organisms and/or disposal spillage release emission or migration of any Hazardous Substance and any legally binding code of practice procedure or standard compliance with which is required in order to discharge any duty or to conform with any legally binding practice recommended by any governmental or regulatory body in relation to any matter referred to in this definition
- 1.15 "EPC" means both an energy performance certificate and a recommendation report (as each term is defined in The Energy Performance of Buildings (England and Wales) Regulations 2012)
- 1.16 "Group Company" means any company which is for the time being a subsidiary or a holding company or another subsidiary of the holding company in each case within the meaning of section 1159 of the Companies Act 2006 which section shall for this purpose be deemed not to have been amended by subsequent legislation
- 1.17 "Hazardous Substance" means any substance including any waste which is capable of causing material harm to the Environment
- 1.18 "Interest" means interest both before and after any judgement at the Interest Rate then prevailing during the period beginning on the date on which the relevant payment is due and ending on the date on which the relevant payment is received by way of cleared funds
- 1.19 "Interest Rate" means three per cent. above the base lending rate from time to time in force of Barclays Bank or such other bank which is a member of CHAPS Limited as the Landlord may from time to time reasonably nominate in writing
- 1.20 "Landlord" means the party described as the Landlord in clause LR3 and includes the party for the time being entitled to the reversion immediately expectant on the determination of the Term
- 1.21 "Landlord and Tenant Act" means the Landlord and Tenant Act 1954
- 1.22 "Landlord Group Company" means a Group Company of the Landlord but excluding the Tenant and any Tenant Group Company
- 1.23 "Landlord's Energy Management Costs" means:
  - (a) the costs of the Landlord of acquiring allowances of any nature and paying all present and future taxes, duties, or assessments of any nature relating to the supply or consumption of energy, or relating to emissions consequential upon that supply or consumption (and whether those emissions are direct or indirect); and
  - (b) the reasonable costs of the Landlord of:
    - (i) monitoring the supply and consumption of energy and such emissions; and
    - (ii) gathering and processing information relating to the supply and consumption of energy and to such emissions

and in this definition "Landlord" means the group of undertakings of which the Landlord is a member for the purposes of such allowances or taxes

- 1.24 "Lease" means this lease and any document which is made supplemental to this lease
- 1.25 "MPT Group" means MPT Operating Partnership, L.P (a company incorporated in Delaware with company number 3701429) and its subsidiaries from time to time
- 1.26 "Net Internal Area" means the total area expressed in square feet or square metres measured in accordance with the Code of Measuring Practice published on behalf of The Royal Institution of Chartered Surveyors current at the date when it is required to be determined
- 1.27 "Operating Standards" means the national minimum standards from time to time applicable to the Premises, being at the date of the Deed of Variation:
  - (a) in England, the standards authorised pursuant to section 23 of the Health and Social Care Act 2008
  - (b) in Wales, the standards authorised pursuant to section 23 of the Care Standards Act 2000
  - (c) in Scotland, the standards authorised pursuant to section 10H of the National Health Service (Scotland) Act 1978
- 1.28 "**Payment Day**" means the 1st day of each month
- 1.29 "Permit" means any consent approval authorisation permission permit licence registration or notification required under any statute (including Environmental Law) in respect of the Premises and/or the Tenant's Assets or the use to which the Premises and/or the Tenant's Assets are being put
- 1.30 "Permitted Part" means such part or parts of the Premises as are capable of separate occupation (each with the benefit of the use of such common parts as may be appropriate) provided that such parts in total will not comprise more than twenty per cent (20%) of the Net Internal Area of the Premises as at the date of such occupation and save that (i) any areas underlet (for the location and use of scanning equipment and/or other special clinical or other diagnostic services) pursuant to Clause 4.11(j) and/or any areas of shared occupation in accordance with Clause 4.11(k) and/or any areas subject to licences to occupy in accordance with Clause 4.11(l) will be disregarded when calculating the said 20% limit and (ii) if at the date of the Supplemental Deed there exist any underleases or licences then the Tenant is entitled to renew or regrant all or any such underleases or licences provided that this does not result in the total Net Internal Area of the Premises so underlet exceeding more than forty per cent (40%) of the Net Internal Area of the Premises as at the date of such occupation.
- 1.31 "Permitted Underlease" means an underlease which:
  - (a) is granted without any fine or premium being paid either to the landlord or to the tenant
  - (b) reserves a rent not less than the then open market rent of the Permitted Part subject to such rent free periods and other incentives as may then be usual in the open market
  - (c) incorporates provisions for the uplift of open market rent at intervals not greater than five (5) years
  - (d) is (so far as is consistent with an underlease) in a form substantially the same as this Lease except that: (i) further subletting shall be prohibited (save that one further subletting shall be permitted in respect of an underlease of non-clinical space); (ii) payment of the Principal Rent shall be excluded; (iii) clause 4.9 shall be excluded; (iv) clause 4.25 shall be excluded; (v) Clause 11 shall be excluded; and (vi) Schedules 2, 4, 5, 7, 8, 9 and 10 shall be excluded, and

- (e) is validly excluded from the operation of sections 24-28 of the Landlord and Tenant Act
- 1.32 "Permitted User" means use as an acute care hospital with or without rehabilitation facility or other medical facility within Use Class C2, D1 and B1(b) of the 1987 Order and such ancillary uses as reasonably required by the Tenant
- 1.33 "Plan" means the plan attached to this Lease (if any)
- 1.34 "Planning Acts" means the Town and Country Planning Act 1990 the Planning (Listed Buildings and Conservation Areas) Act 1990 the Planning (Hazardous Substances) Act 1990 the Planning (Consequential Provisions) Act 1990 and the Planning and Compensation Act 1991 the Planning and Compulsory Purchase Act 2004 and any other legislation relating to town and country planning in force from time to time
- 1.35 "Premises" means the land and buildings known as [Property details] and includes:
  - (a) all soil subsoil substrata surface and groundwater beneath the Premises
  - (b) the foundations piles and all other load-bearing structures supporting the Premises
  - (c) all additions and improvements to the Premises
  - (d) all the Landlord's fixtures and fittings of every kind which shall from time to time be in or upon the Premises (whether originally affixed or fastened to or upon the Premises or otherwise) except any such fixtures installed by the Tenant that can be removed from the Premises without defacing the Premises
  - (e) all Conduits in on under or over the Premises and all electrical and mechanical installations plant equipment and machinery including (without limitation) lifts heating plant air conditioning plant and ventilation plant and boundary walls or fences

and references to the "Premises" in the absence of any provision to the contrary include any and every part of the Premises

- 1.36 "Principal Rent" has the meaning given to it in Schedule 2 (Rent Review)
- 1.37 "Related Fund" means any lease of any of the properties (except the Property) listed in Schedule 3 to the Supplemental Deed subsisting at the relevant time which is vested legally and beneficially in or the Tenant's interest is held by a member of the Circle Group and the immediate reversion to which is vested legally and beneficially in or the Landlord's interest is held by a member of the MPT Group.
- 1.38 "Related Lease" means any lease of any of the properties (except the Property) listed in Schedule 3 to the Supplemental Deed subsisting at the relevant time which is vested legally and beneficially in or the Tenant's interest is held by a member of the Circle Group and the immediate reversion to which is vested legally and beneficially in or the Landlord's interest is held by a member of the MPT Group
- 1.39 "Rent Commencement Date" means the date hereof
- 1.40 "Rents" means the Principal Rent and the other payments reserved as rent and referred to in clause 3

- 1.41 "Representative" means, with respect to a person, its employees, officers, directors, representatives, members, partners, prospective partners, investors, investment managers, investment advisers, management companies, capital sources and its and their Advisers
- 1.42 "Reversionary Lease" means the lease in respect of the Premises dated on or about the date of the Deed of Variation and made between the same parties hereto;
- 1.43 "Security Power of Attorney" means a power of attorney in the form set out in Schedule 8;
- 1.44 "Senior Charge" means the first ranking charge dated on or around the date of the Deed of Variation granted by the Tenant in favour of the Landlord in respect of its rights, title and interest from time to time in and to each of the Call Option Assets (as defined in Schedule 6)
- 1.45 "Services" means water sewerage gas electricity steam hydraulics telephone telecommunications and other services of whatever
- 1.46 "Supplemental Deed" means a supplemental deed to the Lease and the Reversionary Lease dated 16 June 2020 and made between (i) [Name of MPT entity], (ii) BMI Healthcare Limited and (iii) Circle Health Holdings Limited
- 1.47 "Surety" means any party who enters into covenants with the Landlord pursuant to clause 4.11 and in the case of an individual his personal representatives
- 1.48 "Tax Authority" means any taxing or other authority competent to impose, administer or collect any tax
- 1.49 "Tenant" means the party described as the Tenant in clause LR3 and includes the Tenant's successors in title and assigns
- 1.50 "**Tenant's Assets**" means all the Tenant's right title and interest in all equipment contents and other assets and choses in action situate in or at the Premises or used in connection with or otherwise referable to the Permitted User of the Premises (including any replacements or substitutions for or additions thereto during the Term)
- 1.51 "Tenant Group Company" means a Group Company of the Tenant but excluding the Landlord and any Landlord Group Company
- 1.52 "Tenant Group Loan Agreement" means the senior facility agreement dated 23 December 2019 between, inter alia, Circle Health 4 Limited and Ares Management Limited as Arranger, Agent and Security Agent as amended, supplemented, varied, novated, restated, replaced, renewed and/or refinanced from time to time.
- 1.53 "**Term**" means the term specified in clause LR6
- 1.54 "Value Added Tax" and "VAT" each mean value added tax or any other tax of a similar nature that may be substituted for or levied in addition to it in each case at the rate current from time to time
- 1.55 "1987 Order" means the Town and Country Planning (Use Classes) Order 1987.

### 2. **INTERPRETATION**

2.1 The schedules form part of this Lease

- 2.2 Where the Landlord or the Tenant for the time being are two or more persons obligations expressed or implied to be made by or with such party are deemed to be made by or with such persons jointly and severally
- 2.3 Words importing one gender include all other genders and words importing the singular include the plural and vice versa
- 2.4 Rights expressed to be reserved in favour of the Landlord shall be deemed to extend to any superior landlord and any mortgagee of the Landlord or the superior landlord (as applicable) and all persons authorised by the Landlord and by any superior landlord or mortgagee of the Landlord or the superior landlord (as applicable) including its or their agents professional advisers contractors and workmen
- 2.5 Save where inconsistent with the express provisions of this Lease any covenants by the Tenant not to do an act or thing shall be deemed to include an obligation not to knowingly permit or suffer such act or thing to be done
- Any provisions in this Lease referring to the consent or approval of the Landlord shall be construed as also requiring the consent or approval of any mortgagee of the Landlord or the superior landlord (as applicable) where such consent is required but nothing in this Lease shall be construed as implying that any obligation is imposed upon any such mortgagee or any superior landlord not unreasonably to refuse or delay any such consent or approval
- 2.7 Unless otherwise expressly provided any references to a specific statute include any statutory extension or modification amendment or re-enactment of such statute and any regulations instruments or orders made under such statute and any general reference to "statute" or "statutes" includes EC directives decisions and regulations and any regulations instruments orders or other directions made under such statute or statutes
- 2.8 References in this Lease to any clause schedule or paragraph without further designation shall be construed as a reference to a clause or paragraph of or schedule to this Lease so numbered
- 2.9 The clause paragraph and schedule headings and the table of contents are for ease of reference only and shall not be taken into account in the construction or interpretation of this Lease or of the clause paragraph or schedule to which they refer
- 2.10 References to "last year of the Term" include the last year of the Term if the Term shall determine otherwise than by effluxion of time and references to "expiry of the Term" include such other determination of the Term
- 2.11 The terms "parties" or "party" mean the Landlord and (or) the Tenant
- 2.12 References to "emergency" include where there is an immediate risk of harm or damage to persons property or the environment
- 2.13 References to "act or default of the Tenant" or words to similar effect shall be construed as including any omission of the Tenant and any act default or omission of any undertenant or anyone else at the Premises with the Tenant's actual or implied authority
- 2.14 References to "consent of the Landlord" or words to similar effect mean a consent in writing signed by or with the express authority of the Landlord and references to "approved" and "authorised" or words to similar effect mean approved as authorised in writing by or with the express authority of the Landlord

- 2.15 The words "including" and "in particular" are to be construed as being by way of illustration or emphasis only and are not to be construed so as to limit the generality of the words preceding them
- 2.16 Any payment which is due to be made under this Lease on a day that is not a Business Day shall be made on the next Business Day in the same calendar month (if there is one) or the preceding Business Day (if there is not)
- 2.17 References to any agreement, deed or instrument shall be construed so as to include that agreement, deed or instrument as amended, varied, novated, supplemented, extended and/or restated from time to time
- 2.18 Any word importing an individual includes a company and vice versa
- 2.19 References to a "person" includes any person, firm, company, corporation, government, state or agency of a state or any association, trust or partnership (whether or not having separate legal personality) or two or more of the foregoing
- 2.20 Any reference to a person shall be construed so as to include its successors in title, permitted assigns and permitted transferees to, or of, its rights and/or obligations under this Lease

### DEMISE

The Landlord demises to the Tenant the Premises to hold the Premises to the Tenant for the Term subject to and with the benefit of the matters contained or referred to in Schedule 1 yielding and paying to the Landlord:

- 3.1 on and from the date of the Supplemental Deed the Principal Rent payable without any deduction (save where and to the extent required by law), recoupment or set off by equal monthly payments in advance on the Payment Days in every year and proportionately for any period of less than a year the first such payment being a sum in respect of the period from and including 8 January 2020 up to and including 30 June 2020 to be paid on the date of the Supplemental Deed
- 3.2 in respect of the period between 31 October 2018 and 30 October 2019, the Principal Rent payable without any deduction (save where and to the extent required by law), recoupment or set off by making the payments set out in Schedule 9 to this Lease on the dates and in the amounts set out in Schedule 9 to this Lease (and for the avoidance of doubt from 31 October 2019 onwards the provisions of paragraph 3.1 shall apply)
- 3.3 by way of further rent:
  - (a) the monies properly expended by the Landlord in insuring the Premises as detailed in paragraph 2.1 of Annex 2 of the Supplemental Deed
  - (b) the payments of Interest referred to in clause 4.18
  - (c) such Value Added Tax as may be chargeable on the Rents in accordance with clause 7.9
  - (d) on demand any other amounts payable by the Tenant to the Landlord under this Lease

### 4. THE TENANT'S COVENANTS

The Tenant covenants with the Landlord:

#### 4.1 Rents

- (a) To pay the Rents on the days and in the manner set out or referred to in this Lease and not (save where and to the extent required by law) to exercise or seek to exercise any right or claim to make a deduction from rent or to withhold rent or any right or claim to legal or equitable set-off or recoupment
- (b) If so required in writing by the Landlord to make such payments by credit transfer
- (c) That other than as expressly provided by this Lease the Tenant's obligations to pay the Rents and other sums due under this Lease shall be absolute, unconditional and irrevocable and the rights of the Landlord to such Rents and other sums shall be absolute, unconditional and irrevocable and the Tenant shall not (save where and to the extent required by law) be entitled to exercise any right of abatement or reduction, set-off, counterclaim, recoupment, defence or deduction with respect to any Rents or any other sum payable to the Landlord under this Lease

### 4.2 Outgoings

- (a) To pay and to indemnify the Landlord against all rates taxes assessments duties charges impositions and outgoings which are now or during the Term shall be charged assessed or imposed upon the Premises or upon the owner or occupier of them excluding any payable by the Landlord occasioned by receipt of the Principal Rent, or any other sums payable to the Landlord hereunder, or any disposition of or dealing with or ownership of any interest reversionary to the interest created by this Lease and
- (b) To pay and to indemnify the Landlord against:
  - (i) all charges including meter rents, connection and hire charges for all Services consumed or used at or in relation to the Premises;
  - (ii) that part of the Landlord's Energy Management Costs which the Landlord reasonably attributes to the Premises, and to comply with all requirements from time to time of the suppliers of Services to the Premises; and
  - (iii) any costs incurred by the Landlord in complying with the Heat Network (Metering and Billing) Regulations 2014

### 4.3 Repair cleaning and decorating

- (a) Subject to clause 4.3(i) below, from the commencement of the Term to keep the Premises in good and substantial repair and condition (taking into account the age of the buildings on the Premises) and where necessary to rebuild reinstate or replace the Premises unless and to the extent that the Premises have:
  - (i) been damaged or destroyed by any of the Insured Risks; and
  - (ii) the insurance of the Premises is not vitiated, forfeited or avoided nor is the insurance money irrecoverable in consequence of any act or default of the Tenant

- (b) With effect from the fifth anniversary of the date of the Supplemental Deed, not less frequently than once in every five years during the Term and in the last year of the Term to redecorate the exterior of the Premises in a good and workmanlike manner and with appropriate materials of good quality provided that any change in the colours or patterns of such decoration shall in the last year of the Term require the prior written approval of the Landlord which approval shall not be unreasonably withheld or delayed
- (c) With effect from the fifth anniversary of the date of the Supplemental Deed, not less frequently than once in every five years during the Term and in the last year of the Term to redecorate the non-clinical areas of the interior of the Premises in a good and workmanlike manner and with appropriate materials of good quality provided that any change in the colours or patterns of such decoration shall in the last year of the Term require the prior written approval of the Landlord which approval shall not be unreasonably withheld or delayed
- (d) With effect from the fifth anniversary of the date of the Supplemental Deed, in every year that the major equipment within any clinical area is replaced (but not less than every 10 years of the Term) to redecorate the clinical areas of the interior of the Premises in a good and workmanlike manner and with appropriate materials of good quality provided that any change in the colours or patterns of such decoration shall in the last year of the Term require the prior written approval of the Landlord which approval shall not be unreasonably withheld or delayed
- (e) To keep any part of the Premises which may not be built upon or which is landscaped properly surfaced and/or cultivated (as appropriate) and in good condition and generally to keep the same in such condition as is not untidy nor unclean
- (f) Where the use of Conduits boundary structures or other things is common to the Premises and other property to be responsible for and to indemnify the Landlord against all proper sums due from and to undertake all work that is the responsibility of the owner lessee or occupier of the Premises in relation to those Conduits boundary structures or other things
- (g) To keep all plant machinery apparatus and equipment (including the Tenant's Assets that form part of the Premises) upon the Premises properly maintained and in good working order and repair and for that purpose to employ reputable contractors for the regular periodic inspection and maintenance of them and to renew all working and other parts as and when necessary or when recommended by such contractors and to ensure by directions to the Tenant's staff and otherwise that such plant machinery apparatus and equipment are properly operated
- (h) The Tenant's redecorating obligations in clauses 4.2(b), 4.3(b), 4.3(c) and 4.3(d) shall not apply in respect of the last year of the Term where the Reversionary Lease is subsisting at the relevant time
- (i) The Tenant's obligations under sub clause (a) above shall not apply to any Tenant's Assets that form part of the Premises, except to the extent that the Tenant is obliged to repair, maintain, reinstate or replace such Tenant's Assets pursuant to sub clause (g) above

### 4.4 User and Restrictions on Use

- (a) Not to use the Premises other than for a Permitted User
- (b) Not to do or allow to remain upon the Premises anything which may be or become or cause an actionable nuisance or cause damage to the Landlord or its tenants or the owners or occupiers of adjacent or neighbouring premises

- (c) Not to use the Premises for a sale by auction or for any dangerous noxious noisy or offensive trade business manufacture or occupation nor for any illegal or immoral act or purpose
- (d) Not to discharge into the Conduits serving the Premises any oil or grease or any deleterious objectionable or hazardous matter or substance and to take such measures as shall be necessary to ensure that any effluent so discharged into the Conduits will not be corrosive or otherwise harmful to the Conduits or cause obstruction or deposit in them in each case save to the extent authorised by a Permit or in the normal course of the Tenant's business relating to the Permitted User and in any event to keep the Conduits forming part of the Premises in good and proper working order and to comply with all requirements and regulations of the statutory authorities and the supply companies
- (e) Not to deposit on or adjacent to any part of the Premises any trade empties or refuse of any kind other than in proper receptacles
- (f) Not to install or use in or upon the Premises any safes goods machinery or apparatus which may strain or damage the Premises or cause structural damage to nearby premises and not to store or bring on to the Premises any article substance or liquid of a specially combustible inflammable or explosive nature save to the extent authorised by a Permit or in the normal course of the Tenant's business relating to Permitted User
- (g) Not to do anything which adversely interferes with the heating cooling or ventilation of or which imposes an additional load on any heating cooling or ventilation plant and equipment in the Premises beyond that which it is designed to bear
- (h) Not to operate the heating cooling and ventilation plant and equipment in the Premises otherwise than in accordance with the manufacturers' regulations for such plant and equipment
- (i) Not to close the whole or substantially the whole (meaning 50% or more of the Net Internal Area of the Premises) nor permit the whole or substantially the whole (as defined above) to remain vacant except:
  - insofar as it is necessary for the Tenant to do so by reason of destruction of or damage to the Premises by any of the Insured Risks
  - (ii) as may be necessary for the carrying out with all reasonable speed of any major repairs alterations renovations or additions to the Premises as permitted under the terms of this Lease
  - (iii) in so far as it is unlawful to keep it open or is otherwise necessary for the Tenant to do so by reason of any regulation or requirement or at the direction of any competent statutory or local authority

and in each case for a period not exceeding six (6) consecutive months, or such longer period with the consent of the Landlord subject to the Landlord being satisfied (acting reasonably) of the Tenant's plans to re-open or re-occupy the Premises within that period having regard to the reason for the closure and providing or paying for such caretaking or security arrangements as the Landlord shall reasonably require in order to protect the Premises from vandalism theft or unlawful occupation

### 4.5 Waste and Alterations

- (a) Not to:
  - (i) commit any waste on or at the Premises
  - (ii) unite the Premises with any adjoining premises
  - (iii) make any alterations which might result in the EPC rating for the Premises being lower than the EPC rating shown on the Landlord's current EPC immediately before such alterations would be carried out
- (b) Not to make any additions or structural alterations or any variations to the Premises which would require planning permission from the local authority under the Planning Acts without:
  - (i) obtaining and complying with all the necessary permissions licences and consents of any competent authority and the insurers and paying all charges of such authority persons or bodies in respect of such permissions licences and consents
  - (ii) making an application for consent to the Landlord supported by drawings (and where appropriate a specification) prepared by an architect or member of some other appropriate profession (who shall supervise the work throughout to completion) and obtaining the written consent of the Landlord pursuant to such application such consent not to be unreasonably withheld or delayed having regard to the nature of the application and the related works
  - (iii) paying the proper (and in the case of the Landlord only reasonable) fees of the Landlord or any superior landlord any mortgagee and their respective professional advisers in connection with such application
  - (iv) entering into such covenants as the Landlord may reasonably require as to the execution and insofar as in the reasonable opinion of the Landlord such additions alterations or variations would have an adverse impact on the reversionary value of the Premises the reinstatement of such additions alterations or variations and
  - (v) indemnifying the Landlord in respect of any increased insurance costs arising as a result of such alterations being carried out
- (c) Not to make any other variations alterations or additions to the Premises without first obtaining the Landlord's consent (not to be unreasonably withheld or delayed and the Landlord shall within a reasonable period of time indicate whether or not it shall give its consent) unless:
  - (i) such variations additions or alterations relate to the interior of the Premises only; and
  - (ii) such variations additions or alterations have no adverse impact on the reversionary value of the Premises;
  - (iii) such variations additions or alterations will comply with all building regulations and all other requirements of any public or statutory body; and
  - (iv) when completed such variations additions or alterations will not adversely affect the structural integrity of the Premises; and

- (v) the Tenant complying with the Approved Works Covenants in Schedule 5 in relation to the execution of such variations additions or alterations; and
- (vi) the Tenant indemnifies the Landlord in respect of any increased insurance costs arising as a result of such variations additions or alterations being carried out; and
- (vii) such variations additions or alterations are architecturally integrated and consistent with the Premises as then existing
- (d) As soon as reasonably practicable following completion of any variations alterations or additions for which the Landlord's consent is not required under clause 4.5(c) to give written notice to the Landlord of any such variations additions or alterations together with details of the same and a statement of the reasons for the proposed works together with copies of all plans and drawings relating thereto
- (e) In respect of any variations alterations and additions carried out by the Tenant during the Term to supply to the Landlord on reasonable written demand floor plans services drawings and such other information as the Landlord may from time to time reasonably require in order to satisfy itself as to the layout content and other details of the Premises

### 4.6 Aerials Signs and Advertisements

- (a) Not to install any additional pole mast wire aerial dish or screen nor additional equipment on any existing pole mast wire aerial (whether in connection with telephonic radio or television communication or otherwise) upon the Premises without consent to such installation having previously been obtained from the relevant competent authority and (if applicable) subject to clause 4.5
- (b) Not to fix to or exhibit or display on the outside of or to or through any window of the Premises or anywhere on the Premises any:
  - (i) advertising hoarding without first obtaining the Landlord's consent (which shall not be unreasonably withheld or delayed) and any requisite planning permission
  - (ii) any signs placards notices posters or advertisements except those which are reasonably required in connection with the business carried on at the Premises
- (c) The Landlord hereby consents to any such items as are referred to in clauses 4.6(a) and 4.6(b) which are in existence at the Premises as at the date of this Lease

### 4.7 Obstruction and Loading and Unloading

Not to do anything whereby any road path forecourt or other area over which the Tenant may have rights of access or use may be damaged or the proper and lawful use thereof by others may be improperly obstructed in any way

### 4.8 Statutory Obligations and Fire Precautions

(a) At the Tenant's own expense to execute all works and provide and maintain all arrangements upon or in respect of the Premises and/or the Tenant's Assets or the use to which the Premises and/or the Tenant's Assets are being put that are required in order to comply with the requirements of any statute (including for the avoidance of doubt Environmental Law) or any government department local authority other public or competent authority environmental authority or court of competent jurisdiction regardless of whether such requirements are imposed on the Landlord the Tenant or the occupier

- (b) At the Tenant's own expense and without limiting the obligations set out earlier in this clause 4.8:
  - (i) to comply in all respects with the provisions of any statutes and any other obligations imposed by law or by any by-laws applicable to the Premises and/or the Tenant's Assets or Operating Standards or in regard to carrying on the business for the time being carried on at the Premises and
  - (ii) to comply with the requirements and recommendations of the fire authority in relation to fire precautions affecting the Premises

#### 4.9 Permits

- (a) To obtain and maintain all the Permits necessary to carry out the Permitted User as then carried on at the Premises and on the termination of the Lease (except if the Reversionary Lease is subsisting at the relevant time) to take all reasonable steps to ensure that all such Permits are (at the reasonable direction of the Landlord) duly transferred to the extent that the Tenant is lawfully entitled to do so to the Landlord or to any other person nominated by the Landlord or surrendered insofar as those Permits can only be held in relation to the Premises and cannot be held by the Tenant or any employee of the Tenant thereafter in relation to the Tenant's business or the business of any Group Company of the Tenant
- (b) Subject to clause 4.9(c), to notify the Landlord as soon as reasonably practicable of any regulatory notice or legal proceedings (or threat of such notice or proceedings) against the Tenant relating to any matter affecting the Premises and to provide the Landlord with copies of any correspondence notices proceedings or other documents relating thereto where requested in writing by the Landlord
- (c) Clause 4.9(b) shall not extend to any regulatory notice or legal proceedings (or threat of such notice or proceedings) relating to the Tenant's business at the Premises save where the Tenant is required to give notice of such matters pursuant to clause 4.25

### 4.10 Access to Landlord and Notice of Repair

- (a) To permit the Landlord and its employees (or agents in respect of (i) and (ii) only) on no less than 7 Business Days prior written notice (except in case of emergency in which case no notice shall be necessary) and during reasonable times (or at any time in case of emergency):
  - (i) to enter upon the Premises for the purpose of ascertaining that the covenants and conditions of this Lease have been observed and performed
  - (ii) to inspect the state of repair and condition of the Premises (including the carrying out of insurance inspections and valuations and in connection with any rent review)
  - (iii) to give to the Tenant or leave upon the Premises a notice specifying any breach by the Tenant of the terms of this Lease and requesting the Tenant as soon as practicable to remedy the same

and in doing so the Tenant shall be entitled to accompany the Landlord or others permitted to enter in a manner which does not interfere with the exercise of the covenant to permit entry otherwise provided in this clause 4.10(a)

(b) As soon as reasonably practicable to remedy any breach specified by a notice given to the Tenant or left upon the Premises pursuant to clause 4.10(a)(iii)

- (c) If:
  - (i) in respect of any Landlord's notice pursuant to clause 4.10(a)(iii) which is served prior to the fifth anniversary of the date of the Supplemental Deed, within four months (or such longer period as may be: (a) reasonably required to obtain all regulatory consents, planning permissions and similar approvals necessary for the carrying out of such works; or (ii) reasonable having regard to the nature of the works required as shall be agreed between the Landlord and Tenant) of the service of such a notice the Tenant shall not have commenced and be proceeding diligently with the execution of the work referred to in the notice or shall have failed to complete the work within twelve months (or such longer period as may be reasonable having regard to the nature of the works required as shall be agreed between the Landlord and the Tenant); or
  - (ii) in respect of any Landlord's notice pursuant to clause 4.10(a)(iii) which is served on or after the fifth anniversary of the date of the Supplemental Deed, within two months (or such shorter period as may be reasonable in the case of emergency or such longer period as may be reasonable having regard to the nature of the works required as shall be agreed between the Landlord and Tenant) of the service of such a notice the Tenant shall not have commenced and be proceeding diligently with the execution of the work referred to in the notice or shall have failed to complete the work within four months (or such longer period as may be reasonable having regard to the nature of the works required as shall be agreed between the Landlord and the Tenant),

in either case, to permit the Landlord to enter the Premises to execute such work as may be necessary to comply with the notice subject to the Landlord complying with the Tenant's reasonable security and clinical requirements having regard to the Permitted User and to the Landlord causing as little damage as reasonably practicable to the Premises and the Tenant's Assets and to pay to the Landlord the proper and reasonable costs of so doing and all proper and reasonable expenses incurred by the Landlord (including legal costs and surveyor's fees) (together with Interest) within ten Business Days of a written demand such sums being reasonable either as a debt or rent in arrears at the option of the Landlord

### 4.11 Dealings

- (a) Not save as hereafter permitted to:
  - (i) hold the Premises expressly or impliedly on trust for another person
  - (ii) part with possession of the Premises
  - (iii) share possession or occupation of the Premises with another person
- (b) Not to assign or charge a part (as distinct from the whole) of the Premises
- (c) Not to underlet the whole of the Premises
- (d) Not to assign the whole of the Premises without the prior written consent of the Landlord (such consent not to be unreasonably withheld) provided that:
  - (i) for the purposes of section 19(1A) of the Landlord and Tenant Act 1927:

- (A) the Landlord shall be entitled to withhold consent to a proposed assignment where the proposed assignee is not a Qualifying Assignee and/or where any of the Prohibited Circumstances apply and
- (B) the Landlord shall be entitled to grant its consent to a proposed assignment subject to all or any of the Conditions for Assignment and
- (C) provided further that the Landlord shall be entitled to withhold consent to a proposed assignment on other grounds or grant consent subject to further conditions where it is reasonable to do so, and

For the purposes of this clause 4.11(d) the following terms shall have the following meanings:

### (ii) "Qualifying Assignee" means an assignee:

- (A) whose financial strength (taking into account any guarantees and other security for the performance by the assignee of the tenant covenants under this Lease (other than any Authorised Guarantee Agreement) is in the reasonable opinion of the Landlord at least equivalent to the financial strength of the Tenant and any Surety (taking into account the value of any other security for the performance of the tenant covenants of the Lease by the Tenant) at:
  - (aa) either the date of the Deed of Variation or (if such is the case) the date this Lease was assigned to the Tenant or
  - (bb) the date of the application for licence to assign

whichever is the greater

- (B) the assignment of the Lease to whom will not in the reasonable opinion of the Landlord detrimentally affect the value of the Landlord's reversion immediately expectant upon the determination of the Term
- (C) which in the reasonable opinion of the Landlord has demonstrated that it or the group of which the assignee is a member is an entity which is or is recognised within the hospital industry as a leading company or group in the management or operation of hospitals and/or other operations contemplated by the Permitted Use

### (iii) "Prohibited Circumstances" means the following circumstances:

- (A) where in the Landlord's reasonable opinion there is an outstanding material breach of any tenant covenant in this Lease
- (B) where the proposed assignee enjoys diplomatic or state immunity
- (C) where the proposed assignee is not resident in a jurisdiction where reciprocal enforcement of judgments exists

### (iv) "Conditions for Assignment" means the following conditions:

(A) the payment of all Rents and other sums which have fallen due under this Lease prior to the date of the assignment, but in the case of sums other

than the Principal Rent only to the extent that the Landlord has notified the Tenant in writing not less than 5 Business Days prior to the date of the assignment of the fact that such sums are outstanding

- (B) the payment of all Principal Rent which has fallen due under any Related Lease prior to the date of the assignment provided that following any Permitted Assignment to an entity which is not a member of the Circle Group this sub clause (B) will no longer apply
- (C) that where the consent of any superior landlord or mortgagee of the Landlord or the mortgagee of any superior landlord is required to the proposed assignment such consent is obtained and any lawfully imposed conditions of such consent are satisfied
- (D) that the proposed assignee covenants by deed with the Landlord in such form as the Landlord may reasonably require that with effect from the date of the assignment the proposed assignee will pay the Rents and observe and perform all the provisions of this Lease to be observed and performed by the Tenant
- (E) that the Tenant enters into an Authorised Guarantee Agreement in the form set out in the Schedule 3
- (F) that any person who has guaranteed the obligations of the Tenant under this Lease (otherwise than by way of an Authorised Guarantee Agreement) and whose guarantee is subsisting immediately before completion of the instrument of the assignment enter into a sub-guarantee of the obligations of the Tenant contained in an Authorised Guarantee Agreement entered into pursuant to clause 4.11(d)(iv)(E), such sub-guarantee to be in the form provided for by Schedule 3
- (G) that any guarantor for the proposed assignee reasonably required by the Landlord covenants as a primary obligation by deed with the Landlord to observe and perform all the covenants on the part of the Tenant contained in this Lease in the form (mutatis mutandis) of Schedule 3 or such other form as the Landlord may reasonably require
- (H) that the assignee has first granted a Security Power of Attorney in favour of the Landlord and any other persons holding a superior interest in the Premises or this Lease including any superior landlord and the mortgagee of the Landlord and any superior landlord and has delivered to the Landlord (or such other person as aforesaid) such Security Power of Attorney
- (I) that the assignee has first granted a first ranking charge, effective from the date of assignment of this Lease, over all of the proposed assignee's rights, title and interest of in all chattels, equipment, fittings and moveable contents situate in or at the Premises or used exclusively in connection with or otherwise referable to the permitted user of the Premises (including any Call Option Assets that may be transferred to the proposed assignee pursuant to paragraph 4.11 of Schedule 6) in favour of the Landlord in substantially the same form as the Senior Charge with such amendments as the parties thereto may agree (each acting reasonably)

- (J) that to the extent that the assignee is to grant on completion of the assignment security over the Call Option Assets to its secured lenders that assignee has first entered into an intercreditor agreement with its secured lenders (or their agent) and the Landlord in form and substance reasonably acceptable to the Landlord
- (K) that the Reversionary Lease is also assigned to the proposed assignee concurrently with the assignment of this Lease
- (L) that save where the assignment is required due to either:
  - (1) a Competition and Markets Authority direction: (i) pursuant to the approval process being carried out as at the date of the Supplemental Deed and/or (ii) issued during the period from circa 1 January 2020 to the date of the Supplemental Deed; or
  - (2) applicable law or statutory/regulatory requirement (but for the avoidance of doubt not any future Competition and Markets Authority approval process),

### ((1) and (2) each being a "Permitted Assignment")

all Related Leases are also assigned to the proposed assignee concurrently with the assignment of this Lease provided that following any Permitted Assignment to an entity which is not a member of the Circle Group this sub-clause (L) will no longer apply

- (e) Not to charge the whole of the Premises except to a bank or similar financial institution and with the prior written consent of the Landlord (which will not be unreasonably withheld or delayed). The Tenant shall not be permitted to charge the whole of the Premises unless the Reversionary Lease is also charged to the same bank or financial institution concurrently with the charging of the Premises.
- (f) Not to underlet a Permitted Part of the Premises except by way of a Permitted Underlease with the prior written consent of the Landlord (which will not be unreasonably, withheld) and further:
  - (i) unless the proposed undertenant has first covenanted by deed with the Landlord in such form as the Landlord may reasonably require that with effect from the date of the underlease and during the term of the underlease the undertenant will:
    - (A) observe and perform all the provisions of the underlease to be observed and performed by the undertenant and
    - (B) not assign the underlease without the prior written consent of the Landlord (which will not be unreasonably withheld or delayed); or
  - (ii) (where the proposed undertenant is a corporate body and the Landlord reasonably so requires) without first procuring either covenants by deed with the Landlord in the form (mutatis mutandis) set out in Schedule 3 from two individuals who are or a corporate body which is acceptable to the Landlord as guarantor or an alternative form of security reasonably acceptable to the Landlord

- (g) To enforce and not to waive or vary the provisions of a Permitted Underlease and to operate at the relevant dates of review the rent review provisions contained in an underlease
- (h) Not to accept a surrender of part only of the premises demised by any Permitted Underlease without the prior written consent of the Landlord (which will not be unreasonably withheld)
- (i) Not to vary the terms of any Permitted Underlease without the prior written consent of the Landlord (not to be unreasonably withheld or delayed)
- (j) Without prejudice to the Tenant's ability to underlet Permitted Parts, not to grant underleases of areas for the location and use of scanning equipment and/or other special clinical or diagnostic services unless
  - (i) the Tenant has entered into an agreement with the proposed undertenant to carry out the scanning work and use the scanning equipment at the Premises; and
  - (ii) such underleases are:
    - (A) for a term of no more than 10 years;
    - (B) validly excluded from the provisions of sections 24 to 28 of the Landlord and Tenant Act;
    - (C) in respect of an area comprising no more than twenty per cent of the total floor area of the Premises; and
    - (D) otherwise comply with the requirements of this Lease in respect of underlettings
- (k) Notwithstanding any other provision of this Lease the Tenant may share occupation of the whole or part of the Premises with a Tenant Group Company in a manner which does not transfer or create a legal estate or create security of tenure or prevents vacant possession of the Premises being given:
  - (i) for so long as the party shares occupation it remains a Tenant Group Company and
  - (ii) within ten Business Days of the commencement of such sharing the Tenant notifies the Landlord of the Tenant Group Company and the address of its registered office and
  - (iii) within ten Business Days of the termination of such sharing the Tenant notifies the Landlord of such termination
- (I) Notwithstanding any other provisions of this Lease not to grant licences to occupy except for a period of up to 12 months to
  - (i) consultants, GPs, dentists, physiotherapists and their administrative and secretarial staff working at the Premises and

(ii) retail concessions

in each case in a manner which does not transfer or create a legal estate or create security of tenure nor prevent vacant possession of the Premises being given

(m) Within 20 Business Days of any assignment charge underlease or any transmission or other devolution relating to the Premises (or any extension, amendment or variation of an existing assignment charge underlease or any transmission or other devolution) to give written notice thereof to the Landlord's solicitors together with two certified copies of the relevant document and to pay the Landlord's solicitors' reasonable charges for the registration of every such document plus Value Added Tax

### 4.12 Landlord's Costs

To pay to the Landlord and to indemnify the Landlord against all proper (and reasonable in respect of paragraph (a) below) costs fees charges disbursements and expenses (including without prejudice to the generality of the above those payable to counsel solicitors surveyors and bailiffs or enforcement officers) incurred by the Landlord in relation to:

- (a) every application made by the Tenant for a consent approval or licence required by the provisions of this Lease unless the same is unlawfully or unreasonably withheld or delayed or offered subject to some unlawful or unreasonable condition
- (b) the service of a notice under section 146 or section 147 of the Law of Property Act 1925 or incurred by or in reasonable contemplation of proceedings under sections 146 or 147 of that Act notwithstanding that forfeiture is avoided otherwise than by relief granted by the court
- (c) the recovery or attempted recovery of arrears of the Rents or other sums due from the Tenant and the enforcement or remedy of any breach of the Tenant's covenants under this Lease
- (d) any steps taken in contemplation of or in connection with the preparation and service of a schedule of dilapidations during or after the expiry of the Term but which relates to dilapidations caused or occurring during the Term

### 4.13 The Planning Acts

- (a) Not to commit any breach of the Planning Acts and to comply with the provisions and requirements of the Planning Acts that affect the Premises whether as to the Permitted User or otherwise and to indemnify and keep the Landlord indemnified both during and after the expiry of the Term against all liability (including costs and expenses) incurred as a result of any breach occurring during the Term
- (b) At the expense of the Tenant to obtain all planning permissions and to serve all such notices as may be required for the carrying out of any operations or user on the Premises which may constitute Development (or for the reinstatement rebuilding or replacement of the Premises following damage by Insured Risks) PROVIDED THAT no application for planning permission shall be made without the prior written consent of the Landlord and such consent of the Landlord as required in this clause 4.13(b) is not to be unreasonably withheld or delayed where under the provisions of this Lease the consent of the Landlord is:
  - (i) not required; or

(ii) cannot be unreasonably withheld

in respect of the matters the subject of the application PROVIDED THAT notwithstanding the foregoing provisions the Landlord shall be entitled to withhold consent where the implementation of such planning permission would or would be likely to create or give rise to any tax or other fiscal liability for the Landlord and the Tenant fails to indemnify the Landlord against such liability having first been asked in writing to do so

- (c) Subject only to any statutory direction to the contrary to pay and satisfy any charge or levy that may subsequently be imposed under the Planning Acts in respect of the carrying out or maintenance of any such operations or the commencement or continuance of any such user
- (d) Notwithstanding any consent which may be granted by the Landlord under this Lease not to carry out or make any alteration or addition to the Premises or any change of use until:
  - (i) all necessary permissions and consents under or pursuant to the Planning Acts have been obtained and produced to the Landlord and
  - (ii) the Landlord (acting reasonably) has acknowledged that every necessary planning permission is acceptable to it the Landlord being entitled to refuse to acknowledge its acceptance of a planning permission only on the grounds that any condition contained in it or any period referred to in it would be or be likely to be materially prejudicial to the Landlord's interest in the Premises whether during or after the expiry of the Term
- (e) Unless the Landlord shall otherwise direct to carry out and complete before the expiry of the Term:
  - (i) any works stipulated to be carried out to the Premises by a date subsequent to such expiry as a condition of any planning permission granted for any Development commenced during the Term and
  - (ii) any Development commenced during the Term upon the Premises in respect of which the Landlord shall or may be or become liable for any charge or levy under the Planning Acts
- (f) Not to object to any application for planning permission that the Landlord or a Landlord Group Company may make whether jointly or alone in respect of any adjoining or neighbouring property except where such planning permission if granted would be adverse to the Tenant's legitimate business interests
- (g) In relation to community infrastructure levy (or any similar or replacement charge or levy) to:
  - (i) pay any community infrastructure levy (or any similar or replacement charge or levy);
  - (ii) serve a notice assuming liability (and provide a copy of such notice to the Landlord) and not withdraw it (or carry out such equivalent or similar steps as may be required or permitted in relation to any similar or replacement charge or levy); and

(iii) indemnify the Landlord against all liabilities arising out of community infrastructure levy (or any similar or replacement charge or levy),

in each case in respect or by reason of any works carried out at the Premises (except for works carried out by the Landlord)

(h) The obligations contained in clause 4.13(e) shall not apply at the expiry of the Term where the Reversionary Lease is subsisting at the relevant time

### 4.14 Indemnities

To be responsible for and to keep the Landlord fully indemnified against all damage damages losses costs expenses actions demands proceedings claims and liabilities made against or suffered or incurred by the Landlord arising directly or indirectly out of any act omission or negligence of the Tenant or any person at the Premises expressly or impliedly with the Tenant's authority or out of any breach or non-observance by the Tenant of the covenants conditions or other provisions of this Lease or any of the matters to which this demise is subject but:

- (a) the Landlord shall as soon as practicable notify the Tenant in writing of any claims and shall supply copies of relevant documents and correspondence;
- (b) the Landlord shall (at Tenant's cost) take reasonable commercial steps to mitigate the Landlord's liability in respect of such matter; and
- (c) the Landlord shall not admit any liability or settle or compromise any such matter without first obtaining the Tenant's approval in writing such approval not to be unreasonably withheld or delayed

### 4.15 Disposal Boards and Viewing

To permit the Landlord upon reasonable written notice:

- (a) at all reasonable times during the last six months of the Term (except where the Reversionary Lease is subsisting at the relevant time) and at any time thereafter to enter upon the Premises and affix and retain upon any reasonable external part of the Premises a notice of a reasonable size for re-letting the Premises and
- (b) at all reasonable times during the Term to enter upon the Premises and affix and retain upon any reasonable part of the Premises a notice of a reasonable size for the sale of the Landlord's reversion or of any superior interest

and during each such period or periods to permit upon 5 Business Days prior notice persons with the written authority of the Landlord or its agent at reasonable times of the day to view the Premises without interruption provided they are accompanied by the Tenant or its agent provided always that the Landlord shall in exercising those rights use reasonable endeavours to avoid causing disruption to the business carried on at the Premises

### 4.16 Encroachments

To take all necessary steps to prevent any new window light opening doorway path passage right or other encroachment or easement being made or acquired in against out of or upon the Premises and to notify the Landlord as soon as reasonably practicable in writing if any such encroachment or easement shall be made or acquired (or attempted to be made or acquired) and at the request of the Landlord to adopt such means as shall be required to prevent such encroachment or the acquisition of any such easement

### 4.17 Yield Up

At the expiry or sooner determination of the Term (except where the Reversionary Lease is subsisting at the relevant time):

- (a) to yield up the Premises in accordance with the terms of this Lease
- (b) to remove all placards signs notices fascias boards name-plates and advertisements fixed or exhibited by the Tenant in or upon the Premises and immediately to make good to the reasonable satisfaction of the Landlord any damage caused by such removal and
- (c) to give up all keys of the Premises to the Landlord
- (d) to provide the Landlord with a copy of the most recent asbestos survey and associated documents held by the Tenant pursuant to its duties under Regulation 4 of the Control of Asbestos at Work Regulations 2002 and to deliver to the Landlord all original health and safety files in accordance with the Construction (Design and Management) Regulations 2015

### 4.18 Interest on Arrears

- (a) If the Tenant shall fail to pay:
  - (i) the Rents (whether formally demanded or not); or
  - (ii) any other sum due under or pursuant to this Lease

within five Business Days for so long as BMI Healthcare Limited (No. 2164270) is the Tenant under this Lease or otherwise within two days of the date on which payment was due to pay to the Landlord Interest on the Rents or other such sum and such Interest shall be deemed to be and shall be recoverable as rent in arrear

- (b) That nothing in the preceding clause 4.18(a) shall entitle the Tenant to withhold or delay any payment of the Rents or any other sum due under this Lease after the date upon which they fall due or in any way prejudice affect or derogate from the rights of the Landlord in relation to such non-payment including (but not limited to) the proviso for re-entry contained in this Lease
- (c) That for the purposes of clause 4.18(a) sums shall be deemed to be unpaid where the Landlord properly refuses to accept payment to avoid waiving the right of forfeiture and/or waiving a breach of covenant on behalf of the Tenant

### 4.19 Statutory Notices

To give full particulars to the Landlord of any notice direction order or proposal for the Premises made given or issued to the Tenant by any local or public authority within ten (10) Business Days of receipt and if so required by the Landlord to produce it to the Landlord and without delay and at the cost of the Tenant to take all necessary steps to comply with such notice direction or order and at the request of the Landlord and the cost of the Tenant to make or join with the Landlord in making such objection or representation against or in respect of any notice direction order or proposal as the Landlord shall reasonably deem expedient provided that the Tenant shall not be required to join the Landlord in making such representation if to do so would be contrary to the Tenant's bona fide commercial interests

### 4.20 Defective Premises

To give notice to the Landlord of any defect in the Premises which might give rise to an obligation on the Landlord to do or refrain from doing any act or thing in order to comply with the duty of care imposed on the Landlord pursuant to the Defective Premises Act 1972 or otherwise and at all times to display and maintain all necessary notices which the Landlord may from time to time reasonably require to be displayed at the Premises

### 4.21 Compliance with Title

To perform observe and comply with the rights easements privileges restrictions covenants stipulations and other matters contained or referred to in Schedule 1

### 4.22 Consents

- (a) To procure that all such authorities registrations licences and consents as may be necessary for carrying on the actual use (being a Permitted User) of the Premises from time to time (the "Consents") are maintained in force and renewed and to pay all fees and duties in relation thereto
- (b) Not to make any alterations to the Premises or vary the operation of the business conducted at the Premises in accordance with the terms of this Lease without the consent (when required) of the CQC (as defined in Schedule 4) or other relevant dispensing power in respect of all Consents

### 4.23 Registration at the Land Registry

- (a) To use its reasonable endeavours to procure that:
  - (i) the Tenant is registered at the Land Registry as proprietor of this Lease as soon as practicable and
  - (ii) all rights granted or reserved by this Lease are properly noted against the affected titles and
  - (iii) deliver to the Landlord within ten days of registration official copies of the registered title evidencing registration of this Lease
- (b) Promptly on the expiry or sooner determination of the Term to apply to the Land Registry to close the registered title and to cancel the registration of all rights granted or reserved by this Lease and any note of this Lease registered against the Landlord's title

#### 4.24 **EPCs**

- (a) To allow the Landlord and all others authorised by the Landlord to have access to all documentation, data and information in the Tenant's possession or under its control that is reasonably required in order for the Landlord to:
  - (i) prepare an EPC in respect of the Premises; and
  - (ii) comply with any duty imposed upon the Landlord under The Energy Performance of Buildings (England and Wales) Regulations 2012,

and to co-operate with the Landlord and any person so authorised so far as is reasonably necessary to enable them to carry out such functions

- (b) Not to obtain or commission an EPC in respect of the Premises unless required to do so by The Energy Performance of Buildings (England and Wales) Regulations 2012 or otherwise with the prior consent of the Landlord such consent not to be unreasonably withheld or delayed.
- (c) If an EPC is required in respect of the Premises, to obtain such EPC from an assessor approved by the Landlord (acting reasonably) or to pay the Landlord's reasonable costs of obtaining such EPC
- (d) To provide free of charge to the Landlord a copy of any EPC that the Tenant obtains in respect of the Premises

### 4.25 Information

To provide the Landlord with the information specified in Schedule 4 at the times and intervals set out in Schedule 4.

### 5. THE LANDLORD'S COVENANTS

The Landlord covenants with the Tenant that:

- 5.1 the Landlord shall permit the Tenant peaceably and quietly to hold and enjoy the Premises during the Term without any interruption or disturbance from or by the Landlord or by any person lawfully claiming through under or in trust for the Landlord and
- 5.2 on the date of an assignment of this Lease in accordance with clause 4.11, the Landlord shall release and cancel the security constituted by the Senior Charge

### 6. **INSURANCE**

The Landlord shall insure the Premises in accordance with the provisions of Annex 2 of the Supplemental Deed and the Landlord and the Tenant shall each perform their respective obligations and conditions in Annex 2 of the Supplemental Deed.

### 7. PROVISOS

### 7.1 Re-entry

- (a) If and whenever during the Term:
  - (i) the Principal Rent (under this Lease or under any Related Lease) is outstanding for more than ten (10) Business Days (the Principal Rent Default Period) after becoming due whether formally demanded or not (provided that if any two or more payments of Principal Rent (whether under this Lease or a Related Lease) have been outstanding for more than ten (10) Business Days in the immediately preceding twelve month period then the Principal Rent Default Period shall be reduced to three (3) Business Days but only for so long as the foregoing proviso applies after which the Principal Rent Default Period will increase back to ten (10) Business Days) and provided that following a Permitted Assignment to an entity which is not a member of the Circle Group the above references will be limited to the Principal Rent under the Lease and not also the Principal Rent under any Related Lease or

- (ii) the Rents (or any part of them) under this Lease (other than the Principal Rent) is outstanding for more than ten (10) Business Days (the **Rents Default Period**) after becoming due and written demand having been made (provided that if any two or more payments of the Rents other than the Principal Rent (whether under this Lease or a Related Lease) have been outstanding for more than ten (10) Business Days in the immediately preceding twelve month period then the Rents Default Period shall be reduced to three (3) Business Days but only for so long as the foregoing proviso applies after which the Rents Default Period will increase back to ten (10) Business Days) and provided that following a Permitted Assignment to an entity which is not a member of the Circle Group the above references will be limited to the Rents (other than the Principal Rent) under the Lease and not also to the Rents (other than the Principal Rent) under any Related Lease or
- (iii) there is a breach by the Tenant of any covenant or other term of this Lease or any document expressed to be supplemental to this Lease or
- (iv) the Tenant or the Surety commits or permits an Act of Insolvency

then subject to clause 7.1(b) and (c) the Landlord may at any time prior (in the case of (i), (ii) and (iii) above) to such Principal Rent or Rents being paid or the breach having been made good re-enter the Premises or any part of them in the name of the whole at any time and even if any previous right of re-entry has been waived and then the Term will absolutely cease but without prejudice to any rights or remedies which may have accrued to the Landlord against the Tenant in respect of any breach of covenant or other term of this Lease including the breach in respect of which the reentry is made

(v)

- (A) (1) six or more leases (which were, when terminated, Related Leases) have been terminated and/or (in any combination) (2) the relevant landlord has a right and intention to terminate six or more such Related Leases (and for this purpose only the landlord shall, in addition to the rights set out in clause 7.1(a), be deemed also to have a right to terminate if the Tenant (or the tenant of a Related Lease) is in breach of Clause 11.2 (or the equivalent provision in the Related Lease)), provided that, for the purposes of this clause only, the right of the relevant landlord to terminate any such Related Leases shall, in relation to a breach of clause 4.3, only arise where the Tenant is also in breach of its obligations under clause 4.10(c);
- (B) (1) leases (which were, when terminated, Related Leases) have been terminated and/or (in any combination) (2) the relevant landlord has a right and intention to terminate such Related Leases (and for this purpose only the landlord shall, in addition to the rights set out in clause 7.1(a), be deemed also to have a right to terminate if the Tenant (or the tenant of a Related Lease) is in breach of Clause 11.2 (or the equivalent provision in the Related Lease)) and the aggregate Principal Rent payable (or which would, but for such termination, have been payable) under such leases is equal to or greater than 25% of the aggregate Principal Rent payable (or which would, but for such termination, have been payable) under all leases which were, at the date of the Supplemental Deed, Related Leases; or

(C) (1) leases (which were, when terminated, Related Leases) have been terminated and/or (in any combination) (2) the relevant landlord has a right and intention to terminate such Related Leases (and for this purpose only the landlord shall, in addition to the rights set out in clause 7.1(a), be deemed also to have a right to terminate if the Tenant (or the tenant of a Related Lease) is in breach of Clause 11.2 (or the equivalent provision in the Related Lease)) and the aggregate Tenant EBITDAR earned (or which would, but for such termination, have been earned) by the tenant of such leases is equal to or greater than 25% of the aggregate Tenant EBITDAR earned (or which would, but for such termination, have been earned) by the tenants of all leases which were, at the date of the Supplemental Deed, Related Leases.

Provided that termination shall (i) where the Related Lease relates to property located in England or Wales not be permitted if the Landlord has terminated and the Tenant has applied for relief against re-entry or forfeiture until the court has finally disposed of the Tenant's application and (ii) where the Related Lease relates to property located in Scotland only occur where the Landlord has terminated by serving on the Tenant a notice of irritancy and a Scottish court has granted decree in an action for declarator of irritancy at the instance of the Landlord against the Tenant.

And provided further that Clause 7.1(a)(v) shall no longer apply following any Permitted Assignment to an entity which is not a member of the Circle Group.

- (b) The Landlord shall not exercise the right of re-entry pursuant to clause 7.1(i) or (ii) without first giving not less than ten (10) Business Days' written notice (the **Notice Period**) to the Tenant of its intention to exercise such right of re-entry such notice to provide full details of all amounts of Principal Rent and other Rents outstanding and the relevant lease or leases under which such amounts are due (whether such lease or leases be this Lease or a Related Lease) provided that if any two or more payments of Principal Rent or other Rents (whether under this Lease or a Related Lease or Related Leases) have been outstanding for more than ten (10) Business Days in the immediately preceding twelve month period then the Notice Period shall be reduced to three (3) Business Days but only for so long as the foregoing proviso applies after which the Notice Period will increase back to ten (10) Business Days
- (c) The Landlord shall give written notice (including, without limitation, the same period of written notice as detailed at clause 7.1(b) regarding non-payment of the Principal Rent or other Rents) to any mortgagee of the Tenant of which the Landlord has prior written notice prior to exercising the right of re-entry contained in this Lease.

### 7.2 Rights and Easements

The operation of section 62 of the Law of Property Act 1925 shall be excluded from this Lease and save as expressly hereby granted the Tenant shall not by virtue of this Lease be deemed to have acquired or be entitled to and the Tenant shall not during the Term acquire or become entitled to by any means whatever any easement from or over or affecting any other land or premises now or at any time after the date of this Lease belonging to the Landlord or any Group Company of the Landlord and not comprised in this Lease

### 7.3 Exclusion of Use Warranty

Nothing in this Lease or in any consent granted by the Landlord under this Lease shall imply or warrant that the Premises may lawfully be used under the Planning Acts for the purpose authorised in this Lease or any purpose subsequently authorised

### 7.4 Representations

The Tenant acknowledges that this Lease has not been entered into in reliance wholly or partly on any statement or representation made by or on behalf of the Landlord except any such statement or representation that is expressly set out in this Lease

### 7.5 **Tenant's Property**

If after the Tenant has vacated the Premises on the expiry of the Term any property of the Tenant remains in or on the Premises and the Tenant fails to remove it within 21 Business Days after being requested in writing by the Landlord to do so or if after using all reasonable endeavours the Landlord is unable to make such a request to the Tenant within 28 Business Days from the first attempt so made by the Landlord:

- (a) the Landlord may as the agent of the Tenant sell such property
- (b) the Landlord shall (subject to clause 7.5(c) below) forthwith after such sale pay to the Tenant the proceeds of such sale after having deducted the reasonable and proper fees and expenses incurred by or on behalf of the Landlord in connection with such sale
- (c) if the Landlord having made reasonable efforts is unable to locate the Tenant the Landlord shall be entitled to retain such proceeds of sale absolutely unless the Tenant shall claim them within six months of the date on which the Tenant vacated the Premises and
- (d) the Tenant shall indemnify the Landlord against any damage occasioned to the Premises and any actions claims proceedings costs expenses and demands made against the Landlord caused by or related to the presence of the property in or on the Premises

### 7.6 Compensation on Vacating

Any statutory right of the Tenant to claim compensation from the Landlord on vacating the Premises shall be excluded to the extent that the law allows

### 7.7 Loss or Damage

Save where any such matters arise from their wilful or negligent act or default the Landlord its servants agents visitors or licensees the Landlord shall not be liable or responsible to the Tenant or to any person at the Premises with the express or implied consent of the Tenant for any loss injury damage nuisance annoyance or inconvenience which may be sustained by the Tenant or such person either personally or to their property including the Premises

### 7.8 Service of Notices

- (a) Any demand or notice required to be made given to or served on any party under this Lease shall be in writing and validly made given or served if addressed to such party and delivered personally or sent by pre-paid registered or recorded delivery mail addressed (in the case of a company) to its registered office or (whether a company or individual) its last known address, and in the case of notices served on the Tenant, marked for the attention of the Company Secretary
- (b) Any notice or document shall also be sufficiently served if sent by any means of electronic transmission other than facsimile transmission to the party to be served and that service shall be deemed to be made on the day of transmission if transmitted before 4.00 p.m. on a Business Day but otherwise on the next following Business Day

- (c) For so long as BMI Healthcare Limited (No. 2164270) is the Tenant of this Lease, a notice served on the Tenant by electronic transmission shall be by means of e-mail to <a href="mailto:company.secretary@bmihealthcare.co.uk">company.secretary@bmihealthcare.co.uk</a> or such other e-mail address as is otherwise specified by the Tenant by notice in writing to the Landlord
- (d) For so long as [Name of MPT entity] is the Landlord of this Lease, a notice served on the Landlord by electronic transmission shall be by means of e-mail to <a href="mailto:david@duggins.co">david@duggins.co</a> or such other e-mail address as is otherwise specified by the Landlord by notice in writing to the Tenant

#### 7.9 Value Added Tax

- (a) Save as the context requires or as otherwise stated (but subject to the provisions of clause 7.9(e)) all references to payments made in this Lease are references to such payments exclusive of Value Added Tax chargeable in respect of the supply of goods or services for which the payment is consideration and insofar as such payments fall to be made under this Lease the amount of such Value Added Tax shall be paid in addition thereto
- (b) Without prejudice to and save as mentioned earlier in this clause 7.9 (but subject to the provisions of clause 7.9(e)) where any supply is made pursuant to this Lease the recipient of the supply shall pay to the supplier at the same time as any other consideration for such supply the amount of any Value Added Tax chargeable in respect thereof
- (c) Where any payment is required to be made pursuant to this Lease to reimburse the payee for any expenditure incurred by the payee such payment shall include an amount equal to any Value Added Tax comprised in that expenditure which is not recoverable by the payee as input tax under section 25 of the Value Added Tax Act 1994
- (d) Where any taxable supply is made pursuant to this Lease the maker of such supply shall provide a valid VAT invoice in respect thereof
- (e) The Landlord warrants and undertakes to the Tenant that:
  - (i) neither it nor any relevant associate (to be construed in accordance with paragraph 3 of Schedule 10 to the Value Added Tax Act 1994) of it has exercised the option to tax for the purposes of Schedule 10 to the Value Added Tax Act 1994 in relation to the Premises and neither it nor any relevant group member (to be construed in accordance with paragraph 21 of Schedule 10 to the Value Added Tax Act 1994) has made a real estate election for the purposes of paragraph 21 of Schedule 10 to the Value Added Tax Act 1994; and
  - (ii) neither it nor any relevant associate (to be construed in accordance with paragraph 3 of Schedule 10 to the Value Added Tax Act 1994) will exercise the option to tax for the purposes of Schedule 10 to the Value Added Tax Act of 1994 in relation to the Premises and neither it nor any relevant group member (to be construed in accordance with paragraph 21 of Schedule 10 to the Value Added Tax Act 1994) will make a real estate election for the purposes of paragraph 21 of Schedule 10 to the Value Added Tax Act 1994, and neither it, such relevant associate or such relevant group member will take any equivalent action having effect under any legislation which is the successor to Schedule 10 to the Value Added Tax Act 1994 unless:
    - (A) it is required by law to do so or

- (B) the Tenant has agreed in writing in advance that it may do so
- (f) Any reference in this Lease to any person shall in relation to VAT (where appropriate and unless the context otherwise requires) be construed at any time when such person is treated as a member of a VAT group for the purposes of sections 43 to 43D of the Value Added Tax Act 1994 to include a reference to the representative member of such group at such time
- (g) Any reference in this Lease to any person shall in relation to VAT (where appropriate and unless the context otherwise requires) be construed at any time when the grant of an interest in right over or licence to occupy the Premises would if or when made by that person be treated pursuant to paragraph 8 of Schedule 10 to the Value Added Tax Act 1994 as having been made by any other person to include a reference to such other person (the term "the grant of an interest in right over or licence to occupy" has the same meaning as for the purposes of the said paragraph 8 of the said Schedule 10)

#### 7.10 Proper Law and Jurisdiction

- (a) This Lease (including an non-contractual obligations arising out of, or in connection with, this Lease) shall be governed by and construed in accordance with English law
- (b) Each of the parties to this Lease irrevocably agrees that the courts of England shall have exclusive jurisdiction to hear and decide any suit action or proceedings and/or to settle any disputes which may arise out of or in connection with this Lease and for these purposes each party irrevocably submits to the jurisdiction of the courts of England

#### 7.11 Exclusion of the Landlord and Tenant Act 1954

- (a) In relation to the tenancy created by this Lease:
  - (i) the Landlord served on the Tenant a notice dated 27 July 2006 complying with requirements of section 38A(3) of the Landlord and Tenant Act
  - (ii) the Tenant or a person duly authorised by the Tenant made a statutory declaration (the "**Statutory Declaration**") complying with the requirements of schedule 2 to the Regulatory Reform (Business Tenancies) (England and Wales) Order 2003
- (b) Where the Statutory Declaration was made by a person other than the Tenant the Tenant confirms that the declarant was duly authorised to make the Statutory Declaration on the Tenant's behalf
- (c) The Landlord and the Tenant agree that the provisions of sections 24 to 28 (inclusive) of the Landlord and Tenant Act shall be excluded in relation to the tenancy created by this Lease

#### 7.12 Landlord and Tenant (Covenants) Act 1995

This Lease is a "new tenancy" as that expression is defined in the Landlord and Tenant (Covenants) Act 1995

### 7.13 **Invalidity**

If at any time any one or more provisions of this Lease is or becomes invalid illegal or unenforceable in any respect under any law the validity legality and enforceability of the remaining provisions hereof shall not be in any way affected or impaired thereby

#### 8. **CONFIDENTIALITY**

The Landlord and the Tenant each covenant and agree to keep all Confidential Information confidential and no party shall disclose any Confidential Information to any person, except

- 8.1 the Landlord shall be permitted to disclose Confidential Information:
  - (a) with the prior written approval of the Tenant;
  - (b) to its Affiliates and its and their Representatives;
  - (c) to its direct and indirect shareholders and their respective Affiliates and Representatives;
  - (d) to its prospective investors in relation to or in connection with any private placement or public offering of a member of the MPT Group's securities

in each case described in clauses 8.1(a) to 8.1(d) above, provided that any person to whom the Confidential Information is to be disclosed is informed of its confidential nature and agrees to keep it confidential before that disclosure;

- 8.2 the Landlord shall be permitted to disclose Confidential Information:
  - (a) to its existing lenders and their agents and Advisers (including, in respect of the Landlord, the servicer) and any lenders and their agents and Advisers under any replacement debt financing sources, to the extent that such disclosure is necessary or desirable in connection with its relevant financing arrangements;
  - (b) to third parties in connection with the potential sale or transfer of the Landlord's interest in this Lease, or the sale (directly or indirectly) of the Landlord, provided that the third party has first signed a non-disclosure agreement with the Landlord which contains equivalent terms to the confidentiality undertakings contained in this Lease;
- 8.3 each party shall be permitted to disclose Confidential Information:
  - (a) to a Tax Authority or to a professional Tax adviser where reasonably required in connection with the Tax affairs of the disclosing party (or its Affiliates);
  - (b) if requested or required by law, any court of competent jurisdiction or any competent judicial, governmental, supervisory or regulatory body or the rules of any relevant stock exchange, provided that the party receiving such request will, to the extent practicable and legally permissible under the circumstances, and other than with respect to a request or requirement relating to routine audit or examination by a regulatory or self-regulatory authority, examiner or auditor:
    - (i) provide the other parties with prompt written notice of such requirement(s);
    - (ii) reasonably cooperate with the other parties and take into account the reasonable requests of the other parties in relation to the content of such disclosure;
    - (iii) if the party is legally advised by counsel (which may be internal counsel) to disclose Confidential Information to any court, tribunal or government or regulatory body, use reasonable efforts to request assurances that the Confidential Information will be treated confidentially by the recipient

#### 9. THE CONTRACTS (RIGHTS OF THIRD PARTIES) ACT 1999

A person who is not a party to this Lease shall have no rights under the Contracts (Rights of Third Parties) Act 1999 to enforce any of its terms

#### 10. LANDLORD'S OPTION TO PURCHASE THE TENANT'S ASSETS

The Tenant hereby grants the Landlord an option in the terms set out in Schedule 6 and the Landlord and the Tenant agree to comply with the provisions of Schedule 6

#### 11. FINANCIAL COVENANTS

- 11.1 In this Clause 11 the following words and expressions shall have the following meanings:
  - (a) "Birmingham Lease" means the lease of both acute and rehabilitation facilities to be entered into between MPT Circle Birmingham S.A.R.L, Circle Birmingham Limited and Circle Holdings Plc in accordance with the Landlord Works Agreement for Underlease between MPT Circle Birmingham S.A.R.L, Circle Birmingham Limited, Circle Holdings Plc and MPT Operating Partnership L.P. dated 13 April 2017;
  - (b) "Capitalised Maintenance Costs" means routine, non-major expenditures that are capitalised by the relevant Circle Group entity and would be capable of classification as either capital expenditure or income statement expense under UK Generally Accepted Accounting Principles applicable at the date of calculation. Such expenditures include exterior and interior painting and decorating, resurfacing building walls and floors, and similar types of expenditure applied to maintaining the appearance of the Premises. Such expenditure does not include any non-routine capital expenditure, major repairs or alterations, improvements or other expenditures related to plant and equipment or fixtures and fittings (e.g., power, heating and water systems, IT and communications, scanners and theatre / ward equipment etc) or capital expenditures relating to new developments or expansionary or growth capital items.
  - (c) "Consolidated Fixed Charges" means, for any Relevant Period, for the Circle Group on a consolidated basis, an amount equal to the sum for such Relevant Period of (a) Rent Payments, plus, (b) Consolidated Interest Charges, plus (c) consolidated Capitalised Maintenance Costs incurred after the third anniversary of the EBITDAR Calculation Commencement Date, plus (d) Consolidated Funded Debt Payments but excluding Disregarded Items;
  - (d) "Consolidated Interest Charges" means, for any Relevant Period, the aggregate of:
    - (i) the interest (but not the capital) element of any payments in respect of any Finance Leases of members of the Circle Group payable in such Relevant Period; and
    - (ii) accrued interest in respect of all debt facilities of the Circle Group payable in such Relevant Period,

after deducting any interest received in that Relevant Period by any member of the Circle Group on cash.

- (e) "Consolidated Funded Debt Payments" means, for any Relevant Period, the aggregate of:
  - (i) the capital element of any payments in respect of any Finance Leases of members of the Circle Group payable in such Relevant Period; and

- (ii) all scheduled repayments (save for any final bullet payment) in respect of all debt facilities of the Circle Group payable in such Relevant Period excluding any amounts falling due under any overdraft, revolving facility or intra-group payments.
- (f) "Disregarded Items" means any items which would otherwise qualify for inclusion in the calculation of Consolidated Fixed Charges, but which arise in relation to:
  - (i) The South Cheshire Private Hospital, Cheshire in respect of which the tenant is South Cheshire SPV Limited and for which the lease has Land Registry title number CH318155;
  - (ii) Carrick Glen Hospital, Ayrshire in respect of which the tenant is BMI Healthcare Limited and for which the lease is registered in the Land Register of Scotland under title number AYR89430;
  - (iii) The Sandringham Hospital, Norfolk in respect of which the tenant is BMI Healthcare Limited and for which the lease has Land Registry title number NK156178;
  - (iv) Gisburne Park Hospital, Lancashire in respect of which the tenant is BMI Healthcare Limited and for which the lease has Land Registry title number LAN104242;
  - (v) any premises from which operations have ceased (whether before or after the date of the Supplemental Deed) but not where such premises is held under a Related Lease and only for a period of 12 months from the date of the cessation of operations; and
  - (vi) the Birmingham Lease in the period prior to "the third EBITDAR Calculation Commencement Date following the Initial EBITDAR Calculation Date" as described in clause 7.2(a)(iv) of the Birmingham Lease.
- (g) "EBITDAR Calculation Commencement Date" means the first EBITDAR Calculation Date immediately following the first anniversary of the date of the Supplemental Deed;
- (h) "EBITDAR Calculation Dates" means 31 March, 30 June, 30 September and 31 December in every year and EBITDAR Calculation Date means any of them;
- (i) "EBITDAR Rent" means in respect of any Relevant Period the Principal Rent;
- (j) "Exceptional Items" means any exceptional, one-off, non-trading, non-recurring or extraordinary items or any other items of an unusual or non-recurring nature which represent gains or losses including, but not limited to, historical and future costs incurred in connection with the closure of or ceasing of operations, costs incurred to achieve synergy savings and to integrate the businesses of the Tenant (at the date of the Supplemental Deed) and the Circle Group, and any costs incurred in relation to reorganisations, acquisitions, disposals and other business combinations, restructurings or strategic reviews.
- (k) "Finance Leases" means any lease or hire purchase agreement which in accordance with the accounting principles of the Circle Group is treated as a balance sheet liability (other than a lease or hire purchase contract which would in accordance with accounting principles in force prior to 1 January 2019, have been treated as an operating lease).

- "Group EBITDAR" means consolidated earnings before interest taxes depreciation amortisation and rent, in each case
  of the Circle Group, but excluding any Exceptional Items;
- (m) "Portfolio EBITDAR" means consolidated earnings before interest taxes depreciation amortisation and rent, in each case of each entity which is a tenant of a Related Lease, but excluding any Exceptional Items;
- (n) "Portfolio Rent" means, in respect of any Relevant Period, the aggregate Principal Rents payable under all Related Leases;
- (o) "Relevant Period" means each period of 12 months ending on each EBITDAR Calculation Date; and
- (p) "Rent Payments" means, in respect of any Relevant Period, the aggregate of all amounts payable by any member of the Circle Group in connection with any lettings, licences or grants of other rights of use or occupation, including but not limited to (if applicable) each of the following amounts:
  - (i) rent, licence fees and equivalent amounts payable;
  - (ii) any other moneys payable in respect of occupation and/or usage of a property and any fixture and fitting on that property including any fixture or fitting on that property for display or advertisement, on licence or otherwise:
  - (iii) any sum payable under any policy of insurance in respect of loss of rent or interest on rent;
  - (iv) any sum payable, or the value of any consideration given, for the grant, surrender, amendment, supplement, waiver, extension or release of any document under which a member of the Circle Group has a letting, licence or grant of other rights of or use or occupation;
  - (v) any sum payable in respect of a breach of covenant or dilapidations under any lease;
  - (vi) any sum payable by or distribution received or receivable from any guarantor of any occupational tenant under any lease;
  - (vii) any contributions payable to any landlord by way of VAT or a contribution to (i) ground rent; (ii) insurance premia; (iii) the cost of an insurance valuation; (iv) a service or other charge; or (v) a reserve or sinking fund; and
  - (viii) any interest payable on, and any damages, compensation or settlement payable in respect of, any sum referred to above.
- (q) "Tenant EBITDAR" means earnings before interest taxes depreciation amortisation and rent, in each case of the Tenant, but excluding any Exceptional Items.

## **Property Level EBITDAR Test**

For the purpose of clause 7.1(a)(v) of the Lease only and clause 7.1(a)(vii) of the Reversionary Lease only and for no other purpose, including for the avoidance of doubt, not so as to create a covenant in respect of which clause 7.1(a) would otherwise apply the Tenant covenants to maintain on each EBITDAR Calculation Date that occurs after the EBITDAR Calculation Commencement Date a Tenant EBITDAR to EBITDAR Rent cover ratio of at least 1.1:1, calculated by reference to the Relevant Period.

#### **Portfolio Level EBITDAR Test**

11.3 If on any EBITDAR Calculation Date that occurs after the EBITDAR Calculation Commencement Date, the Portfolio EBITDAR to Portfolio Rent cover ratio falls below 1.1:1 calculated by reference to the Relevant Period, the Landlord shall be entitled to exercise its rights under Clause 7.1(a), subject always to the provisions of Clause 11.5.

#### Portfolio Level Consolidated Fixed Charges Test

11.4 If on any EBITDAR Calculation Date that occurs after the EBITDAR Calculation Commencement Date, the Group EBITDAR to Consolidated Fixed Charges cover ratio falls below 1.1:1 calculated by reference to the Relevant Period, the Landlord shall be entitled to exercise its rights under Clause 7.1(a) subject always to the provisions of Clause 11.5.

### **Equity Cure Rights**

- 11.5 If:
  - (a) the Portfolio EBITDAR to Portfolio Rent cover ratio falls below the level required by clause 11.3; or
  - (b) the Group EBITDAR to Consolidated Fixed Charges cover ratio falls below the level required by clause 11.4; or
  - (c) the Tenant is in breach of clause 11.2 for the purposes of clause 7.1(a)(v) of the Lease or clause 7.1(a)(vii) of the Reversionary Lease,

the Landlord shall be entitled to exercise its rights in clause 7.1(a) arising following any failure to comply with the relevant provisions of this clause provided always that the Landlord:

- (d) shall have notified the Tenant and any mortgagee of the Tenant of which the Landlord has prior written notice of its intention to exercise its rights in clause 7.1(a); and
- (e) shall have granted the Tenant a grace period of twenty five (25) Business Days from such notice to allow for a cure in accordance with clause 11.6.

For the avoidance of doubt, the Landlord shall have no right to and shall not be entitled to exercise its rights under clause 7.1(a) pursuant to the provisions of any of (1) clause 7.1(a)(v) of the Lease or clause 7.1(a)(vii) of the Reversionary Lease as regards breach of clause 11.2, (2) 11.3 and/or (3) 11.4, until it has served notice as required by clause 11.5(d) and allowed the grace period provided for in clause 11.5(e) and the Tenant has failed to comply with the relevant provisions of this clause within such grace period.

- 11.6 If Circle Health Holdings Limited receives new common or preferred equity (whether in the form of an increase in share capital or subordinated shareholder loans) ("Additional Equity") the relevant cover ratio shall be recalculated, such recalculation giving effect to an increase of Tenant EBITDAR, Portfolio EBITDAR and/or Group EBITDAR (as applicable) by an amount equal to the Additional Equity:
  - (a) for the relevant EBITDAR Calculation Date at which the breach of clause 11.2, 11.3 and/or 11.4 (as applicable) arose and on the following three EBITDAR Calculation Dates, if the Additional Equity is provided within twenty five (25) Business Days after the relevant written notice served by the Landlord in accordance with clause 11.5(d); or

(b) for each of the four EBITDAR Calculation Dates immediately following the date the Additional Equity is provided, if the Additional Equity is provided in the three months before the EBITDAR Calculation Date at which the breach of clause 11.2, 11.3 and/or 11.4 (as applicable) arose,

and if the cover ratio is then shown to be in compliance on each such EBITDAR Calculation Date no breach shall be deemed to have occurred.

- 11.7 If, following an equity cure in accordance with clause 11.6 (an "Equity Cure"), there is no breach of clause 11.2, 11.3 or 11.4 on any two consecutive EBITDAR Calculation Dates following the EBITDAR Calculation Date in respect of which the Additional Equity was first applied then the shares may be redeemed (in the case of Additional Equity provided through an increase in share capital) and/or the loan repaid (in the case of Additional Equity provided through a subordinated shareholder loan). Any earlier redemption or repayment shall cause a breach of clause 11.2, 11.3 and/or 11.4 (as applicable).
- 11.8 For the avoidance of doubt, an Equity Cure may only be made in connection with the failure to comply with the requirements of Clauses 11.2, 11.3 and 11.4, and no Equity Cure shall be available in connection with any other breach of this Lease.
- 11.9 Additional Equity may only be provided for the purposes of the Equity Cure:
  - (a) a total of ten times during the aggregate term of the Lease and the Reversionary Lease; and
  - (b) in respect of two consecutive EBITDAR Calculation Dates.

#### **Information Provision**

11.10 The Tenant covenants with the Landlord to provide (or procure the provision) to the Landlord such information relating to the calculation of Tenant EBITDAR, Portfolio EBITDAR, Group EBITDAR and Consolidated Fixed Charges as the Landlord may reasonably request in connection with its review of Tenant EBITDAR, Portfolio EBITDAR, Group EBITDAR and/or Consolidated Fixed Charges, including (if the Landlord so requires) certification by an independent accountant appointed for such purposes by and at the cost of the Landlord.

#### **Permitted Assignment**

11.11 The provisions of Clause 11 will no longer apply following any Permitted Assignment to an entity which is not a member of the Circle Group.

IN WITNESS whereof this Lease has been executed as a deed on the date first above written

# Schedule 1 Particulars of Matters to which the Premises are Subject and have the benefit of

All and any wayleaves licences tenancies and other occupational arrangements entered into with medical practitioners and other occupiers granted or entered into by the Landlord or its predecessors in title and subsisting as at the date of this Lease

# Schedule 2 Rent Review Provisions

1. In this Schedule 2 the following expressions have the following meanings:

A means the figure shown in the Index for the month preceding the previous Review Date (or in respect of the first Review Date the month of December 2019 and where paragraph 9.2 of this Schedule 2 applies as such figure may be determined in accordance with paragraph 9.2 of this Schedule 2;

**B** means the figure shown in the Index for the month preceding the relevant Review Date and where paragraph 9.2 of this Schedule 2 applies as such figure may be determined in accordance with paragraph 9.2 of this Schedule 2;

Base Rent means the Reviewed Rent immediately preceding the relevant Review Date (ignoring any suspension of the whole or any part of the Reviewed Rent) and which, for the purposes of the first Review Date, will be the Initial Rent;

Deferred Rent means the amount calculated in accordance with paragraph 2 of this Schedule 2;

**Index** means the Retail Price Index All Items (or the Consumer Price Index if the Tenant exercises the option set out in paragraph 9.1 of this Schedule 2) as such Retail Price Index All Items or Consumer Price Index (as the case may be) is published by the Office for National Statistics or any successor ministry or department;

Indexed Rent means the annual rent calculated by multiplying the Base Rent by B and dividing that by A;

**Initial Rent** means £[Property details];

Principal Rent means: (i) from and including 8 January 2020 to and including 30 October 2021, the Initial Rent; and (i) from and including 31 October 2021, an amount equal to the Reviewed Rent plus the Deferred Rent;

**Reviewed Rent** means from and including each Review Date the amount calculated in accordance with paragraph 3.2 of this Schedule 2 and where paragraph 9.2 of this Schedule 2 applies as such Reviewed Rent may be determined in accordance with paragraph 9.2 of this Schedule 2, and

Review Date means 31 October 2021 and thereafter each anniversary of such date.

- 2. Deferred Rent is to be calculated on the following basis:
- 2.1 D = (W X)/28

where

- D: Deferred Rent.
- W: the Initial Rent multiplied by Z and divided by Y, Provided that:
  - (i) if this calculation produces a figure which shall be less than one hundred and two per cent (102%) of the Initial Rent then W shall be one hundred and two per cent (102%) of the Initial Rent unless the period of indexation on any given occasion is more or less than one year whereupon the 102% figure will be either increased or decreased (as appropriate) on a pro rata basis when undertaking the foregoing collar calculation; and

- (ii) if this calculation produces a figure which shall be more than one hundred and five per cent (105%) of the Initial Rent then W shall be one hundred and five per cent (105%) of the Initial Rent unless the period of indexation on any given occasion is more or less than one year whereupon the 105% figure will be either increased or decreased (as appropriate) on a pro rata basis when undertaking the foregoing cap calculation.
- X: the Initial Rent,
- Y: the figure shown in the Index for the month of December 2019,
- Z: the figure shown in the Index for the month of September 2020,

#### 3. Rent reviews are to take place on the following basis:

- 3.1 The Reviewed Rent shall be reviewed on each and every Review Date.
- 3.2 The Reviewed Rent from each Review Date shall be the Indexed Rent Provided that:
  - (a) if the calculation set out in the definition of Indexed Rent produces a figure which shall be less than one hundred and two per cent (102%) of the applicable Base Rent then the Indexed Rent shall be one hundred and two per cent (102%) of the applicable Base Rent unless the period of indexation on any given occasion is more or less than one year whereupon the 102% figure will be either increased or decreased (as appropriate) on a pro rata basis when undertaking the foregoing collar calculation; and
  - (b) if the calculation set out in the definition of Indexed Rent produces a figure which shall be more than one hundred and five per cent (105%) of the applicable Base Rent then the Indexed Rent shall be one hundred and five per cent (105%) of the applicable Base Rent unless the period of indexation on any given occasion is more or less than one year whereupon the 105% figure will be either increased or decreased (as appropriate) on a pro rata basis when undertaking the foregoing cap calculation.

### 4. Changes to the Index and Disputes

- 4.1 If the basis for the calculation of the Index is changed and any method of reconciliation between the new and the old figures are officially published that method will be used for the purpose of assessing the Deferred Rent and the Reviewed Rent under this Schedule 2.
- 4.2 If the Index ceases to exist then the parties will agree upon an alternative index for determining the rent review being as similar as possible to the Index and then the substituted index will be used in place of the Index for the purpose of assessing the Deferred Rent and the Reviewed Rent under this Schedule 2.
- 4.3 If there is a dispute between the parties by reason of failure to agree upon:
  - (a) any change in the methods used to complete or compile the Index;
  - (b) an alternative method of assessing the Deferred Rent or the Reviewed Rent because the Index ceases to exist;

- (c) the amount of the Deferred Rent or the Reviewed Rent; or
- (d) the construction or effect of this Schedule 2

then such dispute is to be submitted to a single arbitrator by whose decision the parties to this Lease are to be bound who is to be appointed by the parties jointly or if they do not agree on the appointment by the President for the time being of the Royal Institution of Chartered Surveyors (at the request of either party) and the arbitration is to be conducted under the Arbitration Act 1996.

- 4.4 If the Reviewed Rent has not been agreed or determined by the arbitrator before the relevant Review Date (including if such non-agreement or non-determination is as a result of the Supplemental Deed not being completed prior to the Review Date):
  - (a) the Tenant will continue to pay the Reviewed Rent at the rate applicable immediately prior to the relevant Review Date; and
  - (b) when the new Reviewed Rent is ascertained the Tenant will pay within ten (10) Business Days after the date of ascertainment:
    - (i) any difference between the Reviewed Rent payable before the relevant Review Date and the new Reviewed Rent for the period from and including the relevant Review Date to the Payment Day next following the date of ascertainment (the **Shortfall**); and
    - (ii) interest at the rate of 3% below the Interest Rate on the amount of the Shortfall such interest to be payable from the date the relevant proportion of the Shortfall would have fallen due had the new Reviewed Rent been agreed or determined up to the date of payment.

#### 5. Memorandum

When the Reviewed Rent and the Deferred Rent are agreed or determined the Landlord and the Tenant will exchange signed memoranda recording the Reviewed Rent and the Deferred Rent.

#### 6. Intermediate Rent Review

If at any Review Date there is a statutory restriction on the Landlord's right to calculate the Deferred Rent or review the Reviewed Rent then upon the ending or removal of such restriction the Landlord may within one year thereafter give to the Tenant not less than one (1) months' notice requiring calculation of the Deferred Rent and/or an additional rent review upon the succeeding Payment Day which Payment Day will for the purposes of this Lease be a Review Date.

#### 7. Timing

Time is not of the essence in relation to this Schedule 2.

#### 8. Deferred Rent Shortfall

If this Lease is determined prior to the expiry of its contractual term then, within 10 Business Days of such determination the Tenant shall pay to the Landlord an amount equal to 28 times the Deferred Rent payable pursuant to this Lease less the aggregate of all Deferred Rents actually paid by the Tenant pursuant to this Lease prior to the date of determination. The provisions of this paragraph 8 shall survive any determination of this Lease.

### 9. Tenant option to switch Index

- 9.1 The Tenant shall have an option ("**the CPI Option**") exercisable at any time by giving at least three months' advance written notice to the Landlord and to replace the Retail Price Index All Items with the Consumer Price Index as the Index with effect from the next occurring Review Date under the Lease ("**the CPI Start Date**").
- 9.2 In the event that the CPI Option is exercised by the Tenant then for the year commencing on the CPI Start Date and every year thereafter under the Lease the relevant figures in the Consumer Price Index will be used when determining A and B and when calculating the Reviewed Rent in accordance with paragraph 3.2 of this Schedule 2.

# Schedule 3 Authorised Guarantee Agreement

THIS AUTHORISED GUARANTEE AGREEMENT is made on 200

**BETWEEN:** 

(A) [(No.) whose registered office is at] [of] (the "Assigning Tenant")

(B) [(No.) whose registered office is at] [of] (the "Landlord") [and

(C) [(No.) whose registered office is at] [of] (the "Assigning Tenant's

**Guarantor**")]

#### NOW THIS DEED WITNESSES AS FOLLOWS:

#### 1. DEFINITIONS AND INTERPRETATION

#### 1.1 Definitions

In this agreement the following words and expressions have the following meanings:

"Assignee" means

"Assignment" means the assignment by the Assigning Tenant to the Assignee of the Lease

"Authorised Guarantee Period" means the period from and including the date of the Assignment until the date (if any) when the Lease is assigned by the Assignee to a third party or (if such Assignment is an excluded Assignment within the meaning of section 11(1) of the Landlord and Tenant (Covenants) Act 1995) until the next Assignment of the Lease which is not an excluded Assignment

"Landlord" means [insert name of Landlord] and includes the party in whom the reversion immediately expectant on the determination of the Term (as that expression is defined in the Lease) is for the time being vested

"Lease" means [insert details of Lease] and includes all or any deeds and documents made pursuant to or supplemental to that lease whether or not expressed to be so

### 1.2 Interpretation

- (a) Expressions used but not defined in this agreement have the meanings given to them in the Lease
- (b) At any time when the Assigning Tenant or the Landlord comprises two or more parties such expressions shall include all or either of any such parties and obligations expressed or implied to be made by or with any of them shall be deemed to be made by or with such parties jointly and severally

#### 2. RECITALS

- 2.1 The reversion immediately expectant on the determination of the Term is vested in the Landlord
- 2.2 The Term is vested in the Assigning Tenant who wishes to enter into the Assignment

2.3 The Landlord is willing to consent to the Assignment on condition that the Assigning Tenant enters into this agreement as required by clause 4.11 of the Lease

#### 3. COVENANT AND INDEMNITY BY ASSIGNING TENANT

- 3.1 In consideration of the Landlord consenting to the Assignment the Assigning Tenant covenants with the Landlord that:
  - (a) the Assignee (failing whom the Assigning Tenant) will at all times during the Authorised Guarantee Period duly perform and observe all the covenants on the part of the Tenant contained in the Lease (including without limitation the payment of the Rents and all other sums payable under the Lease) in the manner and at the times specified in the Lease and
  - (b) it will indemnify and keep indemnified the Landlord against all proper claims demands losses damages liability costs fees and expenses whatsoever sustained by the Landlord by reason of or arising out of any default by the Assignee in the performance and observance of any of its obligations under the Lease

#### 4. NATURE OF ASSIGNING TENANT'S OBLIGATIONS

- 4.1 The obligations of the Assigning Tenant hereunder are primary obligations
- 4.2 The Assigning Tenant is jointly and severally liable with the Assignee for the fulfilment of all the obligations of the Assignee as Tenant under the Lease
- 4.3 Notwithstanding any legal limitation disability or incapacity on or of the Assignee or any other fact or circumstance whether known to the Landlord or not the Landlord may proceed against and recover from the Assigning Tenant as if the Assigning Tenant was named as the Tenant in the Lease
- 4.4 The Landlord shall not be obliged to make any demand on the Assignee before enforcing its rights against the Assigning Tenant hereunder

#### 5. WAIVER BY ASSIGNING TENANT

The obligations of the Assigning Tenant hereunder are to constitute a continuing security in addition to and without prejudice to any other rights which the Landlord may have and the Assigning Tenant hereby waives any right to require the Landlord to proceed against the Assignee or to pursue any other remedy whatsoever which may be available to the Landlord before proceeding against the Assigning Tenant

#### 6. POSTPONEMENT OF CLAIMS BY ASSIGNING TENANT AGAINST ASSIGNEE

- Unless otherwise agreed by the Landlord the Assigning Tenant will not claim or accept any payment or property in any liquidation bankruptcy composition or arrangement of the Assignee pari passu or in priority to the Landlord nor seek to recover (whether directly or by way of set-off lien counterclaim or otherwise) any money or other property nor exercise any other right or remedy whatsoever in respect of any sum which may be or become due to the Assigning Tenant from the Assignee nor exercise any rights of subrogation or indemnity against the Assignee until in each such case all the Assigning Tenant's obligations hereunder have been performed and discharged in full
- 6.2 The Assigning Tenant will as soon as reasonably practicable pay to the Landlord an amount equal to any set-off in fact exercised by it and will promptly pay or transfer to the Landlord any payment or distribution or benefit in fact received by it notwithstanding the terms of clause 6.1 above

6.3 Until the Assigning Tenant's obligations hereunder have been performed the Assigning Tenant agrees that it will exercise any rights of subrogation against the Assignee and any rights to prove in a liquidation of the Assignee which it may have in accordance with the directions of the Landlord

#### 7. POSTPONEMENT OF PARTICIPATION BY ASSIGNING TENANT IN SECURITY

The Assigning Tenant confirms it has not taken and undertakes it will not take from the Assignee any security in connection with its obligations hereunder and declares that any security so taken shall be held on trust for the Landlord and further the Assigning Tenant agrees that it shall not be entitled to participate in any security held by the Landlord in respect of the Assignee's obligations to the Landlord under the Lease nor to stand in the place of the Landlord in respect of any such security until all the obligations of the Assignee or the Assigning Tenant to the Landlord under or in relation to the Lease have been performed or discharged in full

#### 8. NO RELEASE OF ASSIGNING TENANT

None of the following nor any combination thereof shall release determine discharge or in any way lessen or affect the liability of the Assigning Tenant hereunder or otherwise prejudice or affect the right of the Landlord to recover from the Assigning Tenant to the full extent of its obligations hereunder:

- any neglect delay or forbearance of the Landlord in endeavouring to obtain payment of the Rents or other amounts payable under the Lease by the Assignee or in enforcing the performance or observance of any of the obligations of the Assignee as Tenant under the Lease
- 8.2 any lawful refusal by the Landlord to accept Rents tendered by or on behalf of the Assignee
- 8.3 any extension of time given by the Landlord to the Assignee
- 8.4 the transfer of the Landlord's reversion or (subject to section 18 of the Landlord and Tenant (Covenants) Act 1995) any variation of the terms of the Lease (including any reviews of the Principal Rent) or (save as to the part surrendered) the surrender of any part thereof
- 8.5 the release of any person for the time being jointly or severally liable for the Assignee's obligations or liable as surety for the Assignee's obligations
- 8.6 any change in the constitution structure or powers of any of the Assignee the Assigning Tenant or the Landlord or the liquidation administration or bankruptcy (as the case may be) of any of the Assignee the Assigning Tenant or the Landlord
- 8.7 any legal limitation or any immunity disability or incapacity of the Assignee (whether or not known to the Landlord) or the fact that any dealings with the Landlord or the Assignee may be outside or in excess of the powers of the Assignee or the Assigning Tenant
- 8.8 the taking variation compromise renewal release or refusal or neglect to perfect or enforce any rights remedies or securities against the Assignee or any other person
- 8.9 any disclaimer of this Lease by a liquidator or trustee in bankruptcy or similar officer appointed to or in respect of the Tenant and have such power
- any other act omission matter or thing whatsoever whereby but for this provision the Assigning Tenant would be exonerated or released either wholly or in part (other than a release by deed given by the Landlord)

#### 9. **DISCLAIMER OF LEASE**

- 9.1 The Assigning Tenant undertakes to the Landlord that:
  - (a) if a liquidator or trustee in bankruptcy or similar officer having such power shall disclaim or surrender the Lease or
  - (b) if the Assignee shall cease to have legal existence

then the Assigning Tenant shall if the Landlord by notice in writing given to the Assigning Tenant within three months after such disclaimer or other event occurs so requires accept from and execute and deliver to the Landlord a counterpart of a new lease of the Premises for a term commencing on the date of the disclaimer or other event and continuing for the residue then remaining unexpired of the Term (as specified in the Lease) such new lease to be at the cost of the Assigning Tenant and to be at the same rents and subject to the same covenants conditions and provisions as are contained in the Lease

- 9.2 If the Lease is disclaimed and for any reason the Landlord does not require the Assigning Tenant to accept a new lease of the Premises in accordance with this clause 9 the Landlord shall be entitled to require that the Assigning Tenant pays to the Landlord on demand an amount equal to the Rents for the period commencing with the date of such disclaimer and ending on whichever is the earlier of:
  - (a) the date three months after such disclaimer and
  - (b) the date (if any) upon which the Premises are re-let

### 10. CUMULATIVE POWERS AND AVOIDANCE OF PAYMENTS

- 10.1 The powers conferred on the Landlord hereunder are cumulative without prejudice to its powers under the general law and may be exercised as often as the Landlord thinks appropriate The Landlord may in connection with the exercise of its powers join or concur with any person in any transaction scheme or arrangement whatsoever
- 10.2 If the Landlord reasonably considers that any amount paid by the Assigning Tenant hereunder is capable of being avoided or set aside on the liquidation or administration of the Assigning Tenant or otherwise then for the purposes of this agreement (save in relation to any liability for payment of interest hereunder or under the Lease) such amount shall not be considered to have been paid
- Any settlement or discharge between the Landlord and the Assignee and/or the Assigning Tenant shall be conditional upon no security or payment to the Landlord by the Assignee or the Assigning Tenant or any other person being avoided or set aside or ordered to be refunded or reduced by virtue of any provision or enactment relating to bankruptcy insolvency or liquidation for the time being in force and accordingly (but without limiting the Landlord's other rights hereunder) the Landlord shall be entitled to recover from the Assigning Tenant the value (not exceeding the amount for which the Assignment Tenant would otherwise have been liable hereunder) which the Landlord has placed upon such security or the amount of any such payment as if such settlement or discharge had not occurred

#### 11. BENEFIT OF GUARANTEE

11.1 The covenants undertakings and agreements of the Assigning Tenant hereunder shall enure for the benefit of the successors and assigns of the Landlord to the Lease without the necessity for any Assignment thereof to such successors and assigns

11.2 Without prejudice to clause 11.1 above the Landlord may Assign the benefit of the provisions of the Lease and the covenants undertakings and agreements of the Assigning Tenant hereunder to any third party and the Assigning Tenant shall join in such documents as may be necessary to effect such Assignment

#### 12. INTEREST ON LATE PAYMENT

The Assigning Tenant will pay Interest on all sums payable by it to the Landlord hereunder (save to the extent that interest has already been included in the sum payable to the Landlord pursuant to this agreement)

#### 13. COSTS AND EXPENSES

The Assigning Tenant will indemnify the Landlord against all the Landlord's reasonable and proper legal and other costs losses charges and expenses (on a full indemnity basis) arising in connection with any modification amendment release and/or enforcement or attempted enforcement of or preservation of the Landlord's rights under this agreement (but not so as to pay any such amounts under this provision as well as under the provisions of its Lease)

#### 14. SET-OFF

Save as required by law all payments to be made by the Assigning Tenant hereunder will be made in full without any deduction for any set-off or counterclaim the Assigning Tenant may have against the Landlord

#### 15. WAIVER

No delay or omission by the Landlord in exercising any right power or privilege hereunder shall impair such right power or privilege or be construed as a waiver of such right power or privilege

#### 16. **INVALIDITY**

If at any time any one or more provisions of this agreement is or becomes invalid illegal or unenforceable in any respect under any law the validity legality and enforceability of the remaining provisions hereof shall not be in any way affected or impaired thereby

#### 17. SUB-GUARANTEE

#### 17.1 Sub-Guarantee

- (a) In consideration of the Landlord giving its consent to the Assignment, the Assigning Tenant's Guarantor:
  - (i) irrevocably and unconditionally guarantees to the Landlord that the Assigning Tenant shall duly and punctually comply with all obligations on the part of the Assigning Tenant contained in this agreement; and
  - (ii) covenants with the Landlord that in any case of default by the Assigning Tenant, the Assigning Tenant's Guarantor shall comply with the obligations referred to in clause 17.1(a)(i) above and shall, on demand, make good to the Landlord on a full indemnity basis all losses, costs, damages and expenses caused to the Landlord by such default
- (b) The Assigning Tenant's Guarantor's liability under this agreement shall be as principal debtor and not merely as surety

- (c) The provisions of clause 8 of this agreement shall apply to this clause 17 as if they were set out in full in this clause 17 except that in their application to this clause 17, references in clause 8:
  - (i) to the Assigning Tenant shall be read as references to the Assigning Tenant's Guarantor; and
  - (ii) to the Assignee shall be read as references to the Assigning Tenant

#### 17.2 Supplemental documents

- (a) Clause 17.2(b) applies where the Assigning Tenant enters (or may be required by the Landlord to enter) into any document supplemental to the Lease for the purpose of consenting to the Assignee entering into such document and confirming that (subject to section 18 of the Landlord and Tenant (Covenants) Act 1995) all the obligations of the Assigning Tenant under this authorised guarantee agreement will remain in full force and effect
- (b) The Assigning Tenant's Guarantor shall enter into such supplemental document for the purpose of confirming that (subject to section 18 of the Landlord and Tenant (Covenants) Act 1995) its obligations under this sub-guarantee will remain in full force and effect

#### 17.3 Obligation to join in new lease accepted by the Assigning Tenant

If the Assigning Tenant takes a new lease pursuant to clause 9 of this agreement (such new lease being on the terms referred to in clause 9) the Assigning Tenant's Guarantor shall enter into that new lease as guarantor. The Assigning Tenant's Guarantor shall execute and deliver the counterpart of that new lease to the Landlord within one month of the Landlord having required the Assigning Tenant to take the new lease

#### 17.4 Obligation to take new lease

- (a) The Assigning Tenant's Guarantor shall take a new lease of the Premises from the Landlord if required to do so by the Landlord within the relevant time period:
  - (i) if the Assigning Tenant (having been required to do so pursuant to clause 9 of this agreement) fails to take the new lease within the time limit stated in clause 9 of this agreement; or
  - (ii) if the liability of the Assignee under the Lease is disclaimed and the Landlord has not required the Assigning Tenant to take a new lease pursuant to clause 9 of this agreement,

and the relevant time period in relation to clause (i) above is 14 months after the Landlord has received notice of the disclaimer of the Assignee referred to in clause 9 of this agreement, and in relation to clause (ii) above is 12 months after the Landlord has received at notice

(b) If the Assigning Tenant's Guarantor is required to accept a new lease as tenant, the new lease will be on the terms mentioned in clause 9 of this agreement (except that it will have no provision for a guarantor to be party to it) and the Assigning Tenant's Guarantor shall execute and deliver a counterpart of the new lease to the Landlord within one month of the Landlord having required the Assigning Tenant's Guarantor to take such new lease

(c) The Assigning Tenant's Guarantor shall pay to the Landlord on demand all legal and other costs and expenses (on an indemnity basis) and a sum equal to VAT on them which may be payable in connection with the grant of a new lease pursuant to clause 17.4 of this agreement

IN WITNESS whereof this agreement has been executed as a deed on the date first above written

[Execution clauses to be added as required]

# Schedule 4 Reporting Information

#### 1. Definitions

In this schedule the following expressions shall have the following meanings:

"Capital Expenditure" means expenditure incurred by the Tenant on an arm's length basis to maintain, upgrade, acquire or replace the buildings, fixtures, fittings and equipment at the Premises and which is:

- (a) treated as tangible fixed assets or intangible fixed assets that are integral to a relevant tangible fixed asset in the Tenant's audited financial statements for that Financial Year prepared in compliance with:
  - (i) for so long as BMI Healthcare Limited (No. 2164270) is the Tenant of this Lease, Relevant GAAP;
  - (ii) otherwise, the Tenant's prevailing accounting principles at the time; or
- (b) an investment in a subsidiary, associate or joint venture of the Tenant to the extent that the subsidiary, associate or joint venture owns assets which would have been treated as tangible fixed assets or intangible fixed assets in the Tenant's audited financial statements had they been acquired directly.

"CQC" means (as the case may be):

- (a) the Care Quality Commission or any successor body regulating the Premises in England;
- (b) Healthcare Inspectorate Wales or any successor body regulating the Premises in Wales; or
- (c) Healthcare Improvement Scotland or any successor body regulating the Premises in Scotland

"EBIT" means, for any Relevant Period, the consolidated profits of the Tenant from ordinary activities carried on, at or in respect of the Premises (without double counting):

- (a) before deducting any interest, commission, fees, discounts, prepayment fees, premiums or charges, other similar finance payments and other amounts due under any hedging agreements which relates to any of the foregoing in respect of that Relevant Period, in each case, of the Tenant calculated on a consolidated basis;
- (b) before deducting any amount of Tax on profits, gains or income paid or payable by the Tenant;
- (c) after deducting (to the extent included) any accrued interest, commission, fees, discounts, prepayment fees, premiums or charges and other similar finance payments owing to the Tenant and before adding any amounts paid or payable to the Tenant under any hedging agreement which relates to any of the foregoing;
- (d) excluding any items (positive or negative) of a one-off, non-recurring extraordinary or exceptional nature;

- (e) after deducting (to the extent otherwise included) any gain over book value arising in favour of the Tenant in connection with the disposal of any asset (not being a disposal made in the ordinary course of trading) during that Relevant Period and any gain arising on any revaluation of any asset during that Relevant Period;
- (f) after adding back (to the extent otherwise deducted) any loss against book value incurred by the Tenant in connection with the disposal of any asset (not being any disposals made in the ordinary course of trading) during that Relevant Period or any loss arising on any revaluation of any asset during that Relevant Period;
- (g) after adding back (to the extent otherwise deducted) acquisition and restructuring and similar costs;
- (h) after adding back (to the extent otherwise deducted) any amount attributed to rent in respect of the Premises for accounting purposes to the extent greater than rent payable in respect of the Premises during the Relevant Period or, as the case may be, after deducting (to the extent otherwise included) any amount of rent payable in respect of the Premises during the Relevant Period to the extent greater than the amount attributed to rent in respect of the Premises for accounting purposes; and
- (i) after adding back (to the extent otherwise deducted) any fees, costs or charges of a non-recurring nature related to any equity offering, investments, joint ventures, acquisitions or borrowings permitted under the Tenant's loan or debt facilities (whether or not successful);

provided that each of paragraphs (a) to (i) shall be determined by reference to the ordinary activities carried on, at or in respect of the Premises.

"EBITDA" means, in respect of any Relevant Period, EBIT for that Relevant Period plus:

- (a) any amount attributable to the amortisation or depreciation; and
- (b) any amount attributable to the impairment of assets of the tenant at the Premises (or, as applicable, deducting any amount attributable to the reversal of any previous impairment charge made) to the extent not already added back in determining EBIT for that Relevant Period.

"EBITDAR" means, in respect of any Relevant Period, EBITDA plus Rent

"Financial Year" means the annual accounting period of the Tenant

"Quarter" means each period of three months ending on 31 March, 30 June, 30 September and 31 December in any year or, if the Tenant's Financial Year does not start or end on one of those dates, such dates as the Tenant notifies to the Landlord to accord with its Financial Year

"Rent" means, in respect of any Relevant Period, the aggregate of all amounts payable by the Tenant in connection with the letting, licence or grant of other rights of or use or occupation of the Premises, including (if applicable) each of the following amounts:

- (a) rent (including the Rents), licence fees and equivalent amounts payable;
- (b) any other moneys payable in respect of occupation and/or usage of a relevant property and any fixture and fitting on that property including any fixture or fitting on that property for display or advertisement, on licence or otherwise;
- (c) any sum payable under any policy of insurance in respect of loss of rent or interest on rent;

- (d) any sum payable, or the value of any consideration given, for the grant, surrender, amendment, supplement, waiver, extension or release of any document under which the Tenant has a letting, licence or grant of other rights of or use or occupation of any part of the Premises (a "Lease Document");
- (e) any sum payable in respect of a breach of covenant or dilapidations under any Lease Document;
- (f) any sum payable by or distribution received or receivable from any guarantor of any occupational tenant under any Lease Document;
- (g) any contributions payable by the Tenant to any applicable landlord by way of VAT or a contribution to (i) ground rent; (ii) insurance premia; (iii) the cost of an insurance valuation; (iv) a service or other charge; or (v) a reserve or sinking fund; and
- (h) any interest payable on, and any damages, compensation or settlement payable in respect of, any sum referred to above

"Relevant GAAP" means in respect of the Tenant, UK Generally Accepted Accounting Principles (FRS 101), as reported in the audited financial statements for the Financial Year ending on 31 March 2018

"Relevant Period" means each period of four consecutive Quarters ending on the last day of each Quarter

"Revenue" means, in respect of a twelve month period, the net revenue, sales or turnover of the Tenant for that twelve month period as shown in the Tenant's audited consolidated financial statement (if applicable) or management accounts for that period.

#### 2. **Quarterly information**

The Tenant shall provide to the Landlord, as soon as the same become available, but in any event within 45 days after the end of each Quarter, the following information in respect of the business operated and/or managed at the premises for the preceding twelve months:

- 2.1 Revenue
- 2.2 the CQC rating for any hospital operated at the Premises
- 2.3 EBITDAR
- 2.4 details of any Capital Expenditure, broken down by:
  - (a) Facilities, including refurbishment
  - (b) Medical equipment, including diagnostics
  - (c) Information technology
  - (d) Other
- 2.5 notification of the following details in respect of each underletting at the Premises notified to the Landlord during that Quarter
  - (a) the rent payable (including details of any rent free period)

- (b) the term of the underlease
- (c) the part(s) of the Premises which are underlet
- (d) the identity of the undertenant
- 2.6 For so long as [Name of MPT entity] is the Landlord:
  - (a) Details of any litigation affecting the buildings at the Premises in excess of £250,000 (and, for the avoidance of doubt, not including any clinical matters, or any details the provision of which would breach the Tenant's legal or regulatory duties or obligations) provided that the Tenant shall not be obliged to report such litigation if the Tenant reasonably considers that its business or operations would be materially adversely affected if details of the relevant litigation were made public
  - (b) Details of any compulsory purchase proposals for that Quarter affecting the Premises in excess of £50,000
  - (c) Details of any insurance claims in respect of any buildings at the Premises in excess of £250,000 (and, for the avoidance of doubt, not including any clinical matters, or any details the provision of which would breach the Tenant's legal or regulatory duties or obligations) provided that the Tenant shall not be obliged to report such insurance claim if the Tenant reasonably considers that its business or operations would be materially adversely affected if details of the relevant insurance claim were made public
- 2.7 quarter and year to date statements of operations as compared to previous year periods

#### 3. Monthly information

The Tenant shall provide to the Landlord, as soon as the same becomes available, but in any event within 45 days after the end of each calendar month, a monthly statement of operations, balance sheet and cash flow in respect of the Tenant.

#### 4. Year End information

The Tenant shall provide to the Landlord:

- (a) as soon as the same become available after the end of Tenant's financial year-end, draft annual financial statements of (a) the Tenant's ultimate parent company and (b) any guarantor to this lease; and
- (b) as soon as the same become available, but in any event within 270 days after the end of Tenant's financial year-end, audited financial statements of (a) the Tenant's ultimate parent company and (b) any guarantor to this lease.

#### 5. CQC notifications

The Tenant shall notify the Landlord in writing as soon as reasonably practicable (and in any event within two Business Days) after the termination or suspension or invalidation of any CQC licence and/or certificate necessary for the operation of the Premises as a hospital (except where such termination relates only to some services carried out at the Premises and was requested by the Tenant for business reasons)

#### 6. Warranties

The Tenant warrants to the Landlord on each date that the Tenant delivers the financial and other operating information referred to in this schedule to the Landlord pursuant to this schedule that such information:

- 6.1 is, where relevant, properly extracted from the Tenant's up to date books and records which form the basis of the Tenant's audited financial statements;
- 6.2 is not misleading in any material respect;
- in the case of any financial information, is prepared using UK generally accepted accounting principles consistently applied;
- 6.4 in the case of unaudited financial information, fairly represents in all material respects, taking into account that it was prepared for internal management purposes, the results of operations for the period to which the information relates; and
- 6.5 in the case of audited financial information, gives a true and fair view of its financial condition and results of operations for the period to which the Reporting Information relates

and the Landlord acknowledges and agrees that it shall have no rights or remedies in respect of any statement, representation, assurance or warranty (whether made innocently or negligently and whether oral or written, express or implied) in respect of any information provided by the Tenant except as expressly set out in this paragraph 6 (provided that nothing in this paragraph 6 shall limit or exclude any liability for fraud).

# Schedule 5 Approved Works Covenants

#### 1. **Definitions**

In this schedule:

"Approved Works" means the variations alterations and/or additions to the Premises permitted by the Lease

"CDM Regulations" means the Construction (Design and Management) Regulations 2015

"Consents" means all necessary permissions, licences and approvals for the Approved Works under the Planning Acts, the building and fire regulations, and any other statute, bye law or regulation of any competent authority and under any covenants or provisions affecting the Premises and as otherwise properly required from owners, tenants or occupiers or any adjoining premises

#### 2. Approved Works Covenants

- 2.1 Before starting the Approved Works the Tenant must:
  - (a) obtain any Consents that are required before they are begun;
  - (b) fulfil any conditions in the Consents required to be fulfilled before they are begun;
  - (c) where relevant comply with its obligations in clause 4.13(g); and
  - (d) provide the Landlord with any information relating to the Approved Works properly required by its insurers
- 2.2 If it starts the Approved Works, the Tenant must carry out and complete them;
  - (a) diligently and expeditiously, and in any event before the expiry or sooner determination of the Term (except where the Reversionary Lease is subsisting at the relevant time);
  - (b) in a good and workmanlike manner and with good quality materials;
  - (c) in compliance with the Consents (where relevant) and all applicable legislation (including the Planning Acts) and with the requirements of the insurers of the Premises and (where applicable) of any competent authority or utility provider;
  - (d) with as little interference as reasonably practicable to the owners and occupiers of any adjoining premises; and
  - (e) in compliance, to the extent applicable, with the CDM Regulations
- 2.3 The Tenant must make good as soon as reasonably practicable any physical damage caused by carrying out the Approved Works
- 2.4 The Tenant must permit the Landlord and its employees or agents to enter the Premises on no less than 7 Business Days prior written notice and during reasonable times to inspect the progress of the Approved Works and in doing so the Tenant shall be entitled to accompany the Landlord to enter in a manner which does not interfere with the exercise of this covenant to permit entry
- 2.5 As soon as reasonably practicable following completion of the Approved Works the Tenant must:

- (a) obtain any Consents that are required on their completion;
- (b) remove all debris and equipment used in carrying out the Approved Works;
- (c) permit the Landlord and its employees or agents to enter the Premises on no less than 7 Business Days prior written notice and during reasonable times to inspect the completed Approved Works and in doing so the Tenant shall be entitled to accompany the Landlord to enter in a manner which does not interfere with exercise of this covenant to permit entry
- 2.6 If the CDM Regulations apply to the Approved Works, the Tenant must:
  - (a) comply with them and use reasonable endeavours to ensure that any person involved in the management, design, and construction of the Approved Works complies with their respective obligations under the CDM Regulations;
  - (b) if the Landlord would be treated as a client for the purposes of the CDM Regulations, agree to be treated as the only client in respect of the Approved Works; and
  - (c) promptly following completion of the Approved Works provide the Landlord with a copy of any health and safety file relating to the Approved Works and deliver the original file to the Landlord at the expiry or sooner determination of the Term (except where the Reversionary Lease is subsisting at the relevant time)
- 2.7 If the Approved Works invalidate an existing EPC or require the commissioning of an EPC, the Tenant must:
  - (a) obtain an EPC from an assessor approved by the Landlord (acting reasonably) and give the Landlord written details of the unique reference number for that EPC; or
  - (b) pay the Landlord's reasonable costs of obtaining an EPC

#### 3. No warranty relating to Approved Works

- 3.1 The Landlord gives no express or implied warranty (and the Tenant acknowledges that the Tenant must satisfy itself):
  - (a) as to the suitability, safety, adequacy or quality of the design or method of construction of any Approved Works;
  - (b) that any Approved Works may lawfully be carried out; or

that any of the services supplying the Premises will either have sufficient capacity for or otherwise not be adversely affected by any Approved Works

#### Schedule 6 Landlord's call option

#### 1. Definitions

In this schedule the following expressions shall have the following meanings:

- "Asset Price" means the net book value (as determined in accordance with Relevant GAAP) of a Relevant Asset as stated on the Call Option Asset Register
- "Call Option" means the option granted by the Tenant to the Landlord by paragraph 4.1 (Landlord's Call Option)
- "Call Option Asset Register" Means (i) the list of assets as at 31 December 2018 delivered by the Tenant to the Landlord on 31 January 2019 and (ii) any replacement list of assets delivered to the Landlord in accordance with paragraph 3 of this Schedule
- "Call Option Assets" means all of the Tenant's right title and interest in all chattels, equipment, fittings and moveable contents situate in or at the Premises or used exclusively in connection with the permitted user of the Premises (including any replacements or substitutions for or additions thereto during the Term whether acquired in accordance with any obligations of the Tenant in relation to capital expenditure from any other person or with any contributions in respect of capital expenditure or otherwise but excluding any such assets that have been acquired by the Landlord pursuant to this schedule and any Call Option Assets disposed of or transferred in accordance with paragraph 4.11 (Assignment)
- "Completion Date" means the date specified in the Option Notice for completion of the transfer of the Relevant Assets in accordance with paragraph 4.2(b)
- "Lease Termination Event" means any of the events set out in clause 7.1(a) (Re-entry)
- "Option Notice" Means a notice referred to in paragraph 4.2 (Exercise of Call Option)
- "Option Price" means the consideration payable for the Relevant Assets as determined in accordance with paragraph 4.3 (Option Price)
- "Relevant Assets" means those Call Option Assets specified in an Option Notice delivered in accordance with paragraph 4.2 (Exercise of Call Option), which may be a generic description such as "all of the Call Option Assets"
- "Relevant GAAP" means UK Generally Accepted Accounting Principles (FRS 101), as reported in the audited financial statements for the Tenant's financial year ending on 31 March 2018
- "Subsidiary" means in relation to any company or corporation, a company or corporation:
- (a) which is controlled, directly or indirectly, by the first mentioned company or corporation;
- (b) more than half the issued share capital of which is beneficially owned, directly or indirectly by the first mentioned company or corporation; or
- (c) which is a Subsidiary of another Subsidiary of the first mentioned company or corporation,

and for this purpose, a company or corporation shall be treated as being controlled by another if that other company or corporation is able to direct its affairs and/or to control the composition of its board of directors or equivalent body

"Tenant Group" means the parent company of the Tenant for the time being and each of its Subsidiaries for the time being

"Transferee" means the Landlord or such other person as the Landlord directs in the Option Notice to whom the Relevant Assets are to be transferred pursuant to this schedule

#### 2. Interpretation

For the purposes of this Schedule 6 only, a "regulation" includes any regulation, rule, official directive, request or guideline (whether or not having the force of law) of any governmental, intergovernmental or supranational body, agency, department or of any regulatory, self-regulatory or other authority or organisation.

#### 3. Delivery of call option asset register

The Tenant shall provide to the Landlord a Call Option Asset Register on 31 January 2019 in respect of the Tenant's assets as at 31 December 2018. Thereafter, the Tenant shall provide to the Landlord the Call Option Asset Register (as at 30 September) on 31 October in each year.

#### 4. Option To Purchase Call Option Assets

#### 4.1 Landlord's Call Option

The Tenant hereby grants to the Landlord an option to purchase the Call Option Assets (or any part of the Call Option Assets) on the terms set out below

#### 4.2 Exercise of Call Option

- (a) The Call Option may be exercised by notice in writing by the Landlord to the Tenant in substantiality the form of the Option Notice at Schedule 7, within three months after:
  - the Landlord has exercised its right of re-entry under the terms of the Lease and has forfeited this Lease as a result of any of the Lease Termination Events having occurred; or
  - (ii) the end or sooner determination of the Term for whatever reason (unless (i) the Reversionary Lease is subsisting at the relevant time and (ii) the Term has not determined or ended as a result of the Landlord exercising its right of re-entry pursuant to clause 7.1)
- (b) The Option Notice shall specify the Relevant Assets and the Transferee and state the date for completion of the transfer of the Relevant Assets, being not less than 30 days after the date of the Option Notice

### 4.3 Option Price

The price payable for the Relevant Assets shall be:

(a) in the case of any Relevant Assets acquired by the Tenant after the date of this Lease, provided the Relevant Asset and its net book value (as determined in accordance with Relevant GAAP) is stated on the Call Option Asset Register most recently delivered to the Landlord prior to its exercise of the Call Option, the Asset Price; and

(b) in the case of all other Relevant Assets (including Relevant Assets owned by the Tenant on or prior to the date of this Lease), the sum of £1.00 only

#### 4.4 No Security Interests

- (a) The Tenant covenants that it will not create incur or permit to subsist any security interest on any of the Call Option Assets to secure any liability whether actual or contingent
- (b) Sub-paragraph (a) above does not apply to:
  - (i) any lien arising by operation of law or in the ordinary course of the Tenant's business (including retention of title arrangements) securing amounts not more than 30 days overdue;
  - (ii) any security interest arising as a consequence of any finance or capital lease the outstanding principal amount of which does not exceed £50,000,000 (or its equivalent) at any time in aggregate applicable to all members of the Tenant Group or such other amount permitted under the terms of the Tenant Group Loan Agreement;
  - (iii) any security interest created in favour of the Landlord; and
  - (iv) any security interest granted in favour of the lenders and other finance parties which ranks in priority behind any security granted in favour of the Landlord under the terms of an intercreditor agreement in form and substance satisfactory to the Landlord

#### 4.5 Sale free of encumbrances

The Tenant covenants that the Relevant Assets shall be sold subject to paragraph 4.4(b) free from all liens charges and encumbrances (but subject in the case of any Relevant Assets held under a lease or contract for hire to the terms of those agreements) and with all rights attached thereto at the date of the exercise of the Call Option

#### 4.6 Completion

- (a) Completion of the transfer of the Relevant Assets shall take place on the Completion Date
- (b) The Tenant covenants to complete the transfer of the Relevant Assets to the Transferee on the Completion Date (subject to payment of the Option Price being made by the Landlord (or on its behalf) to the Tenant as it directs on the Completion Date) and the Tenant covenants that it will on the Completion Date:
  - cause to be delivered or made available to the Transferee at such reasonable location as the Transferee shall specify:
    - (A) such documents as the Transferee may reasonably require to complete the transfer to it of the Relevant Assets together with all deeds and documents of title relating thereto; and
    - (B) possession of all the Relevant Assets which are capable of passing by delivery; and

- (ii) will use reasonable endeavours to provide all manuals, service records and other associated documents in respect of the Relevant Assets provided that such items are:
  - (A) within the Tenant's possession and control; and
  - (B) are readily available on the Completion Date
- (c) Subject to payment of the Option Price being made by the Landlord (or on its behalf) to the Tenant as it directs on the Completion Date, beneficial ownership and risk in respect of the Relevant Assets shall pass to the Transferee on the Completion Date. In respect of any Relevant Assets not delivered or formally transferred, novated or assigned to the Transferee on the Completion Date, the Tenant shall thereafter be trustee for the Transferee until they are so delivered and the Tenant covenants that is shall deal with those Relevant Assets in such manner
- (d) To the extent that the Relevant Assets comprise the benefits of contracts which cannot effectively be assigned to the Transferee except by the agreements of novation or by consents to assignment the Landlord and the Tenant covenant that:
  - (i) the Landlord shall (and shall procure that any other Transferee shall) and the Tenant shall cooperate to take all reasonable steps to procure the novation of the said contracts or to obtain the necessary consents; and
  - unless or until each of the said contracts has been novated or assigned the Tenant shall hold the benefit thereof upon trust for the Transferee and where the Landlord is the Transferee it shall, and where the Landlord is not the Transferee, the Landlord shall procure that the Transferee shall, on behalf of the Tenant (at the Transferee's expense) perform all the obligations of the Tenant thereunder arising after the Completion Date and the Landlord shall indemnify the Tenant against all liability (and all costs reasonably incurred by the Tenant in connection therewith) which may be incurred by the Tenant as a result of any act neglect or default on the part of the Landlord and (if it is not the Landlord) the Transferee after the Completion Date to perform or comply with any such obligation of the Tenant

#### 4.7 Further Assurance

The Tenant covenants that it shall with all due diligence execute such other documents and take such other steps as may reasonably be required by the Landlord to vest the title to the Relevant Assets in the Transferee and to give effect to the Call Option

#### 4.8 No Removal

The Tenant covenants that:

- (a) save as permitted pursuant to paragraph 4.11 of this Schedule, for the period during which the Call Option is exercisable it will not remove, without the consent of the Landlord, any of the Call Option Assets; and
- (b) following exercise of the Call Option and prior to the Completion Date, it will not remove, without the consent of the Landlord, any of the Relevant Assets,

from the Premises

#### 4.9 **Indemnity**

The Transferee shall not be liable for any indebtedness of the Tenant outstanding as at the Completion Date or any breach of contract, negligence, breach of duty or other breach of circumstance giving rise to liability to any third party which is attributable to any act, neglect or default of the Tenant and the Tenant covenants that it shall indemnify the Transferee against any liability which the Transferee may incur in respect of or in connection with any such indebtedness or as a result of any such act neglect or default (and all costs reasonably incurred by the Transferee in connection therewith)

#### 4.10 Power of Attorney

- (a) In order to secure the Landlord's interest in the Call Option Assets the Tenant has entered into a security power of attorney dated on or around the date of the Deed of Variation which shall be irrevocable until six months after the end of the Term and the Landlord undertakes to the Tenant not to exercise its power of attorney unless the Call Option has been exercised
- (b) In the event of an assignment of or any other dealing with the interest of a Landlord under a Lease, the Landlord for the time being shall be entitled to require from the Tenant, and the Tenant covenants that it shall, enter into a Security Power of Attorney in favour of the Landlord and the mortgagee of the Landlord
- (c) In the event of an assignment of the interest of the Tenant under this Lease, the Landlord for the time being shall be entitled to require from the proposed assignee that it shall enter into a Security Power of Attorney in favour of the Landlord and any other persons holding a superior interest in the Premises or this Lease including any superior landlord and the mortgagee of the Landlord or any superior landlord

#### 4.11 Assignment

The Tenant shall not transfer or otherwise dispose of or part with possession of any of the Call Option Assets save for:

- (a) a transfer to a proposed assignee pursuant to any assignment of this Lease as permitted pursuant to clause 4.11 (Dealings)
- (b) a sale in the ordinary course of business of the Tenant operated at the Premises; or
- (c) a disposal or sale of obsolete assets which have outlasted their useful life and which are no longer required for the efficient operation of its business

#### 4.12 Option Notice not to prejudice rights

The service of an Option Notice by the Landlord shall be without prejudice to the other rights and remedies available to the Landlord, whether under this Lease or otherwise, any other right to determine this Lease which is available to the Landlord and any liability of any party in relation to any earlier breach

#### 4.13 Term of Call Option

This Call Option shall remain in full force and effect for the entire period of the Term and for a period of 3 months after the expiry of the Term. This Call Option shall, for the avoidance of doubt, survive any termination, forfeiture or expiry of this Lease.

#### 5. **VAT**

The Option Price and all other sums payable by the Landlord to the Tenant pursuant to this schedule are exclusive of VAT. Where under or pursuant to this schedule the Tenant makes a supply to any Landlord for VAT purposes and the Tenant or any member of the Tenant Group is required to account for VAT in respect of that supply each Landlord shall be subject to the receipt of a valid VAT invoice pay to the Tenant an amount equal to such VAT in addition to any other consideration for that supply.

#### 6. Non-Merger

On the exercise of the Call Option by the Landlord, the provisions of this schedule shall not merge with the transfer of the Relevant Assets to the Transferee but shall continue in full force and effect to the extent that anything remains to be performed or observed under it.

### 7. Invalidity Of Certain Provisions

If any term of this schedule or the application of such term to any person or circumstances shall to any extent be invalid or unenforceable, the same shall be severable and the remainder of this schedule or the application of such term to persons or circumstances other than those as to which it is held invalid or unenforceable shall not be affected by such invalidity or unenforceability and each term and provision of this schedule shall be valid and be enforced to the fullest extent permitted by law.

## Schedule 7 Call Option Notice

To: [Name and address of Tenant or successor in title] (the "Tenant")

### **BY HAND**

### BY RECORDED DELIVERY POST

### IMPORTANT TIME CRITICAL NOTICE

1.	We refer to Schedule 6 of the lease dated 31 July 2006 (as amended, varied and/or supplemented from time to time) between (1 the Landlord and (2) the Tenant (the "Lease") in respect of the Call Option Assets situated in the property known as land and buildings at [Property details] (the "Property").
2.	We, [name and address, and (where appropriate) company registration no. of Landlord], as "Landlord" under the Lease, give notice to you of our exercise of the Call Option referred to in Schedule 6 of the Lease in respect of [all of the Call Option Assets [those of the Call Option Assets set out in the attached schedule] (the "Relevant Assets").
3.	The date of completion for the transfer of the Relevant Assets shall be [].
4.	The "Transferee" in relation to the Relevant Assets shall be [the Landlord/[-]].
	Dated [ ]

Director/Company Secretary and on behalf of [name of Landlord]

# Schedule 8 Security Power of Attorney

Inis rower or alloring is made on by     Limited (the appointer) whose registered onice is at	by [] Limited ("the <b>Appointer</b> ") whose registered office is at [_	F ATTORNEY is made on	THIS POWER OF
---	--	-----------------------	---------------

1.	The Appointer hereby irrecoverably appoints by way of security any director from time to time (the "Attorney") of [] Limited
	(registered number [_]) whose registered office is at [_] (the "Company") to be the true and lawful attorney of the Appointer for
	the purpose only of transferring or selling in accordance with Schedule 6 of the lease dated 31st day of July 2006 and made
	between (1) [Name of MPT entity] and (2) BMI Healthcare Limited as varied by a deed of variation dated 31 March 2008 (the
	"Lease") relating to the property at [Property details] (the "Premises") to the Company (or as the Attorney directs) all the
	Appointor's right title and interest in all or any of the equipment contents and other assets and choses in action situate in or at the
	Premises or used exclusively in connection with or otherwise referable to the Premises (the "Appointer's Assets") to sign on
	behalf of the Appointor any document required to be executed by the Appointor (and if necessary seal and deliver the same as a
	deed) to give effect to such sale or transfer and to execute any other documents and to all acts things that are in the opinion of the
	Attorney or the Company necessary or expedient to the said documents and the intention thereto.

- 2. The Appointor hereby undertakes to ratify and confirm whatever the Attorney shall lawfully do or cause to be done by virtue of this Power of Attorney and to indemnify them and the Company and keep them indemnified against all costs and expenses properly incurred by him pursuant to this Power of Attorney.
- 3. The Appointor HEREBY DECLARES that this Power of Attorney shall be irrevocable until the earliest of:
  - (a) the date falling 6 months after the end of the term of the Lease;
  - (b) the date on which the Appointor grants a further Power of Attorney in the terms of this Power of Attorney to such other person or persons as the Attorney shall direct; and
  - (c) the date on which the Appointor having obtained all consents necessary under the Lease assigns all its interest in the Appointor's Assets to such other person who shall have granted to the Attorney or such other person as the Attorney directs a power of attorney in accordance with paragraph 4.10 (*Power of Attorney*) of Schedule 6 of the Lease (as amended from time to time) and made between, amongst others, the Appointor and the Attorney.
- 4. This Power of Attorney shall be governed and construed in accordance with English law.

# Schedule 9 Schedule of rent payments between 31 October 2018 and 30 October 2019

Rent payment (£[Property details])

For the period 31 October 2018 to 30 January 2019

On 15 November 2018: £[Property details]
On the date of the Deed of Variation: £[Property details]
On 15 December 2018: £[Property details]

For the period 31 January 2019 to 29 April 2019

On 8 January 2019: £[Property details]

For the period 30 April 2019 to 30 July 2019

On 8 April 2019: £[Property details]

For the period 31 July 2019 to 30 October 2019

On 8 July 2019: £[Property details]

Executed as a deed by [Name of MPT entity] acting by two directors	
	Director
	Director
Executed as a deed by <b>BMI HEALTHCARE LIMITED</b> acting by a director in the presence of a witness:	Director
	Director
	Witness Signature:
	Name:
	Address:
	Occupation:
Executed as a deed by <b>CIRCLE HEALTH HOLDINGS LIMITED</b> acting by a director in the presece of a witness:	
	Director
	Witness Signature:
	Name:
	Address:
	Occupation:
- 69	)-

# CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO RULE 13a-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934

#### I, Edward K. Aldag, Jr., certify that:

- 1) I have reviewed this quarterly report on Form 10-Q of Medical Properties Trust, Inc.;
- 2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4) The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5) The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 7, 2020 /s/ Edward K. Aldag, Jr.

Edward K. Aldag, Jr.

Chairman, President and Chief Executive Officer

# CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO RULE 13a-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934

#### I, R. Steven Hamner, certify that:

- 1) I have reviewed this quarterly report on Form 10-Q of Medical Properties Trust, Inc.;
- 2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4) The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5) The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 7, 2020 /s/ R. Steven Hamner

R. Steven Hamner

Executive Vice President and Chief Financial Officer

# CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO RULE 13a-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934

#### I, Edward K. Aldag, Jr., certify that:

- 1) I have reviewed this quarterly report on Form 10-Q of MPT Operating Partnership, L.P.;
- 2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4) The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5) The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 7, 2020

/s/ Edward K. Aldag, Jr.

Edward K. Aldag, Jr.

Chairman, President and Chief Executive Officer of the sole member of the general partner of MPT Operating Partnership, L.P.

# CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO RULE 13a-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934

#### I, R. Steven Hamner, certify that:

- 1) I have reviewed this quarterly report on Form 10-Q of MPT Operating Partnership, L.P.;
- 2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4) The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5) The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 7, 2020

/s/ R. Steven Hamner

R. Steven Hamner

Executive Vice President and Chief Financial Officer of the sole member of the general partner of MPT Operating Partnership, L.P.

# CERTIFICATION OF CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER PURSUANT TO RULE 13a-14(b) UNDER THE SECURITIES EXCHANGE ACT OF 1934 AND 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with this quarterly report on Form 10-Q of Medical Properties Trust, Inc. (the "Company") for the quarter ended June 30, 2020 (the "Report"), each of the undersigned, Edward K. Aldag, Jr. and R. Steven Hamner, certifies, pursuant to Section 18 U.S.C. Section 1350, that:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 7, 2020

/s/ Edward K. Aldag, Jr.

Edward K. Aldag, Jr.

Chairman, President and Chief Executive Officer

/s/ R. Steven Hamner

R. Steven Hamner

Executive Vice President and Chief Financial Officer

# CERTIFICATION OF CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER PURSUANT TO RULE 13a-14(b) UNDER THE SECURITIES EXCHANGE ACT OF 1934 AND 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with this quarterly report on Form 10-Q of MPT Operating Partnership, L.P. (the "Company") for the quarter ended June 30, 2020 (the "Report"), each of the undersigned, Edward K. Aldag, Jr. and R. Steven Hamner, certifies, pursuant to Section 18 U.S.C. Section 1350, that:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 7, 2020

#### /s/ Edward K. Aldag, Jr.

Edward K. Aldag, Jr.

Chairman, President and Chief Executive Officer of the sole member of the general partner of MPT Operating Partnership, L.P.

#### /s/ R. Steven Hamner

R. Steven Hamner

Executive Vice President and Chief Financial Officer of the sole member of the general partner of MPT Operating Partnership, L.P.